## UC Berkeley UC Berkeley Electronic Theses and Dissertations

### Title

A search for new physics in multijet final states of pp collisions at s=13-TeV using the ATLAS detector at the Large Hadron Collider

Permalink

https://escholarship.org/uc/item/5rv264z1

**Author** Amadio, Brian Thomas

**Publication Date** 2019

Peer reviewed|Thesis/dissertation

A search for new physics in multijet final states of pp collisions at  $\sqrt{s} = 13 \ TeV$  using the ATLAS detector at the Large Hadron Collider

by

Brian Thomas Amadio

A dissertation submitted in partial satisfaction of the

requirements for the degree of

Doctor of Philosophy

 $\mathrm{in}$ 

Physics

in the

Graduate Division

of the

University of California, Berkeley

Committee in charge:

Professor Marjorie Shapiro, Chair Professor Barbara Jacak Professor Karl A. van Bibber

Summer 2019

A search for new physics in multijet final states of pp collisions at  $\sqrt{s} = 13 \ TeV$  using the ATLAS detector at the Large Hadron Collider

Copyright 2019 by Brian Thomas Amadio

#### Abstract

A search for new physics in multijet final states of pp collisions at  $\sqrt{s} = 13 \ TeV$  using the ATLAS detector at the Large Hadron Collider

by

Brian Thomas Amadio Doctor of Philosophy in Physics University of California, Berkeley Professor Marjorie Shapiro, Chair

Supersymmetry, one of the of the most promising extensions to the Standard Model of Particle Physics, predicts a number of as-yet-undiscovered particles, evidence of which could be found in LHC collision events. This thesis presents a search for R-parity violating Supersymmetric signal events in ATLAS proton-proton collision data. The search uses 36.1  $fb^{-1}$ of  $\sqrt{s} = 13 \ TeV$  collision data, taken by ATLAS in 2015 and 2016. As no excess of events above the background prediction was found, limits are set on the production cross sections of gluinos and neutralinos under two different decay scenarios. The resulting limits are a significant extension beyond the reach of the Run-1 search for the same process. To all the teachers in my life

# Contents

C	onten	nts		ii								
$\mathbf{Li}$	st of	Figure	es	v								
$\mathbf{Li}$	st of	Tables	3	xii								
1	Intr	oduction										
<b>2</b>	The	Stand	lard Model of Particle Physics	3								
	2.1	Electro	oweak sector and the Higgs mechanism	4								
		2.1.1	Matter fields	4								
		2.1.2	Symmetries and Lagrangian	5								
		2.1.3	Spontaneous symmetry breaking	5								
	2.2	Quant	um Chromodynamics	7								
		2.2.1	Matter fields	7								
		2.2.2	Symmetries and Lagrangian	7								
		2.2.3	QCD coupling constant	8								
		2.2.4	Parton distribution functions	9								
	2.3	Limita	tions	11								
		2.3.1	Fine tuning and the Higgs mass	11								
		2.3.2	Grand unification	11								
		2.3.3	Dark matter	12								
3	Sup	$\mathbf{ersym}$	metry	13								
	3.1	Motiva	ation for new physics	13								
		3.1.1	Naturalness	13								
		3.1.2	Grand unification	15								
		3.1.3	Dark matter	16								
	3.2	Theory	y and phenomenology	17								
		3.2.1	Superfields and superpotentials	18								
		3.2.2	SUSY particles	18								
	3.3	R-Pari	ty and R-parity violation	20								

		3.3.1 3.3.2 3.3.3 3.3.4	R-parity	20 20 21 22
4	The	LHC	and ATLAS Detector 2	23
	4.1	The La	arge Hadron Collider	23
		4.1.1	Layout	25
		4.1.2	Magnets	27
		4.1.3	Radio frequency cavities	27
		4.1.4	Beams and luminosity	28
	4.2	The A	TLAS detector	30
		4.2.1	Coordinates	30
		4.2.2	Magnet systems	30
		4.2.3	Inner detector	32
		4.2.4	Calorimeters	39
		4.2.5	Muon spectrometer	45
		4.2.6	Trigger	49
5	Tota			1
J	5 1	Tho pr	roton proton collision onvironment	51
	5.2	Clusto	ring algorithms	52
	0.2	5 9 1	IRC sofoty	55
		5.2.1	Sequential recombination algorithms	56
	53	J.2.2		50
	5.0 5.4	Becone	$\Delta S = 1$	50 61
	0.4	5 4 1	Topological coll clusters	61
		5.4.1	Topo cluster calibration	64
		5.4.2 5.4.3	Integlibration	65
		5.4.0		68
		0.4.4		00
6	Ana	lysis C	Dverview 7	75
	6.1	The R	PV UDD signature	75
	6.2	Previo	bus searches and limits	75
	6.3	Search	strategy and discriminating variables	77
7	Мо	nte Ca	rlo Models of Signal and Background	21
•	7 1	Signal	modeling	81
	1.1	7.1.1	Event generation	81
		719	Cross-section	82
		713	Fast simulation with Atlfast-II	83
	7.2	Backg	round modeling	85

8 D	Data Sample, Event Selection, and Jet Definitions	88
8.	.1 Data sample	88
8.	.2 Trigger	88
8	3 Event selection	88
8	4 Iet definitions h-tagging and h-matching	89
0. 8	5 Control signal validation and uncertainty determination regions	00
0.	.5 Control, signal, valuation, and uncertainty determination regions	90
9 B	Background estimation method	93
9.	.1 Jet mass templates	93
	9.1.1 Template creation	93
	9.1.2 Template validation	93
	9.1.3 Background estimation using jet mass templates	95
	9.1.4 Systematic uncertainty	06
0	2 Effect of signal contamination on background estimation	08
9.	2 Effect of signal contamination on background estimation $\dots \dots \dots \dots$	90
9.	.3 Effect of <i>tt</i> on background estimation uncertainty	99
9.	.4 Background estimation method applied to multijet Monte Carlo	99
10 S	ignal systematics	107
11 P	Posulta	112
11 I( 1'	1.1. Dradieted and observed wields	119
1.		110
1.	1.2 Statistical Interpretation	113
12 C	Conclusion	119
Bibli	iography	120

iv

# List of Figures

2.1	Summary of Standard Model production cross sections measured by ATLAS, compared to theoretical predictions [2]	
2.2	Measured values of the strong coupling constant $\alpha_s$ and its predicted values across a range of energies [5].	
2.3	Parton distribution functions for the proton, evaluated at NLO for two different energy scales [8].	1
3.1	Top-quark loop diagram, the leading correction to the Higgs mass-squared. This contribution is quadratically divergent in the cutoff scale [15].	1
3.2	Stop loop diagram, the leading SUSY correction to the Higgs mass-squared. This contribution is quadratically divergent in the cutoff scale [15].	1
3.3	A representative sample of the various SUSY searches performed by ATLAS, and their resulting mass lower bounds [17].	1
3.4	Running of the three Standard Model coupling constants under the Standard Model (left) and MSSM (right) RGEs. For the MSSM, unification occurs at	
3.5	$\mu \approx 2 \times 10^{16} \ GeV \ [16].$ Tree-level diagram for the decay $p^+ \rightarrow \pi^0 e^+$ , with both L-violating and B-violating vertices [15]	1
3.6	Diagrams for the two decay processes that are the subject of this search. The direct decay (left) and cascade decay (right) both involve effective RPV vertices	2
	containing off-shell squark propagators	2
4.1	Cumulative luminosity delivered by the LHC over time for each of the years from 2011 to 2018 [25]	2
4.2	Planned operation timeline of the LHC, showing the different run periods, shut- down periods, and the energy and luminosity goals, up to and including the	
	HL-LHC upgrades [26]	2
4.3	Schematic of the LHC layout. The clockwise-rotating beam, shown in red, is referred to as beam 1, and the counter-clockwise-rotating beam, shown in blue,	
4.4	is beam 2. Locations of the four major detector experiments are labelled [27] Cross-section schematic of an LHC dipole magnet [28]	$\frac{2}{2}$

4.5	Schematic of a cryomodule, containing four RF cavities used to accelerate and store the LHC beam [27].	29
4.6	A diagram showing the layout of the ATLAS magnet system. The central solenoid is used to apply a force that curves the trajectories of charged particles in the inner detector. In red are the toroid coils, used to bend the tracks of charged particles passing through the muon spectrometer [23]	31
4.7	A picture of the ATLAS barrel toroid after installation. In the center of the image is the calorimeter and central solenoid, before being moved into the final	01
4.8	A quarter-section plan showing the layout of inner detector subsystems and their dimensions. Not shown is the innermost layer of the pixel detector, the IBL,	32
4.9	which was installed in May 2014 [27]	33
4 10	the electrons and holes are accelerated in opposite directions under the electric field. The flow of electrons generates a current in the readout electronics Graphic illustrating how $n^+$ in $n$ pixel sensors continue to operate after the type	34
4.10	inversion that results from irradiation. In (a), the unirradiated state, the bulk is n-type, and the depletion zone occurs between the $p^+$ -doped back side. After type inversion, in (b), the depletion zone occurs between the now p-type bulk	
	and the $n^+$ -doped front side [30]	35
4.11	The IBL as it was inserted into the pixel detector $[32]$	36
4.12	Part of a TRT end-cap, showing the radial layout of the $37 \ cm$ straw tubes [36].	37
4.13	Schematic of the TRT straw wall design. Two coated Kapton layers are sealed	
	together with polyure than $[36]$	38
4.14	Layout of the ATLAS calorimeter system $[23]$	40
4.15	Schematic of EM calorimeter, showing the granularity of each layer $[37]$	41
4.16	Schematic of a tile calorimeter module $[23]$	42
4.17	Schematic of the central (left) and extended (right) tile calorimeter, showing the	
	$\eta$ - and depth-dependent segmentation [23]	43
4.18	Schematic of hadronic end-cap calorimeter module $[23]$	43
4.19	Schematic showing the layout of the forward calorimeters and end-cap calorime-	
	ters [23]	44
4.20	Amount of material, in units of interaction length, versus pseudorapidity, includ-	
	ing all material before the EM calorimeter, the EM calorimeter, and the hadronic	
	calorimeters. The light blue indicates the amount of material in front of the muon	
1.01	spectrometer for $ \eta  < 3.0$ [23]	45
4.21	Layout of the muon spectrometer, indicating the location of barrel and end-cap	10
4 00	toroids, along with the iour subsystems $[23]$	40
4.22	Cross-sections in $x - y$ (left) and $y - z$ (right) of the muon spectrometer [39].	41

vi

4.23	The contribution to muon spectrometer $p_T$ resolution from various sources over a range of $p_T$ [39]	47
4.24	Flowchart outlining the level 1 and high-level trigger design and data acquisition system [40]	50
5.1	Diagram of a proton-proton collision event. Illustrated are the initial state radi- ation, underlying event, hard-scatter process, final state parton shower, fragmen- tation, hadron decays, and final state OED radiation [41]	59
5.2	Distribution of third-leading jet $p_T$ vs. leading jet $p_T$ (left) and jet multiplicity (right) for events simulated with several different event generators and parton shower methods. The bottom page of each plot shows the ratio with respect to	02
5.3	the Pythia 8 result. [43]	54
5.4	represents rapidity, and y-axis represents transverse momentum [44] Illustration of infrared unsafety. In an event with a W boson decaying to two hard partons, the emission of a soft gluon can change the result of an infrared-unsafe	56
	algorithm [44].	56
5.5	Results of running four different jet clustering algorithm on the same set of parton- level simulated event data. SISCone (lower-left) is a cone-based algorithm, while the other three are sequential recombination algorithms defined by different values	
5.6	of $p$ [50] Radial separation between top decay products vs. top-quark $p_T$ for top quarks produced from a theoretical heavy resonance $Z'$ [51]	59 60
5.7	Illustration of the trimming procedure, one of several grooming methods used to remove the contribution of unassociated radiation to fat jet mass [54]	61
5.8	Jet mass distributions from simulated LHC proton-proton collisions events before and after grooming, for various grooming techniques. HERWIG is used to simu- late $\sqrt{s} = 7 \ TeV$ QCD dijet events (left) and Standard Model $t\bar{t}$ events (right)	
5.9	with parton-level $p_T$ between 500 and 600 $GeV$ [55] Average energy and energy resolution measured for 180 $GeV$ and 20 $GeV$ pions with different values of the topo-clustering parameters $S$ , $N$ , and $P$ . The x-axis of each plot represents choices for all three parameters, and the y-axis indicates either the average energy deposited in each cluster $\leq E_{thetern} \geq$ or the energy	62
5.10	either the average energy deposited in each cluster $\langle E_{clusters} \rangle$ of the energy resolution $RMS / \langle E_{clusters} \rangle$ [57]	63
	electromagnetic showers, while clusters below that line have a higher probabil- ity $[56]$ .	65

vii

5.11	Illustration of the procedure used to account for signal lost to lost cells and dead material [56].	66
5.12	Jet mass response as a function of $\eta$ before and after JMS calibration, for anti- $k_T$ jets with $R = 1.0$ at the LCW scale, without grooming [51].	67
5.13	Transverse and longitudinal impact parameter significance distributions for a sample of simulated $t\bar{t}$ events [60]	69
5.14	Secondary vertex reconstruction rate versus $p_T$ and $\eta$ for b-jets, c-jets, and light flavor jets from a simulated sample of $t\bar{t}events$ [60].	70
5.15	Illustration of the difference between the secondary vertex (a) and multi-vertex (b) algorithms [62]	70
5.16	JetFitter vertex reconstruction rate versus $p_T$ and $\eta$ for b-jets, c-jets, and light flavor jets from a simulated sample of $t\bar{t}$ events [63]	71
5.17	MV2c20 BDT output for b-jets, c-jets, and light flavor jets from a simulated sample of $t\bar{t}$ events [62]	72
5.18	B-tagging efficiency versus light-jet (a) and c-jet (b) rejection for a range of MV2 output cut values. The red curves are for MV2c20, in which 20% c-jets are included in the background training sample. The blue curves are for Mv2c00, which only includes light jets in the background training sample. The improved c-jet rejection and worsened light-jet rejection can seen by comparing the two [63].	73
6.1	Diagrams for the two decay processes that are the subject of this search. The direct decay (left) and cascade decay (right) both involve effective RPV vertices containing off shell squark propagators	76
6.2	Observed and expected limits in the $(m_{\tilde{g}}, m_{\tilde{\chi}})$ plane for the cascade-decay model in ATLAS Bun 1 with $\sqrt{s} = 8 TeV$ [64]	70
6.3	Observed and expected limits on $m_{\tilde{g}}$ ), along with theoretical cross-section for the direct-decay model in ATLAS Run 1 with $\sqrt{s} = 8 \ TeV$ [64]. Different limits are set for four different branching fraction scenarios. In sub-figure (a), only light-quark decays are allowed. In sub-figure (b), each gluino decay results in a top quark in the final state. In sub-figure (c), each gluino decay results in a bottom quark in the final state. In sub-figure (d), each gluino decay results in both a top	
6.4	and a bottom quark in the final state [64]	78
7.1	The grid of gluino and neutralino masses for which simulated cascade decay events	79
	were generated. The triangles indicate the mass points for which simulated events were generated.	82

viii

7.2	Gluino pair-production cross-section and uncertainty for $\sqrt{13} TeV$ proton-proton collisions at the LHC, calculated at NLO+NLL [77]	83
7.3	The process used for the AFII/FullSim comparison study. The signal is very similar to the cascade decay model used in the search, but the gluinos decay to a	
7.4	neutralino+ $t\bar{t}$ rather than a neutralino and two light quarks Jet mass response vs true jet mass (left) and truth jet $p_T$ for simulated signal events using both the full detector simulation (full sim) and Atlfast-II (fast sim). Points and error bars show the best-fit mean and width for a Gaussian fit to the core of the distribution in each bin. The model used for this study is similar to the cascade decay model described in 6, except each gluino decays to a neutralino + $t\bar{t}$ rather than a neutralino and two light quarks. The gluino mass is 1250 GeV and neutralino mass is 450 GeV. Jets are required to have $p_T$ greater than 100 GeV and $ \eta  < 2.8$ .	84
7.5	Kinematic distributions of large- $R$ jets for simulated events using both the full detector simulation (full sim) and Atlfast-II (fast sim). The model used for this study is similar to the cascade decay model described in 6, except each gluino decays to a neutralino + $t\bar{t}$ rather than a neutralino and two light quarks. The gluino mass is 1250 GeV and neutralino mass is 450 GeV. Jets are required to have $p_T$ greater than 100 GeV and $ \eta  < 2.8$ .	86
7.6	Jet multiplicity and distribution (left) and $M_J^{\Sigma}$ for events with at least 5 large- R jets (right) for simulated events using both the full detector simulation (full sim) and Atlfast-II (fast sim). The model used for this study is similar to the cascade decay model described in 6, except each gluino decays to a neutralino + $t\bar{t}$ rather than a neutralino and two light quarks. The gluino mass is 1250 GeV and neutralino mass is 450 GeV. Jets are required to have $p_T$ greater than 100 GeV and $ \eta  < 2.8.$	87
8.1	Trigger efficiency versus large- $R$ jet $p_T$ threshold for events with $\geq 4$ and $\geq 5$ large- $R$ jets. For events with $\geq 4$ large- $R$ jets, and additional requirement of leading-jet $p_T > 400 \ GeV$ ensures 100% trigger efficiency.	89
8.2	Workflow illustrating how the different regions are used in the analysis	91
9.1 9.2	Two representative template distributions used in the analysis, showing a com- parison between data in black and simulation in red. The solid markers show the templates derived from b-matched jets, while the empty markers show the templates derived from non-b-matched jets. In (a), template jets are required to have 600 $GeV < p_T < 644 \ GeV$ and $0.5 <  \eta  < 1.0$ . In (b), templates jets are required to have 733 $GeV < p_T < 811 \ GeV$ and $1.5 <  \eta  < 2.0 \ [69]$ Number of jets contributing to each template bin for the non-b-matched (left)	94
9.3	and b-matched (right) templates. $\dots \dots \dots$	94 95

9.4	Jet mass response plots showing the discrepancy between average dressed and kinematic jet masses in the two uncertainty determination regions, binned by $p_T$ and $ \eta $ . Since the discrepancy is always larger in UDR1, only UDR1 is used to	
	derive uncertainties [69]. $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	97
9.5	Signal injection test. The background prediction is first run only on a data sample (a), and then on a data sample injected with simulated signal events (b). The	
	increase in background prediction and systematic uncertainty can be seen	99
9.6	The effect of signal contamination on the nominal background prediction. At each neutralino and gluino mass point, simulated signal events are used in the kinematic sample to generate a background prediction. These predictions are	
	compared to the prediction from data only. In both cases, the templates are	
9.7	drawn only from data	100
	multijet only (right column). The top two plots show the response for non-b- matched jets only, and the bottom two plots show the response for b-matched	
	jets only.	101
9.8	Jet mass response in UDR2 for Pythia8 multijet and $t\bar{t}$ (left column) and Pythia8	
	multijet only (right column). The top two plots show the response for non-b-	
	matched jets only, and the bottom two plots show the response for b-matched	
0.0	jets only	102
9.9	Predicted and observed $M_J^2$ distributions for Pythia8 multijet and $t\bar{t}$ Monte Carlo events in the 4jSR region. The red histogram shows the predicted distribution including statistical uncertainty. The group and blue histograms show the system	
	atic uncertainty. The left plot includes only the high- $p_T$ systematic uncertainty,	109
0.10	and the right plot shows only the low- $p_T$ systematic uncertainty	103
9.10	events in the 4iSBb region. The red histogram shows the predicted distribution	
	including statistical uncertainty. The green and blue histograms show the system-	
	atic uncertainty. The left plot includes only the high- $p_T$ systematic uncertainty,	
	and the right plot shows only the low- $p_T$ systematic uncertainty	104
9.11	Predicted and observed $M_J^{\Sigma}$ distributions for Pythia8 multijet and $t\bar{t}$ Monte Carlo	
	events in the 5jSR region. The red histogram shows the predicted distribution	
	including statistical uncertainty. The green and blue histograms show the system-	
	atic uncertainty. The left plot includes only the high- $p_T$ systematic uncertainty,	105
0.19	and the right plot shows only the low- $p_T$ systematic uncertainty	105
9.12	Predicted and observed $M_{\tilde{J}}$ distributions for Pythias multijet and $t\bar{t}$ Monte Carlo events in the 5iSPh region. The red histogram shows the predicted distribution	
	including statistical uncertainty. The green and blue histograms show the system-	
	atic uncertainty. The left plot includes only the high- $p_T$ systematic uncertainty.	
	and the right plot shows only the low- $p_T$ systematic uncertainty	106

10.1	Nominal and systematically-shifted $N_{jet}$ distributions for four signal points, show- ing the shifted distributions for internal generator event weights, QCD scale, and	
	$\alpha_s$	108
10.2	Nominal and systematically-shifted $N_{b-jet}$ distributions for four signal points, showing the shifted distributions for internal generator event weights, QCD scale,	
	and $\alpha_s$ .	109
10.3	Nominal and systematically-shifted $ \Delta \eta_{1,2} $ distributions for four signal points, showing the shifted distributions for internal generator event weights, QCD scale,	
	and $\alpha_s$ .	110
10.4	Nominal and systematically-shifted $M_J^{\Sigma}$ distributions for four signal points, show- ing the shifted distributions for internal generator event weights, QCD scale, and	
	$\alpha_s$	111
11.1	Distribution of $M_J^{\Sigma}$ in the four-jet inclusive (4jSR), four-jet b-tag (4jSRb), five- int inclusive (5jSR) and four-jet b tag (5jSRb) parises. The solid and lines show	
	jet inclusive (5)SR), and five-jet b-tag (5)SRb) regions. The solid red lines snow	
	the predicted $M_{\tilde{J}}$ distributions, along with the total uncertainty shaded in red. The black data show the observed $M^{\Sigma}$ distributions. The corresponding $M^{\Sigma}$ suita	
	The black dots show the observed $M_J$ distributions. The corresponding $M_J$ cuts are indicated with the solid black vertical line and arrow for each region. Two	
	separate signal regions are defined for the 5iSBb region one with an $M^{\Sigma}_{\Sigma}$ cut	
	of 0.6 $TeV$ , and one with a cut of 0.8 $TeV$ . The green and blue dashed lines show the predicted $M_{\Sigma}^{\Sigma}$ contribution from cascade-decay signal events from two	
	different mass points	114
11.2	Predicted and observed limits for the cascade decay model over a range of $m_{\tilde{\alpha}}$ and	
	$m_{z^0}$ values. Both predicted and observed limits are shown with $1\sigma$ uncertainty	
	bands. Mass points to the left of the red solid curve are excluded. The gray	
	dashed curve shows the limits from the Run-1 analysis. Due to slight excess in	
	the signal region, the observed limit is less than expected [69]	117
11.3	Predicted and observed cross-section limits for the direct decay model over a range of $m_{\tilde{z}}$ values. The black solid curve indicates the upper bound on the	
	signal production cross section for each gluino mass point. Also shown are the	
	estimated gluino pair production cross-sections at each mass point [69]. Since the	
	theoretical cross-section is smaller than the upper bound for every mass point	
	studied, this signal cannot be excluded for any mass point. $\ldots$	118

# List of Tables

3.1	The MSSM chiral superfields, including their names, symbols, and components. Quantum numbers for the Standard Model symmetry group transformations are also given [15]	10
3.2	The MSSM gauge superfields, including their names, symbols, and components. Quantum numbers for the Standard Model symmetry group transformations are also given [15].	19
4.1	Summary of number of layers crossed per track as well as intrinsic accuracy in $R - \phi$ and z for each of the inner detector subsystems.	39
5.1	Description of each of the 24 inputs for the MV2 BDT $[63]$	72
8.1	Summary of the requirements defining the control, uncertainty determination, validation, and signal regions. Requirements are placed on the large-R jet multiplicity $(N_{jet})$ , the presence or absence of a b-tagged small-R jet $(b\text{-tag})$ , the $p_T$ of the leading jet $(p_{T,1})$ , the pseudorapidity difference between the two leading jets $( \Delta \eta_{12} )$ , and the scalar sum of the first four leading jets in the event $(M_J^{\Sigma})$ [69].	92
10.1	Summary of contributions to systematic uncertainties on the signal yield from various sources, including the number of nuisance parameters and approximate range of uncertainty size for each.	112
11.1	Predicted and observed yields in all signal regions used in the analysis. The number of events in the corresponding normalization regions, $N_{NR}$ is shown. For each signal region, the minimum value of $M_J^{\Sigma}$ used to define that region is also shown. Additionally, the statistical uncertainty on the background yield, as well as the two systematic uncertainties, derived from the high- $p_T$ and low- $p_T$ UDRs	
11.2	are shown [69]	115 116

#### Acknowledgments

First of all I have to thank Marjorie Shapiro for her invaluable advice and guidance, and Haichen Wang for leading by example with his tireless work ethic and for always taking the time to teach. Thanks to Juerg Beringer and Carl Haber for their mentorship and support when I first joined the LBL ATLAS group. Thanks to Ian Hinchliffe, Zach Marshall, Simone Pagan Griso, and the rest of the LBL ATLAS group for sharing their expertise. Special thanks to Michael Barnett for getting me interested in ATLAS when I came to Berkeley for the first time in 2006. Thanks to my family and friends for helping me get through this, and for always encouraging me, even when it seemed like it would never get done.

# Chapter 1 Introduction

Experimental particle physics is the systematic probing of the fundamental constituents of matter, in order to uncover new insights into laws of nature applicable at the smallest distance scales. To date, the most successful and predictive theory explaining the interactions of elementary particles is the Standard Model. The Standard Model accounts for all of the known elementary particles and their interactions, with the notable exception of gravitational interactions.

In July 2012, the final piece in Standard Model particle puzzle, the Higgs boson, was discovered by the CMS and ATLAS experiments at the Large Hadron Collider. This was the last particle predicted by the Standard Model that had yet to be discovered. But there are still many questions left unanswered by the Standard Model. Answering these questions will require a new theory. This theory must be consistent with all the experimentally-confirmed predictions of the Standard Model, but also go further to explain what the Standard Model can't. Such theories are known as Beyond-the-Standard-Model (BSM) theories.

One of the most promising BSM theories is known as Supersymmetry, or SUSY, which posits a new symmetry of nature. This new symmetry connects the two types of fundamental fields - fermions and bosons. SUSY provides an elegant solution to many of the mysteries of the Standard Model. However, it also predicts a slew of new fundamental particles which have yet to be observed.

The ATLAS detector at the Large Hadron Collider was built with a wide-ranging set of physics goals. In addition to the discovery of the Higgs, the LHC and ATLAS were built to search for evidence of Supersymmetric particles. SUSY has a large number free parameters, whose values can't be predicted from first principles, but would have to be measured experimentally. A SUSY search that fails to discover a new particle nonetheless will often result in constraining the viable parameter space for the theory.

This thesis focuses on a search for a proposed SUSY particle called the gluino. In a specific variant of SUSY, known as R-Parity-Violating or RPV SUSY, the gluinos produced in LHC collisions would decay to large numbers of Standard Model quarks and gluons. These events would leave a characteristic signature in the detector, consisting of a large number of high-mass jets. If gluinos with a low enough mass exist, the rate of these events could

be significantly higher than the rate of such events generated by known Standard Model processes. The aim of the first half of this thesis is to explain the context and motivation for the analysis. The second half will consist of explaining the methods used in the analysis, and a discussion of the results.

Chapter 2 gives more detail about the Standard Model. The mathematical formalism, in the form of Quantum Field Theory (QFT) will be described. The representations of the fields and the Lagrangian governing their dynamics will also be explained. After a discussion of the phenomenological predictions of the theory, section 2.3 will detail several of the limitations of the Standard Model, which help motivate the need for a BSM theory like SUSY. Chapter 3 details how SUSY can resolve several of the questions left unanswered by the Standard Model. A brief description of the theory and the particles it predicts is given. The chapter ends with a discussion of R-Parity violation, and the specific process that is the subject of this search. Next, chapter 4 provides technical descriptions of the Large Hadron Collider and the ATLAS detector. The first section of the chapter explains how the LHC accelerates protons, including the magnet systems that guide the beams around the ring, as well as the ones used to focus the beams at the collision points. The second section details the ATLAS detector, one of the two general-purpose detectors built at the LHC. ATLAS is comprised of many sub-detectors, each with their own specific purposes, which will be described in detail. Chapter 5 explains the phenomenon of jets in particle collision experiments. The mechanisms that give rise to jets in proton-proton collisions are first explained, before detailing the experimental methods used to measure jets. This includes a discussion of jet clustering algorithms that can be used in any particle detector, as well as the specifics of how jets are measured and calibrated in ATLAS in particular. Chapter 6 introduces the strategy used in this particular search. A description of the signal and its signature in the detector is given, followed by details of the discriminating variables and how they can be used to distinguish signal from background. Chapter 7 details the simulated events that were generated for use in the analysis. Simulated signal events are generated for limit-setting, and simulated background events are generated for validating the data-driven background estimation method. Following this, chapter 8 describes the data and event selection used for the analysis, as well as the particular choice of jet reconstruction parameters and requirements. The data-driven background estimation method, involving jet mass templates, is described in detail in chapter 9. A key part of the analysis is the estimation of systematic background uncertainties. Additionally, the treatment of possible signal contamination is discussed. The determination of systematic and statistical uncertainties on the signal yields is given in chapter 10. Finally, chapter 11 shows the results of the search, including the observed and predicted yields in the signal region. The statistical interpretation of these results is also given in this chapter. Conclusions are presented in chapter 12.

# Chapter 2

# The Standard Model of Particle Physics

The Standard Model is a theory describing the interactions between elementary particles. It explains three of the four known fundamental forces, namely the electromagnetic force, the weak nuclear force, and the strong force. It does not include or account for gravitational interactions. It can be fairly characterized as one of the most successful theories in science.

A huge number of predictions made by the theory have been confirmed experimentally, some to an astounding degree of accuracy. Precision measurements of the magnetic moment of the electron have shown the experimental and theoretical values of the fine structure constant to agree better than one part in one billion [1]. The ATLAS detector at the LHC has confirmed the predicted rate of particle production for a very wide range of production processes and final states. A summary of Standard Model measurements made by ATLAS, and their comparisons to theoretical predictions can be seen in figure 2.1. The Standard Model predicted the existence of the W boson, the top quark, and the Higgs boson, which were all later confirmed by experiment.

The Standard Model is expressed in the language of Quantum Field Theory. The theory consists of three generations of matter fields, specified by their representation under the gauge group  $SU(3) \times SU(2) \times U(1)$  and the Poincaré group, as well as a complex scalar field. Poincaré symmetry consists of the Lorentz symmetry of special relativity, plus global translational symmetry.

The Standard Model is a complete theory in the sense that it is internally self-consistent, and all particles predicted by the theory have been observed experimentally. However, the Standard Model does not account for all known physical phenomena, and so cannot be considered a complete theory of nature.



Figure 2.1: Summary of Standard Model production cross sections measured by ATLAS, compared to theoretical predictions [2].

### 2.1 Electroweak sector and the Higgs mechanism

The symmetry constraining the electroweak sector of the Standard Model Lagrangian is  $SU(2)_L \times U(1)_Y$ . The subscript *L* for the SU(2) group indicates that it only acts on the left-handed fields in the theory. Right-handed fields appear as SU(2) singlets.

#### 2.1.1 Matter fields

All left-handed matter fields are SU(2) doublets, while their corresponding right-handed fields are SU(2) singlets. There are three generations of left-handed lepton fields:

$$\psi_L = \begin{pmatrix} \nu_e \\ l_e \end{pmatrix}_L, \ \begin{pmatrix} \nu_\mu \\ l_\mu \end{pmatrix}_L, \ \begin{pmatrix} \nu_\tau \\ l_\tau \end{pmatrix}_L$$
(2.1)

and three generations of right-handed lepton fields:

$$\psi_R = e_R, \ \mu_R, \ \tau_R \tag{2.2}$$

Similarly, there are three generations each of the left-handed SU(2)-doublet quark fields, each of which comes in three colors:

$$\psi_L = \begin{pmatrix} q_u \\ q_d \end{pmatrix}_L, \ \begin{pmatrix} q_c \\ q_s \end{pmatrix}_L, \ \begin{pmatrix} q_t \\ q_b \end{pmatrix}_L$$
(2.3)

Quark color plays a role in QCD interactions, which will be described in the next section. There are also the corresponding SU(2)-singlet right-handed fields:

$$\psi_R = q_{uR}, \ q_{dR}, \ q_{cR}, \ q_{sR}, \ q_{tR}, \ q_{bR} \tag{2.4}$$

#### 2.1.2 Symmetries and Lagrangian

The Lagrangian density satisfying the required symmetries, including all renormalizable terms, can be written as:

$$\mathcal{L}_{EW} = \mathcal{L}_{kin} + \mathcal{L}_{\text{interactions}} \tag{2.5}$$

where the kinetic part of the electroweak Lagrangian is:

$$\mathcal{L}_{kin} = -\frac{1}{4} W^{\mu\nu}_{a} W^{a}_{\mu\nu} - \frac{1}{4} B^{\mu\nu} B_{\mu\nu}$$
(2.6)

 $W^a_\mu$  are the three  $SU(2)_L$  gauge bosons,  $B_\mu$  is the  $U(1)_Y$  (hypercharge) gauge boson, and  $W_{\mu\nu} = \partial_\mu W_\nu - \partial_\nu W_\mu$ . The part of the electroweak Lagrangian describing electroweak interactions is:

$$\mathcal{L}_{\text{interactions}} = \sum_{\text{generations}} \left[ \frac{g}{2} \bar{\psi}_L \gamma^\mu \sigma^i W^i_\mu \psi_L + g' B_\mu \left( Y_{\psi_L} \bar{\psi}_L \gamma^\mu \psi_L + Y_{\psi_R} \bar{\psi}_R \gamma^\mu \psi_R \right) \right]$$
(2.7)

where  $\gamma^{\mu}$  are the Dirac gamma matrices, and Y is the hypercharge associated with the relevant field.

Left-handed leptons and quarks have weak hypercharge Y = -1 and Y = 1/3, respectively. Right-handed leptons have weak hypercharge Y = -2. Right-handed up-type quarks have weak hypercharge Y = 4/3, and right-handed down-type quarks have weak hypercharge Y = -2/3.

This Lagrangian describes the charged-current and neutral-current weak interactions, as well as weak gauge boson self-interactions. However, it is insufficient to describe nature because the gauge bosons are massless in this theory. It would be impossible to include gauge boson mass terms in the electroweak Lagrangian without explicitly breaking the symmetry. Similarly, it's impossible to introduce Lorentz-invariant fermion mass terms that keep the Lagrangian invariant under  $SU(2)_L \times U(1)_Y$ . In order to reconcile this theory with the experimentally observed fact of massive weak gauge bosons and massive fermions, a scalar field will have to be introduced to the theory.

#### 2.1.3 Spontaneous symmetry breaking

The original  $SU(2)_L \times U(1)_Y$  symmetry will be spontaneously broken, via the nonzero Higgs vacuum expectation value, to  $U(1)_{QED}$ . This process of spontaneous symmetry breaking (SSB) generates fermion masses, electroweak gauge boson masses, and the Yukawa coupling terms between fermions. It also generates a new massive real scalar field, known as the Higgs

field. We can introduce a new SU(2)-doublet complex scalar field,  $\phi$ , as well as a potential of the form:

$$V(\phi) = \mu^2 \phi^{\dagger} \phi + \lambda (\phi^{\dagger} \phi)^2$$
(2.8)

with  $\lambda > 0$  and  $\mu^2 < 0$ . The minimum energy state satisfies:  $\phi^{\dagger}\phi = \frac{-\mu^2}{2\lambda}$ . We can then re-parameterize the scalar field as:

$$\phi = \exp\left(i\frac{\sigma_i}{2}\theta^i(x)\right)\frac{1}{\sqrt{2}}\begin{pmatrix}0\\v+H(x)\end{pmatrix}\tag{2.9}$$

where H(x) and  $\theta^i(x)$  are real-valued fields, and  $v = \sqrt{\frac{-\mu^2}{\lambda}}$  is the vacuum expectation value of the Higgs field. Because of the  $SU(2)_L$  symmetry of the theory, we are free to choose convenient values for  $\theta^i(x)$ , without affecting the outcome of any observable predictions. Selecting  $\theta^i(x) = 0$ , the additional kinetic term required in the Lagrangian is:

$$\Delta \mathcal{L}_{EW} = (D_{\mu}\phi)^{\dagger} (D_{\mu}\phi) = \frac{1}{2} \partial_{\mu} H \partial^{\mu} H + \frac{1}{2} (v+H)^2 \left(\frac{g^2}{4} W^{\dagger}_{\mu} W^{\mu} + \frac{g^2}{4\cos^2\theta_W} Z_{\mu} Z^{\mu}\right) \quad (2.10)$$

where  $W_{\mu}$  and  $Z_{\mu}$  are mass eigenstates of the electroweak gauge fields. Each mass eigenstate is a linear combination of the original  $W^a_{\mu}$  and  $B_{\mu}$ . This results in interaction terms between the Higgs field and the W and Z bosons, as well as quadratic terms for these same bosons. A quadratic term in the Lagrangian is physically the same as having a mass term in the Lagrangian. So we find that the W and Z boson masses are:

$$M_w = \frac{1}{2}vg, \ M_Z = \frac{M_W}{\cos\theta_W} \tag{2.11}$$

Thus, through spontaneous symmetry breaking, electroweak gauge boson masses are generated without explicit mass terms in the Lagrangian. The introduction of this scalar field and its associated potential also generates fermionic mass terms, couplings between the fermions and the scalar field, and a mass term for the scalar field itself. Like with the gauge bosons, adding an explicit fermionic mass term would have broken the  $SU(2)_L \times U(1)_Y$  symmetry of the theory. However, the introduction of the new scalar field allows for additional terms in the Lagrangian, which can be written, in unitary gauge, as:

$$L_Y = \frac{1}{\sqrt{2}} \left( v + H \right) \left( c_1 \bar{d}d + c_2 \bar{u}u + c_3 \bar{e}e \right)$$
(2.12)

Once again, after spontaneous symmetry breaking, there appear terms in the Lagrangian that are quadratic in the fields. These are mass terms for the fermions. As with the gauge bosons, fermion masses are proportional to the Higgs vacuum expectation value:

$$m_d = -c_1 \frac{v}{\sqrt{2}}, \ m_u = -c_2 \frac{v}{\sqrt{2}}, \ m_e = -c_3 \frac{v}{\sqrt{2}}$$
 (2.13)

Unlike for the gauge bosons, we find no testable relationship between fermion masses, since the coefficients are independent free parameters of the theory.

The Higgs mechanism allows for the generation of masses for fermions and electroweak gauge bosons without explicitly breaking the original symmetry of the theory. Additional consequences arising from the Higgs mechanism are the relationship between Z and W boson masses, the existence of a new massive scalar field, and interactions between this field and the fermions and gauge bosons. The massive particle associated with this field, the Higgs boson, was discovered in 2012 by the ATLAS and CMS collaborations [3, 4].

### 2.2 Quantum Chromodynamics

Quantum Chromodynamics (QCD) is the part of the Standard Model concerned with strong interactions. Because protons are bound states of quarks and gluons, QCD is needed for making any calculation of event rates at a proton-proton collider. It's especially important for understanding the behavior of jets, which will be described in detail in chapter 5. The particular structure of the theory yields very different behavior at high energy and low energy. At high energy, where the coupling strength is low, quarks and gluons are only weakly bound, so the theory behaves as an approximately free theory, a property known as asymptotic freedom. At lower energies, the coupling strength grows, and quarks and gluons become strongly bound, a property known as confinement.

#### 2.2.1 Matter fields

The fields involved are the quark fields, which also participate in electroweak interactions. In the case of QCD, the quarks are in the fundamental **3** representation of SU(3):

$$\psi = \begin{pmatrix} \psi_R \\ \psi_G \\ \psi_B \end{pmatrix} \tag{2.14}$$

where R, G, B stand for the red, green, and blue color charges. The theory will be invariant under local transformations in this color space.

#### 2.2.2 Symmetries and Lagrangian

Like the electroweak theory, QCD is a non-Abelian gauge theory. The Lagrangian is generated by positing the transformations under which the Lagrangian is invariant, and including all terms that respect these symmetries. For QCD, the symmetry group is SU(3), yielding the following Lagrangian:

$$\mathcal{L}_{QCD} = \sum_{generations} i\bar{\psi}D_{\mu}\gamma^{\mu}\psi - \frac{1}{4}G^{a}_{\mu\nu}G^{\mu\nu}_{a}$$
(2.15)

where

$$D_{\mu} = \partial_{\mu} - ig_s G^a_{\mu} T^a \tag{2.16}$$

is the gauge-covariant derivative,  $g_s$  is the strong coupling constant,  $G^a_{\mu}$  are the eight gluon fields, and  $T^a$  are the infinitesimal generators of SU(3). The gluon field strength tensor,  $G^a_{\mu\nu}$ , is defined as:

$$G^a_{\mu\nu} = \partial_\mu G^a_\nu - g_s f^{abc} G^b_\mu G^c_\nu \tag{2.17}$$

where  $f^{abc}$  are the SU(3) structure constants. This Lagrangian describes the kinematics of massless quarks, their interactions with gluons, and gluon self-interactions. Quarks are given mass through electroweak symmetry breaking, as described in 2.1.3.

#### 2.2.3 QCD coupling constant

The value of the strong coupling constant,  $g_s$ , depends on the energy scale of the process under consideration. The rate of change of the coupling as energy increases is governed by the  $\beta$  function,

$$\beta\left(\alpha_{s}\right) = \frac{\alpha_{s}}{2\pi} \left(\frac{11}{3}n_{\text{colors}} - \frac{4}{3}n_{\text{flavors}}\right)$$
(2.18)

where  $\alpha_s = g_s^2/4\pi$ . In QCD, there are three colors and three flavors, the consequence of which is that the coupling constant decreases with energy. This is known as the "running" of the coupling constant, and results in very different predictions for the theory at different energy scales. Figure 2.2 shows the predicted value of  $\alpha_s$  over a range of energies, along with values measured through a variety of physics processes.



Figure 2.2: Measured values of the strong coupling constant  $\alpha_s$  and its predicted values across a range of energies [5].

#### Asymptotic freedom

At high energies, at or above the GeV scale, the strong coupling constant becomes small. As a result, QCD calculations in this regime are *perturbative*, meaning they can be approximated by a finite sum of terms expanded in powers of the coupling constant. This is similar to the way calculations are done in quantum electrodynamics (QED).

Asymptotic freedom means that as energy grows infinite, quarks and gluons can be treated as completely un-coupled, free particles. At very high energies, their interactions can be treated as small perturbations about the free theory. At the LHC, the hard-scattering cross-section of a quark from one proton colliding with a quark from another proton can be calculated in this perturbative way. However, the full proton-proton collision cross-section calculation requires a combination of both perturbative and non-perturbative calculations.

#### **Confinement and Hadronization**

At lower energies, the strong coupling constant grows larger, and the theory can no longer be treated as perturbative. Because the coupling constant is large, a finite series expanded in powers of the coupling constant is no longer sufficient to estimate the full predictions. Instead, higher-order terms grow larger and larger, which means an infinite number of such terms would be needed to make an accurate prediction. As can be seen in figure 2.2, perturbation theory begins to fail below  $\approx 1 \text{ GeV}$ .

Confinement is a consequence of the large coupling constant at low energies. Since quarks and gluons are strongly coupled, they can no longer exist as free states. Instead, they can only exist as color-neutral bound states, called hadrons. In a collision event, when a quark is generated in the final state, the strong force between the quark and other quarks in the event remains constant as distance between them increases. Equivalently, the potential energy between quarks increases with distance. Eventually it becomes energetically favorable for new quarks to be created from the vacuum, with the correct color charge to neutralize the color charge of the escaping quark. These new particles become bound together in colorneutral free states. This process is known as hadronization. Experimental evidence suggests that hadronization occurs at energies of a few hundred MeV.

Hadronization, along with parton showering, is responsible for the emergence of jets from LHC collisions. Rather than observing the quarks and gluons produced in the collision, the detector can only observe the spray of hadrons that result. Jets will be discussed in detail in chapter 5. Confinement has not been rigorously proven from a theoretical standpoint. However, the existence of color confinement is consistent with empirical evidence.

#### 2.2.4 Parton distribution functions

The QCD Lagrangian describes how individual quarks and gluons behave. But the LHC collides protons, not individual quarks. Protons consist of three valence quarks: two up quarks and one down quark, and a "sea" of quarks, antiquarks, and gluons. When calculating

collision cross-sections for the LHC, the internal structure of the protons must be taken into account.

This internal structure is accounted for using parton distribution functions (PDFs). PDFs are expressed as probability density functions  $f_i(x, Q^2)$ , where *i* denotes the flavor of the parton, *x* is the fraction of momentum carried by the parton, and *Q* is the momentum transfer at which the proton structure is probed, or the energy scale of the hard-scatter process. The values of these PDFs cannot be calculated from first principles, but thanks to the QCD factorization theorem, they can be measured experimentally. The factorization theorem allows for the cross section to be factorized into a hard-scattering part and a normalization factor from the PDFs. By comparing the experimental and theoretical cross sections for a given process, the PDFs can be inferred. Since the PDFs are independent of the hard-scattering process, the data from a variety of processes can be combined. Using the Dokshitzer-Gribov-Lipatov-Altarelli-Parisi (DGLAP) equations [6], which describe how the PDFs depend  $Q^2$ , data from different processes at different momentum-transfer scales can all be combined into a global fit. For PDFs used by ATLAS, the global fit includes data from deep inelastic scattering (DIS) e - p collisions from the HERA-II experiment, jet and  $t\bar{t}$  production from ATLAS and CMS, and observables from several other processes [7].

Figure 2.3 shows the proton PDFs evaluated at NLO for  $Q^2 = 10 \ GeV^2$  and  $Q^2 = 10^4 \ GeV^2$ 



Figure 2.3: Parton distribution functions for the proton, evaluated at NLO for two different energy scales [8].

When calculating LHC cross-sections, the hard-scattering cross-section for different types of partons must be convolved with the PDFs for each parton. The proton-proton crosssections are thus an average of partonic cross-sections, weighted by the PDFs for those partons.

### 2.3 Limitations

Experimental tests of the Standard Model to date have shown no significant deviations from its predictions, across a very wide range of physical processes and energy scales [2]. However, it is quite clear that the Standard Model is not a complete theory of nature. There are many problems and questions that it leaves unanswered.

#### 2.3.1 Fine tuning and the Higgs mass

The Higgs field, which is responsible for the mass of all fundamental fermions and the massive electroweak gauge bosons, was confirmed by the discovery of 125 GeV Higgs boson by ATLAS and CMS in 2012 [3, 4]. The fact that the Higgs mass should be so small is an unsolved mystery of particle physics. When performing QFT calculations, there often arise terms in the perturbation expansion that are formally infinite. A procedure known as renormalization is used in order to tame these divergences. One method of renormalization involves cutting off the divergent integrals at some scale,  $\Lambda$ , then canceling the infinities by redefining some of the free parameters of the theories.

When calculating the physical mass-squared of the Higgs, there are some terms in the series which are proportional to the square of the cutoff scale. The cutoff scale is considered to be some very large energy scale, usually the Planck energy  $10^{19} \ GeV$ . As a result, an infinite sum of positive and negative terms with absolute values on the order of  $10^{38} \ GeV^2$  must cancel out to yield physical Higgs mass-squared, which is on the order of  $10^4 \ GeV^2$ . This kind of cancellation is philosophically unsatisfying, because it seems to be an unreasonable amount of fine-tuning [9].

The fine-tuning is closely related to the so-called "hierarchy problem". Why is gravity so much weaker than the other fundamental forces? That is, why do we observe  $G_F \gg G_N$ ? Since  $G_F \propto 1/M_W^2$ , and  $G_N \propto 1/\Lambda_{Pl}^2$  and because  $M_W$  is proportional to the Higgs vev, the Higgs-mass fine-tuning and the hierarchy problem are really the same thing. A possible way to avoid fine tuning and the hierarchy problem comes from Supersymmetry, which will be covered in chapter 3

#### 2.3.2 Grand unification

The symmetry groups and representations of the Standard Model have great explanatory power, but they are, at heart, axiomatic rather than derived. And the Standard Model contains a large number of free parameters, the values of which are not explained by the model, but must be measured by experiment. It's possible that a simpler model could be found, one which embeds the baroque structure of the Standard Model in a much simpler set of symmetry groups and representations. Such an idea is known as *Grand Unification*. A Grand Unified Theory (GUT) would predict relationships between some of the free parameters of the Standard Model, resulting in a simpler and more elegant theory. This would be analogous to electroweak unification, which explains the relationship between W and Z boson masses and their couplings, and results in a theory with only three free parameters.

A GUT would explain all particle interactions with a single symmetry group. This symmetry would be broken below some energy scale to the  $SU(3)_C \times SU(2)_L \times U(1)_Y$ symmetry of the Standard Model. The details of the symmetry breaking process would yield relations between the strong and electroweak coupling constants [10]. Going a step further, if gravitational interactions could also be incorporated into this single theory, it would be a so-called "Theory of Everything" (TOE). A very popular TOE candidate, known as String Theory, requires the existence of Supersymmetry.

#### 2.3.3 Dark matter

The type of matter explained by the Standard Model is known to account for approximately 20% of the matter in the universe. The remaining 80% exists in the form of dark matter. The exact explanation for dark matter is not yet known, those many of its properties can be inferred. It is massive and invisible, and its primary interaction with visible matter is through gravity [11]. Dark matter is known to be massive, because it interacts gravitationally. This can be seen in its effect on the rotation curves of galaxies, gravitational lensing of distant galaxies, and on the anisotropy of the Cosmic Microwave Background Radiation [12].

Dark matter could possibly be explained by the Standard Model, if it exists in the form of Massive Compact Halo Objects (MACHOs). MACHOs are primordial black holes, formed during the very early universe. If these primordial black holes exist in large enough quantities to account for the majority of dark matter in the universe, they could in principle be detected in gravitational lensing surveys. However, these surveys have so far failed to observe MACHOs in large enough quantities to account for the known quantity of dark matter. In fact, dark matter in the form of MACHOs with masses greater than  $10^{-7}$  solar masses has been ruled out entirely [13].

Another likely explanation for dark matter is that it is composed of a new fundamental particle, one which is not explained by the Standard Model. Many versions of Supersymmetry, described in 3, provide a viable dark matter candidate.

# Chapter 3

# Supersymmetry

Supersymmetry (SUSY) is an extension to the Standard Model, which introduces a new spacetime symmetry relating fermionic and bosonic fields [11]. It is a remarkable theory, which has the potential to resolve many of the known problems of the Standard Model in one fell swoop.

### **3.1** Motivation for new physics

As described in 2.3, the Standard Model leaves open some very important questions. These mysteries all point towards a more fundamental theory, one which is applicable at short distances scales, or equivalently, higher energies. The most important task of this new theory is to explain why the Higgs mass is so small, without resorting to the extreme amount of fine-tuning required by the Standard Model. But a more fundamental theory of particle physics could potentially simplify the symmetry structure of the model and provide an explanation for the seemingly arbitrary quantum numbers of the Standard Model, but rather an extension of it. Whatever the new theory is, it should be applicable at very short distance scales, and at low energies, it must reproduce the experimental evidence gathered so far indicates the Standard Model is a correct theory up to the energy scales at which it has been probed. In this view, the Standard Model is considered an effective field theory (EFT), valid below the scale of new physics,  $\Lambda_{NP}$ .

#### 3.1.1 Naturalness

As explained in 2.3.1, the Standard Model suffers from a problem of extreme fine tuning. Quadratic diverges in the quantum corrections to the Higgs mass-squared require the bare mass of the Higgs to be specified to one part in  $10^{19}$ , which is phenomenally unlikely to happen by chance [14]. The quadratic divergences occur in loop diagrams, the largest con-

tribution coming from the top quark loop, shown in figure 3.1. The correction to the Higgs

t



Figure 3.1: Top-quark loop diagram, the leading correction to the Higgs mass-squared. This contribution is quadratically divergent in the cutoff scale [15].

mass-squared coming from this diagram is [15]:

$$\delta m_h^2 = \frac{3m_t^2}{2\pi^2 v^2} \Lambda_{UV}^2$$
 (3.1)

where  $m_t$  is the top quark mass, v is the Higgs vacuum expectation value, and  $\Lambda_{UV}$  is the ultraviolet cutoff.

Supersymmetry introduces a so-called superpartner for each Standard Model particle. Standard Model fermions have bosonic superpartners, and Standard Model bosons have fermionic superpartners. Superpartners always have the same quantum numbers as their Standard Model partners, except for spin. The details of how this comes about will be discussed in 3.2. The superpartner of the top is called the stop. It's a scalar particle with all the same quantum numbers as the top quark, except for spin. The introduction of superpartners leads to new quantum corrections to the Higgs mass-squared. One of those corrections can be seen in 3.2. The correction to the Higgs mass-squared coming from this



Figure 3.2: Stop loop diagram, the leading SUSY correction to the Higgs mass-squared. This contribution is quadratically divergent in the cutoff scale [15].

diagram is [15]:

$$\delta m_h^2 = -\frac{3m_t^2}{2\pi^2 v^2} \Lambda_{UV}^2$$
 (3.2)

Thus, the quadratically-divergent top-loop correction is cancelled exactly. A similar cancellation occurs for all other quadratic divergences in the calculation, leaving terms that are only logarithmically divergent in the cutoff. In SUSY, the leading correction to the Higgs mass-squared is proportional to the log of the cutoff scale, and the squared difference between the top and stop masses [16]:

$$\delta m_h^2 \propto \left( m_{\tilde{t}}^2 - m_t^2 \right) \Lambda_{UV} \tag{3.3}$$

The amount of fine-tuning required can be quantified by  $m_h/\delta m_h$ , so in order to preserve naturalness, the stop mass cannot be much heavier than the top quark mass. If we allow for fine-tuning of only 1 part in 10, then the stop mass shouldn't be higher than the 1 TeVscale, which indicates that SUSY should be accessible by the LHC. Lower bounds on the masses of various suppersymmetric particles from ATLAS range from a few hundred GeVto 2.4 TeV [17]. Figure 3.3 shows a sample of the SUSY particles and decay channels that have been searched for with ATLAS, and the resulting lower bounds.

#### 3.1.2 Grand unification

Grand Unification is the idea that the  $SU(3)_C \times SU(2)_L \times U(1)_Y$  symmetry of the Standard Model can be embedded in a simpler symmetry group, which constrains the theory at energies above  $\Lambda_{GUT}$ . In a Grand Unified Theory (GUT), there would be no distinction between quarks and leptons at energies above  $\Lambda_{GUT}$ , and the differences in interactions between these two types of matter fields would be explained by the details of the theory and the mechanism by which its symmetry is broken. Furthermore, at energies above  $\Lambda_{GUT}$ , the three forces of the Standard Model would be replaced by a single, unified force. Grand unification would be the logical next step after electroweak unification, in which the electromagnetic and weak forces are shown to be the same force at energies above  $\Lambda_{EW} \approx 246 \ GeV$ . Similarly to how electroweak unification gives a relationship between electric charge and weak isospin, grand unification would give a relationship between color charge and the electroweak quantum numbers. Several different symmetry groups have been proposed as GUTs. Among these are SU(5), SO(10),  $E_6$ , and  $SU(5) \times SU(5)$  [18].

In order for unification to occur, the value of the three Standard Model coupling constants must converge at some energy scale. The running of coupling constants is determined by the renormalization group equation (RGE):

$$\frac{\partial g}{\partial \log \mu} = \beta(g) \tag{3.4}$$

where g is the coupling constant of interest,  $\mu$  is the energy scale at which the coupling is being measured, and  $\beta$  is a function that depends on the gauge structure and matter content of the theory. Using the RGE, and measured values of the coupling constants at specific starting energies, one can calculate the value of the three constants over a large range of energy scales. For the Standard Model, there is no energy scale at which all three Standard Model

M	arch 2019	ches .	- 95 /		. LU							$\sqrt{s} = 13 \text{ TeV}$
	Model	Si	gnatur	e ∫.	L dt [fb <sup>-</sup>	<sup>1</sup> ] Ma:	ss limit					Reference
S	$\tilde{q}\tilde{q}, \tilde{q} \rightarrow q \tilde{\chi}_{1}^{0}$	0 e, µ mono-jet	2-6 jets 1-3 jets	$E_T^{miss}$ $E_T^{miss}$	36.1 36.1	<ul> <li> <i>q</i> [2x, 8x Degen.]         <i>q</i>         [1x, 8x Degen.]         </li> </ul>	0.43	0.9 0.71	1.55		$m(\bar{k}_1^0) < 100 \text{ GeV}$ $m(\bar{q})-m(\bar{k}_1^0)=5 \text{ GeV}$	1712.02332 1711.03301
clusive Searche	$\tilde{g}\tilde{g}, \tilde{g} \rightarrow q \bar{q} \tilde{\chi}_1^0$	0 <i>e</i> , <i>µ</i>	2-6 jets	$E_T^{\rm miss}$	36.1	Ř Ř		Forbidden	0.95-1.6	2.0	m(𝔅̃ <sup>0</sup> <sub>1</sub> )<200 GeV m(𝔅̃ <sup>0</sup> <sub>1</sub> )=900 GeV	1712.02332 1712.02332
	$\tilde{g}\tilde{g}, \tilde{g} \rightarrow q\bar{q}(\ell\ell)\tilde{\chi}_{1}^{0}$	3 e,μ ee,μμ	4 jets 2 jets	$E_T^{miss}$	36.1 36.1	ğ ğ			1.2	85	m(𝔅1)<800 GeV m(𝔅)-m(𝔅1)=50 GeV	1706.03731 1805.11381
	$\tilde{g}\tilde{g}, \tilde{g} \rightarrow qqWZ\tilde{\chi}_{1}^{0}$	0 e,μ 3 e,μ	7-11 jets 4 jets	$E_T^{\rm miss}$	36.1 36.1	ğ ğ		0.98	1.	.8	$m(\tilde{\chi}_1^0) < 400 \text{ GeV}$ $m(\tilde{g})-m(\tilde{\chi}_1^0)=200 \text{ GeV}$	1708.02794 1706.03731
Ē	$\tilde{g}\tilde{g}, \tilde{g} \rightarrow t t \tilde{\chi}_1^0$	0-1 e,μ 3 e,μ	3 b 4 jets	$E_T^{\rm miss}$	79.8 36.1	Ĩ Ĩ			1.25	2.25	m( $\tilde{k}_1^0$ )<200 GeV m( $\tilde{g}$ )-m( $\tilde{k}_1^0$ )=300 GeV	ATLAS-CONF-2018-041 1706.03731
	$\tilde{b}_1 \tilde{b}_1, \tilde{b}_1 \rightarrow b \tilde{\chi}_1^0 / t \tilde{\chi}_1^{\pm}$		Multiple Multiple Multiple		36.1 36.1 36.1	δ <sub>1</sub> Forbidden δ <sub>1</sub> δ <sub>1</sub>	Forbidden Forbidden	0.9 0.58-0.82 0.7		$m(\tilde{\chi}_{1}^{0})=200$	$m(\tilde{k}_1^0)=300 \text{ GeV}, BR(b\tilde{k}_1^0)=1$ =300 GeV, BR(b\tilde{k}_1^0)=BR(b\tilde{k}_1^\pm)=0.5 GeV, $m(\tilde{k}_1^\pm)=300 \text{ GeV}, BR(b\tilde{k}_1^\pm)=1$	1708.09266, 1711.03301 1708.09266 1706.03731
larks ction	$\tilde{b}_1 \tilde{b}_1, \tilde{b}_1 {\rightarrow} b \tilde{\chi}^0_2 {\rightarrow} b h \tilde{\chi}^0_1$	0 $e, \mu$	6 b	$E_T^{\rm miss}$	139		0.23-0.48	c	.23-1.35	Δm( <sup>j</sup> Δr	$\hat{\chi}_{2}^{0}, \hat{\chi}_{1}^{0} = 130 \text{ GeV}, m(\hat{\chi}_{1}^{0}) = 100 \text{ GeV} $ $n(\hat{\chi}_{2}^{0}, \hat{\chi}_{1}^{0}) = 130 \text{ GeV}, m(\hat{\chi}_{1}^{0}) = 0 \text{ GeV} $	SUSY-2018-31 SUSY-2018-31
n. squ produ	$\tilde{t}_1 \tilde{t}_1, \tilde{t}_1 \rightarrow W b \tilde{\chi}_1^0$ or $t \tilde{\chi}_1^0$ $\tilde{t}_1 \tilde{t}_1$ , Well-Tempered LSP	0-2 e, µ 0-	-2 jets/1-2 Multiple	$b E_T^{miss}$	36.1 36.1	<i>ī</i> 1 <i>ī</i> 1		1.0 0.48-0.84		$m(\bar{\chi}_{1}^{0})=150$	$m(\tilde{\chi}_1^0)=1 \text{ GeV}$ GeV, $m(\tilde{\chi}_1^{\pm})-m(\tilde{\chi}_1^0)=5 \text{ GeV}$ , $\tilde{r}_L \approx \tilde{r}_L$	1506.08616, 1709.04183, 1711.11520 1709.04183, 1711.11520
' ge	$\tilde{t}_1 \tilde{t}_1, \tilde{t}_1 \rightarrow \tilde{\tau}_1 b \nu, \tilde{\tau}_1 \rightarrow \tau \tilde{G}$	$1\tau+1e_{*}\mu,\tau$	2 jets/1 b	$E_T^{miss}$	36.1	ĩ <sub>1</sub>			.16		m(Ť <sub>1</sub> )=800 GeV	1803.10178
di 3	$\tilde{t}_1 \tilde{t}_1, \tilde{t}_1 \rightarrow c \tilde{\chi}_1^0 / \tilde{c} \tilde{c}, \tilde{c} \rightarrow c \tilde{\chi}_1^0$	0 е, µ 0 е и	2 c mono-iet	$E_T^{miss}$ $F^{miss}$	36.1 36.1	č 7, 7,	0.46	0.85			$m(\tilde{k}_{1}^{0})=0 \text{ GeV}$ $m(\tilde{t}_{1},\tilde{c})-m(\tilde{k}_{1}^{0})=50 \text{ GeV}$ $m(\tilde{c},\tilde{c})-m(\tilde{k}_{1}^{0})=5 \text{ GeV}$	1805.01649 1805.01649 1711.03301
	$\tilde{t}_2 \tilde{t}_2, \tilde{t}_2 \rightarrow \tilde{t}_1 + h$	1-2 e, µ	4 b	$E_T^{miss}$	36.1	ĩ <sub>2</sub>	0.40	0.32-0.88		m(ž	$m(i_1, e) - m(\tilde{\chi}_1^0) = 0 \text{ GeV}$ $m(i_1, e) - m(\tilde{\chi}_1^0) = 180 \text{ GeV}$	1706.03986
	$\tilde{\chi}_1^{\pm} \tilde{\chi}_2^0$ via WZ	2-3 e,μ ee,μμ	≥ 1	$E_T^{miss}$ $E_T^{miss}$	36.1 36.1	$ \frac{\tilde{\chi}_{1}^{*}/\tilde{\chi}_{2}^{0}}{\tilde{\chi}_{1}^{*}/\tilde{\chi}_{2}^{0}} = 0.17 $		0.6			$m(\tilde{\chi}_{1}^{0})=0$ $m(\tilde{\chi}_{1}^{0})=m(\tilde{\chi}_{1}^{0})=10 \text{ GeV}$	1403.5294, 1806.02293 1712.08119
	$\tilde{\chi}_{1}^{\pm} \tilde{\chi}_{1}^{\mp}$ via WW	2 e, µ		$E_T^{miss}$	139	$\tilde{X}_{1}^{\pm}$	0.42				$m(\bar{\chi}_{1}^{0})=0$	ATLAS-CONF-2019-008
	$\tilde{\chi}_1^{\pm} \tilde{\chi}_2^0$ via Wh	0-1 e, µ	2 b	$E_T^{miss}$	36.1	$\tilde{\chi}_1^{\pm}/\tilde{\chi}_2^0$		0.68			$m(\tilde{\chi}_1^0)=0$	1812.09432
si ∠	$\tilde{\chi}_{1}^{\pm} \tilde{\chi}_{1}^{\mp}$ via $\tilde{\ell}_{L} / \tilde{v}$	2 e, µ		$E_T^{miss}$	139	$\tilde{X}_{1}^{\pm}$		1.0			$m(\tilde{\ell}, \tilde{\nu})=0.5(m(\tilde{\ell}_{1}^{\pm})+m(\tilde{\ell}_{1}^{0}))$	ATLAS-CONF-2019-008
dire	$\tilde{\chi}_1^{\pm} \tilde{\chi}_1^{+} / \tilde{\chi}_2^{0}, \tilde{\chi}_1^{+} \rightarrow \tilde{\tau}_1 \nu(\tau \tilde{\nu}), \tilde{\chi}_2^{0} \rightarrow \tilde{\tau}_1 \tau(\nu \tilde{\nu})$	2 τ		$E_T^{miss}$	36.1	$ \tilde{\chi}_{1}^{*}/\tilde{\chi}_{2}^{*} $ $ \tilde{\chi}_{1}^{*}/\tilde{\chi}_{2}^{0} $ 0.22		0.76	m	$(\tilde{t}_{1}^{\pm})-m(\tilde{t}_{1}^{0})=10$	$\tilde{\chi}_{1}^{"} = 0, m(\tilde{\tau}, \tilde{\nu}) = 0.5(m(\tilde{\chi}_{1}^{n}) + m(\tilde{\chi}_{1}^{"}))$ $0 \text{ GeV}, m(\tilde{\tau}, \tilde{\nu}) = 0.5(m(\tilde{\chi}_{1}^{n}) + m(\tilde{\chi}_{1}^{"}))$	1708.07875 1708.07875
	$\tilde{\ell}_{L,R} \tilde{\ell}_{L,R}, \tilde{\ell} \rightarrow \ell \tilde{\chi}_1^0$	2 e,μ 2 e,μ	0 jets ≥ 1	$E_T^{miss}$ $E_T^{miss}$	139 36.1	₹ ₹ 0.18		0.7			$m(\tilde{\ell}_{1}^{0})=0$ $m(\tilde{\ell})-m(\tilde{\ell}_{1}^{0})=5 \text{ GeV}$	ATLAS-CONF-2019-008 1712.08119
	$\tilde{H}\tilde{H}, \tilde{H} \rightarrow h\tilde{G}/Z\tilde{G}$	0 e,μ 4 e,μ	$\ge 3 b$ 0 jets	$E_T^{miss}$ $E_T^{miss}$	36.1 36.1	Й 0.13-0.23 Й 0.3		0.29-0.88			$BR(\tilde{\chi}_1^0 \rightarrow h\tilde{G})=1$ $BR(\tilde{\chi}_1^0 \rightarrow Z\tilde{G})=1$	1806.04030 1804.03602
lived	Direct $\tilde{\chi}_1^+ \tilde{\chi}_1^-$ prod., long-lived $\tilde{\chi}_1^\pm$	Disapp. trk	1 jet	$E_T^{\rm miss}$	36.1	$ \tilde{\tilde{X}}_{1}^{\pm} \\ \tilde{\tilde{X}}_{1}^{\pm} = 0.15 $	0.46				Pure Wino Pure Higgsino	1712.02118 ATL-PHYS-PUB-2017-019
arti	Stable g R-hadron		Multiple		36.1	ğ				2.0		1902.01636,1808.04095
P	Metastable $\tilde{g}$ R-hadron, $\tilde{g} \rightarrow qq \tilde{\chi}_1^0$		Multiple		36.1	$\tilde{g} = [\tau(\tilde{g}) = 10 \text{ ns}, 0.2 \text{ ns}]$				2.05 2.4	m( $\tilde{k}_{1}^{0}$ )=100 GeV	1710.04901,1808.04095
	LFV $pp \rightarrow \tilde{v}_{\tau} + X$ , $\tilde{v}_{\tau} \rightarrow e\mu/e\tau/\mu\tau$	$e\mu,e\tau,\mu\tau$			3.2	$\tilde{\nu}_{\rm T}$				1.9	$\lambda'_{311}$ =0.11, $\lambda_{132/133/233}$ =0.07	1607.08079
	$\tilde{\chi}_{1}^{\pm}\tilde{\chi}_{1}^{\mp}/\tilde{\chi}_{2}^{0} \rightarrow WW/Z\ell\ell\ell\ell\nu\nu$	4 e, µ	0 jets	$E_T^{miss}$	36.1	$\tilde{\chi}_{1}^{\pm}/\tilde{\chi}_{2}^{0} = [\lambda_{i33} \neq 0, \lambda_{12k} \neq 0]$		0.82	1.33		m( $\tilde{\chi}_{1}^{0}$ )=100 GeV	1804.03602
>	$\tilde{g}\tilde{g}, \tilde{g} \rightarrow qq\tilde{\chi}'_1, \tilde{\chi}'_1 \rightarrow qqq$	4-5	5 large- <i>R</i> je Multiple	ets	36.1 36.1	ğ [m(𝑋" <sub>1</sub> )=200 GeV, 1100 GeV] ğ [𝑌",=2e-4, 2e-5]		1.0	1.3	2.0	Large $\lambda_{112}^{\prime\prime}$ m $(\tilde{\lambda}_{1}^{0})$ =200 GeV bins like	1804.03568 ATLAS-CONE-2018-003
d b	$\tilde{t} \tilde{t} \rightarrow t \tilde{\tilde{V}}^0 \tilde{\tilde{V}}^0 \rightarrow the$		Multiple		36.1		0.5	5 10	5		m(x <sup>0</sup> )=200 GeV bino like	ATLAS-CONE-2018-003
	$\tilde{t}_1 \tilde{t}_1, \tilde{t}_1 \rightarrow bs$	2	2 jets + 2 b	,	36.7	$\tilde{t}_1  [qq, bs]$	0.42	0.61			(i ) = 200 Gev, birD-like	1710.07171
	$\tilde{t}_1\tilde{t}_1, \tilde{t}_1 \rightarrow q\ell$	2 e,μ 1 μ	2 <i>b</i> DV		36.1 136		<3e-9]	1.0	0.4-1.45 1.6		$BR(\tilde{t}_1 \rightarrow be/b\mu) > 20\%$ $BR(\tilde{t}_1 \rightarrow q\mu) = 100\%$ , $\cos\theta_t = 1$	1710.05544 ATLAS-CONF-2019-006
*Only pher	a selection of the available mas omena is shown. Many of the l	ss limits on n imits are bas	ew state sed on	s or	1	0-1					Mass scale [TeV]	

#### ATLAS SUSY Searches\* - 95% CL Lower Limits

Figure 3.3: A representative sample of the various SUSY searches performed by ATLAS, and their resulting mass lower bounds [17].

constants converge, as would be required for a grand unified theory. However, when the  $\beta$  function is modified to include supersymmetric particles, it is possible for all three Standard Model forces to unify. In one particular SUSY model, known as the Minimal Supersymmetric Standard Model (MSSM), unification of the forces occurs at  $\mu \approx 2 \times 10^{16} \text{ GeV}$  [18]. The running of the coupling constants under the Standard Model and MSSM can be seen in figure 3.4.

#### 3.1.3 Dark matter

As mentioned in 2.3.3, SUSY can provide an explanation for Dark Matter, which comprises approximately 80% of matter in the universe, but cannot be explained by any known particle. In SUSY models that conserve R-parity<sup>1</sup>, the lightest supersymmetric particle (LSP)

ATLAC Proliminary

<sup>&</sup>lt;sup>1</sup>See section 3.3 for a discuss of R-parity and R-parity violation



Figure 3.4: Running of the three Standard Model coupling constants under the Standard Model (left) and MSSM (right) RGEs. For the MSSM, unification occurs at  $\mu \approx 2 \times 10^{16} \text{ GeV}$  [16].

must be stable because R-parity prevents the LSP from decaying to a final state without SUSY particles, and conservation of energy prevents it from decaying to a final state with SUSY particles. In so-called "natural" SUSY, the LSP is a neutralino,  $\chi_0^1$ . The neutralino is a mixture of the electroweak gauge superpartners, the wino and bino, and the Higgs superpartner, the higgsino. The neutralino in this theory would be stable, massive, electrically neutral, and it would interact via the weak force. As such, it is a viable candidate for WIMP dark matter.

Intriguingly, if one assumes that dark matter was thermally produced in the early universe, and calculates the required production cross-section to match the dark matter abundance measured today, the result is  $\langle \sigma v \rangle \approx 3 \times 10^{-26} \ cm^3/s$ , which is very close to what would be expected for a weakly-interacting LSP [19].

### **3.2** Theory and phenomenology

Supersymmetry starts by positing a transformation that converts a boson field to a fermion field and vice versa, and a Lagrangian which is invariant under such transformations. Fields are represented in supermultiplets, each of which always contains a Standard Model particle and its superpartner. Standard Model bosons are always paired with fermionic superpartners, and Standard Model fermions are always paired with bosonic superpartners. SUSY transformations commute with all symmetries of the Standard Model, except for the Lorentz transformations. As a result, a Standard Model particle has all the same quantum numbers as its superpartner, except for spin. Additionally, SUSY requires both members of the supermultiplet to have the same mass. Experiments have ruled out the existence of same-mass superpartners for all Standard Model particles, so for SUSY to be compatible with experimental evidence, it must be broken. The method by which SUSY is broken should have an impact on the phenomenology of the model, specifically on the differences in mass between the Standard Model particles and their superpartners. A framework called soft SUSY breaking allows for the specifics of the SUSY breaking to be factored out, by adding effective terms to the SUSY Lagrangian which account for the consequences of SUSY breaking without specifying the mechanism. These new terms break SUSY explicitly, and so this is an effective field theory, valid only at energies well below the SUSY-breaking scale [20].

#### **3.2.1** Superfields and superpotentials

Superfields can be categorized into *chiral* and *gauge* superfields. Standard model fermions and their superpartners will belong to chiral supermultiplets, while Standard Model gauge bosons and their superpartners will belong to gauge supermultiplets. A generic chiral superfield can be represented as  $\Phi(x, \theta)$ , where x represents spacetime, and  $\theta$  denotes the additional two fermionic degrees of freedom needed for supersymmetry transformations. It can alternatively be represented as  $\Phi = (\phi, \psi)$ , where  $\phi$  is the complex scalar component, and  $\psi$  is the fermion component. A generic vector superfield can be represented at  $V(x, \theta, \overline{\theta})$ , where  $\overline{\theta}$  are the conjugate degrees of freedom to  $\theta$ . In component form, vector supermultiplets can be represented as  $V = (A^{\mu}, \lambda)$ , where  $A^{\mu}$  is the gauge boson and  $\lambda$  is the fermionic superpartner.

The generic supersymmetric action can then be written as [18]:

$$S = \int d^4x \int d^2\theta d^2\bar{\theta} \Phi^{\dagger} e^V \Phi + \int d^4x \int d^2\theta \left( W(\Phi) + W_{\lambda}(V)W_{\lambda}(V) \right) + h.c.$$
(3.5)

where the first integrand is the kinetic term for the matter fields, and the second integrand contains the generic superpotential,  $W(\Phi)$  and the gauge kinetic term  $W_{\lambda}(V)W_{\lambda}(V)$ . The function  $W_{\lambda}(V)$  is defined as:

$$W_{\lambda}(V) = (D)^2 \bar{\mathcal{D}} V \tag{3.6}$$

where  $\mathcal{D} \equiv \partial_{\theta} - i\sigma \dot{\partial}_x$  [18].

#### 3.2.2 SUSY particles

In the MSSM, the minimum number of new particles and free parameters are added to the already large assortment of Standard Model ones. Each Standard Model fermion is paired with a spin-0 superpartner, and each Standard Model gauge boson is paired with a spin-1/2 superpartner. For the Higgs boson, introducing a single superpartner is not enough. There must be at least two Higgs chiral supermultiplets, with hypercharge values 1/2 and -1/2. This is needed in order to keep the electroweak theory anomaly-free, and for the Higgs mechanism to still give mass to all the fermions [15].

Superpartner states will be denoted with a tilde, and are also referred to as sparticles. For example, the superpartner of the electron, called the selectron, will be denoted as  $\tilde{e}$ .

Names		spin 0	spin $1/2$	$SU(3)_C, SU(2)_L, U(1)_Y$
squarks, quarks	Q	$(\widetilde{u}_L \ \widetilde{d}_L)$	$(u_L \ d_L)$	$({f 3},{f 2},{1\over 6})$
$(\times 3 \text{ families})$	$\overline{u}$	$\widetilde{u}_R^*$	$u_R^\dagger$	$(\overline{f 3},{f 1},-{2\over3})$
	$\overline{d}$	$\widetilde{d}_R^*$	$d_R^\dagger$	$(\overline{f 3},{f 1},{1\over 3})$
sleptons, leptons	L	$(\widetilde{ u} \ \widetilde{e}_L)$	$( u \ e_L)$	$( {f 1}, {f 2}, -{1\over 2})$
$(\times 3 \text{ families})$	$\overline{e}$	$\widetilde{e}_R^*$	$e_R^\dagger$	(1, 1, 1)
Higgs, higgsinos	$H_u$	$\begin{pmatrix} H_u^+ & H_u^0 \end{pmatrix}$	$(\widetilde{H}^+_u \ \widetilde{H}^0_u)$	$({f 1},{f 2},+{1\over 2})$
	$H_d$	$ (H^0_d \ H^d) $	$(\widetilde{H}^0_d \ \widetilde{H}^d)$	$( {f 1}, {f 2}, -{1\over 2})$

The MSSM chiral superfield content is summarized in table 3.1, and gauge superfields are summarized in table 3.2.

Table 3.1: The MSSM chiral superfields, including their names, symbols, and components. Quantum numbers for the Standard Model symmetry group transformations are also given [15].

Names	spin $1/2$	spin 1	$SU(3)_C, SU(2)_L, U(1)_Y$
gluino, gluon	$\widetilde{g}$	g	(8, 1, 0)
winos, W bosons	$\widetilde{W}^{\pm}$ $\widetilde{W}^{0}$	$W^{\pm} W^0$	(1, 3, 0)
bino, B boson	$\widetilde{B}^0$	$B^0$	(1, 1, 0)

Table 3.2: The MSSM gauge superfields, including their names, symbols, and components. Quantum numbers for the Standard Model symmetry group transformations are also given [15].

The MSSM superpotential is:

$$W = h_l^{ij} e_i^c L_j H_d + h_d^{ij} Q_i d_j^c H_d + h_u^{ij} Q_i u_j^c H_u + \mu H_u H_d$$
(3.7)

where the indices i, j, k are generation indices. The superpotential includes all renormalizable terms that preserve the gauge symmetries and are holomorphic in the chiral superfields,
except for terms that violate baryon or lepton number conservation. The superpotential is expressed here only in terms of the left-handed superfields and components. The hermitian conjugate terms are implied.

# 3.3 R-Parity and R-parity violation

In equation 3.7, certain terms were excluded from the superpotential because they violate either baryon number or lepton number conservation. But this requirement is in fact overly constraining. While processes that violate baryon number or lepton number conservation have never been observed experimentally, neither have they been ruled out. Additionally, while there are no explicit B or L-violating terms in the Standard Model, and no perturbative process in the Standard Model violates either, there are in fact non-perturbative effects in the Standard Model which lead to violations of both [21].

## 3.3.1 R-parity

R-parity is a symmetry which is introduced in order to eliminate the explicit B and Lviolating terms from the theory, without have to postulate that B and L are individually conserved. Each particle has an R-Parity quantum number, defined as:

$$P_R = (-1)^{3(B-L)+2s} \tag{3.8}$$

where B is the particle's baryon number, L is its lepton number, and s is its spin. Quarks and squarks carry baryon number 1/3, leptons an sleptons carry lepton number 1, and their corresponding antiparticles carry baryon number -1/3 and lepton number -1. Every Standard Model particle has  $P_R = +1$ , and every sparticle has  $P_R = -1$ .

A necessary consequence of R-Parity conservation is that the lightest supersymmetric particle (LSP) must be stable. In an R-parity-conserving (RPC) theory, interaction vertices that involve Standard Model particles cannot involve an odd number of sparticles. As a result, when a sparticle decays, there must be at least one sparticle amongst the decay products, along with any Standard Model particles. So the lightest sparticle cannot decay, since there is no lighter sparticle for it to decay to. Furthermore, in RPC SUSY, sparticles can only be produced in pairs at collider experiments, since their production vertices necessarily include Standard Model particles.

## 3.3.2 R-parity violating superpotential terms

If R-Parity conservation is not required, additional terms must be added to the MSSM superpotential. Each term explicitly violates either baryon number or lepton number by one unit. These R-Parity-violating (RPV) superpotential terms are:

$$W_{RPV} = \frac{1}{2}\lambda_{ijk}L_iL_j\bar{e}_k + \lambda'^{ijk}L_iQ_j\bar{d}_k + \mu'^iL_iH_u + \frac{1}{2}\lambda''^{ijk}\bar{u}_i\bar{d}_j\bar{d}_k$$
(3.9)

where  $\mu'$ ,  $\lambda$ ,  $\lambda'$ , and  $\lambda''$  are new coupling constants. The first three terms result in interaction vertices with  $\Delta L = 1$ , and the last term has  $\Delta B = 1$ . The term with the coupling constant  $\lambda''_{ijk}$  gives rise to the signal processes under consideration in this analysis. The coupling constant has three generation indices, and must be antisymmetric in the last two terms in order to preserve gauge invariance.

# 3.3.3 Rapid proton decay

In general, RPV SUSY would result in proton decay rates far higher than experimental upper bounds. The main process leading to rapid proton decay requires both the  $\lambda'$  and  $\lambda''$  terms from equation 3.9. In this process, the proton decays to a neutral pion and a positron, via an off-shell squark. The tree-level diagram leading to such a decay is shown in figure 3.5 The tree-level rate for this process is approximately [15]:



Figure 3.5: Tree-level diagram for the decay  $p^+ \to \pi^0 e^+$ , with both L-violating and B-violating vertices [15].

$$\Gamma_{p^+ \to \pi^0 e^+} \sim \frac{m_p^5}{m_{\tilde{d}}^4} \sum_{i=2,3} \left| \lambda'^{11i} \lambda''^{11i} \right|^2 \tag{3.10}$$

Experimental bounds on the proton lifetime are greater than  $10^{23}$  years, so in order for RPV SUSY to be consistent with experiment, either the squark masses have to be extremely large, or one of the couplings  $\lambda'$ ,  $\lambda''$  must be extremely small. Specifically, the coupling and squark masses must satisfy [22]:

$$\lambda_{11k}' \lambda_{11k}'' < 10^{-23} \left(\frac{m_{\tilde{q}}}{100 \; GeV}\right)^2 \tag{3.11}$$

For the search presented here, the ad-hoc assumption that  $\lambda' = 0$  is made, in order to prevent predicted proton decay rates inconsistent with experiment.

# 3.3.4 R-parity violating gluino decays

The  $\lambda''_{ijk}$  RPV term in the superpotential allows for two potential gluino discovery channels at the LHC. The first involves pair-produced gluinos each decaying to three quarks via an effective vertex with an off-shell squark propagator. This decay mode will be referred to as the direct decay model. The second decay mode has the gluino first decaying to two quarks and an on-shell neutralino, via an RPC vertex, and then the neutralino decaying to thre quarks through an effective RPV vertex, which also includes an off-shell squark propagator. This is the cascade decay model. Both decay modes result in a large number of high-momentum jets in the event. Diagrams for these two decay modes are shown in figure 3.6. As mentioned in 3.3.2, the generation indices on  $\lambda''_{ijk}$  must be antisymmetric in the last two terms. The values of  $\lambda''$  for different combinations of i, j, k will determine the branching fractions for the gluinos decaying to different quark flavors.



Figure 3.6: Diagrams for the two decay processes that are the subject of this search. The direct decay (left) and cascade decay (right) both involve effective RPV vertices containing off-shell squark propagators.

The production and decay rates will depend on the values of the gluino mass, neutralino mass, and  $\lambda''$ . A detailed discussion of the signal modeling is presented in chapter 7.

# Chapter 4

# The LHC and ATLAS Detector

ATLAS is a system of particle detectors built to measure collisions of both proton-proton and heavy ion collisions generated by the LHC [23]. The full detector is 44 m long and 25 m in diameter. It consists of an inner detector subsystem for charged particle tracking and electron identification, electromagnetic and hadronic calorimeters, and a muon spectrometer.

# 4.1 The Large Hadron Collider

The Large Hadron Collider (LHC) is the world's largest, highest-energy, highest-luminosity particle accelerator. It is located inside a 27 km circular underground tunnel below the French-Swiss border. The LHC was designed to produce up to 14 TeV center-of-mass energy proton-proton collisions, as well as lead-ion collisions at a center-of-mass energy of 2.7 TeV/u. There are two counter-rotating beams of particles, which are accelerated to nearly the speed of light and steered into each other at predefined collision points along the ring, where detectors have been built to measure the particles generated in the collisions. The highest collision energy achieved to date for proton-proton collisions is 13 TeV, with a peak luminosity of  $10^{34} \ cm^{-2}s^{-1}$  [24].

Collisions generated by the LHC are measured by seven different experiments. The original four experiments are ATLAS, CMS, LHCb, and ALICE . ATLAS and CMS are the two general-purpose detectors designed for a variety of measurements and searches for physics beyond the Standard Model. They are mainly used to study proton-proton collisions, but are also used for heavy-ion collisions. LHCb, as the name implies, is a specialized detector built to measure the decays of B mesons. ALICE is a specialized detector for studying Pb - Pb collisions. In addition to these four experiments, there are three smaller, specialized experiments called TOTEM, LHCf, and MoEDAL .

The LHC is actually the last in a chain of progressively larger accelerators, each of which adds additional energy to the proton beams until they are injected into the LHC for the final acceleration. Atoms from a bottle of Hydrogen gas are passed through an electric field to strip their electrons, resulting in bare protons. These bare protons are passed into a linear accelerator known as Linac2, which accelerates them to 50 MeV. The Proton Synchrotron Booster (PSB) then accelerates the beams to 1.4 GeV, before injecting them into the Proton Synchrotron (PS), which accelerates them to 25 GeV. The final link in the injection chain before the LHC is the Super Proton Synchrotron, which accelerates the beams up to 450 GeVper proton. The LHC provides the final boost up to 6.5 TeV per proton. This final stage of acceleration takes approximately 20 minutes, after which the beams can circulate at this energy for at least 10 hours.

The LHC did not always operate at 13 TeV. It took a number of years, over which the energy was increased in steps, before this energy could be reached. The cumulative luminosity delivered over time each year from 2011 to 2018 is shown in figure 4.1.



Figure 4.1: Cumulative luminosity delivered by the LHC over time for each of the years from 2011 to 2018 [25].

The first two years of data taking, during which the LHC operated at  $\sqrt{s} = 7 \ TeV$  and later at  $\sqrt{s} = 8 \ TeV$  are known as Run 1. The first long shutdown (LS1) took place during 2013 and 2014, during which the accelerator was upgraded needed to reach  $\sqrt{s} = 13 \ TeV$ . The most time-consuming part of the upgrade involved cycles of cooling and quenching of the superconducting dipole magnets, referred to as "training" the magnets. The LHC restarted for Run 2 in early 2015, operating at  $\sqrt{s} = 13 \ TeV$ , with a goal of delivering 150  $fb^{-1}$  of proton-proton collision data over a four-year period. After Run 2, the LHC will be shut down again for 2 years to enable the upgrade to the full 14 TeV design energy. In 2024, it will be shut down again to begin work on the upgrades needed for the high-luminosity LHC (HL-LHC), which aims to increase the instantaneous luminosity by a factor of five [26]. Figure 4.2 shows the planned operation timeline of the LHC from the first collisions in 2011 through to the HL-LHC upgrade in the 2020s.



Figure 4.2: Planned operation timeline of the LHC, showing the different run periods, shutdown periods, and the energy and luminosity goals, up to and including the HL-LHC upgrades [26].

## 4.1.1 Layout

The LHC was built inside an already existing tunnel, which originally housed a previous accelerator called the Large Electron-Positron Collider (LEP). The tunnel is roughly circular and 27 km in circumference. It is buried at an average depth of 100 m below ground, sloping at a gradient of 1.4% from its deepest point of 175 m to its shallowest at 50 m below ground.

The ring consists of eight arcs and eight straight regions. The straight regions are referred to as insertions, and the arcs are called sectors. Octants span from the halfway point of one sector to the halfway point of the next sector, and each contains contain exactly one insertion. The sectors contain dipole magnets which generate the field needed to bend the beams around the ring. Four of the insertions are used for particle collisions, where the two beams are focused with quadrupole magnets in order to collide the greatest number of particles in the smallest possible space. Each of the four main detector experiments are located at one of these insertions. The other four insertions are used for beam injection, cleaning, and dumping. Figure 4.3 shows a schematic of the layout of the LHC, including locations of the four major detector experiments.

There are also three smaller experiments, not shown in the figure, called TOTEM, LHCf, and MoEDAL. TOTEM and LHCf are specialized for measuring protons and heavy ions

emerging from LHC collisions at extremely small angles. The TOTEM detectors are located close to the interaction point along beamline on either side of the CMS experiment, and LHCf detectors are similarly situated near the ATLAS collision point. MoEDAL detectors are located within the LHCb insertion, and are designed to search for evidence of magnetic monopoles.



Figure 4.3: Schematic of the LHC layout. The clockwise-rotating beam, shown in red, is referred to as beam 1, and the counter-clockwise-rotating beam, shown in blue, is beam 2. Locations of the four major detector experiments are labelled [27].

### 4.1.2 Magnets

#### **Dipoles**

Each of the eight sectors uses 154 dipole magnets to bend the beams around the ring. All dipole magnets in a sector are connected in series and cooled in the same continuous crystotat. Each sector is powered independently. The magnetic field generated by these magnets points in the y-direction in order to generate a force that points towards the center of the LHC, according to the Lorentz force law  $\vec{F} = q \left(\vec{E} + \vec{v} \times \vec{B}\right)$ .

The design and performance of the dipole magnets is crucial to achieving the high energies needed by the LHC physics goals. This is because the maximum beam energy that can be kept in orbit is proportional to the strength of the magnetic field generated by the dipole magnets. The dipole magnets are superconducting electromagnets, which generate a field of 8.3 T. The superconducting wires are made of NbTi, which has a critical temperature of 10 K, but are cooled to and operated at 1.9 K. Cooling is done with superfluid helium, which has a very high heat capacity at this temperature. The wires are made of 7  $\mu m$ filaments, thousands of which are twisted together into 15 mm strands. Each cable is then composed of 36 of these strands twisted together. In order to generate the 8.3 T magnetic field, a current of 11.85 kA passes through the superconducting wires. Each dipole magnet is 15 m long and weighs 35 tonnes. A schematic of the cross-section of an LHC dipole magnet can be seen in figure 4.4.

#### Quadrupoles and higher order

Quadrupole magnets are located near the interaction points of each detector experiment, for focusing the beams into the smallest possible area before collisions. When a beam passes through a quadrupole magnet, it is squeezed along one axis perpendicular to the direction of travel of the beam. It is simultaneously un-squeezed in the direction perpendicular to both the beam and the squeezing direction, but the magnitude of the un-squeezing is less than the magnitude of squeezing. When a beam passes through two successive quadrupole magnets with squeezing directions at 90 degrees to each other, the net effect is a uniform squeezing of the beam towards its center. Before entering ATLAS, a series of quadrupole magnets squeezes the beams from approximately  $0.2 \ mm$  to less than  $20 \ \mum$ . Additional higher-order magnets are used in locations around the LHC for beam corrections and damping of oscillations. There are also special single-bore dipoles which are used to separate the beams in the region of the RF cavities, so that separate RF cavities can be used for each beam.

### 4.1.3 Radio frequency cavities

Superconducting radio frequency (RF) cavities, operating at 400 MHz, and 2 MV, are used to accelerate and store the beams. The RF cavities accelerate the beams to their nominal energy, and continue to supply power over the lifetime of the beams to compensate for energy



# LHC DIPOLE : STANDARD CROSS-SECTION

Figure 4.4: Cross-section schematic of an LHC dipole magnet [28].

lost through synchrotron radiation. There are a total of 16 RF cavities, contained within 4 cryomodules. Figure 4.5 is a schematic of an RF cavity cryomodule.

The RF cavities are tuned such that the resonant frequency of electromagnetic waves inside the cavity is 400 *MHz*. Protons passing through the cavity will be accelerated or decelerated by the electromagnetic field, depending on their time of arrival in the cavity. A proton travelling with the exactly the right energy will enter the cavity with just the right timing to experience zero overall force. Protons travelling slightly too slowly or too quickly will be decelerated or accelerated until their energy is exactly right. This process results in protons bunching up around the beam, with all protons traveling at nearly the same 13 *TeV* energy. The energy spread (rms  $\delta p/p$ ) for the LHC is approximately  $1 \times 10^{-4}$  [11].

# 4.1.4 Beams and luminosity

Each beam consists of 2,808 bunches of  $10^{11}$  protons each. The frequency of cycles around the LHC is 11,245 Hz. Bunches are mostly evenly spaced, with a few gaps that are needed



Figure 4.5: Schematic of a cryomodule, containing four RF cavities used to accelerate and store the LHC beam [27].

for beam injection or dumping. Except for the gaps, bunches are spread out to arrive at the collision points every 25 *ns*. When the gaps are taken into account, actual bunches cross with an average frequency of 30 *MHz*. At a luminosity of  $7.7 \times 10^{33} \text{ cm}^{-2} \text{s}^{-1}$ , there will be approximately 40 inelastic collisions per bunch crossing. Thus the average number of events delivered to ATLAS per second is approximately 1 billion [24]. Instantaneous luminosity is defined as:

$$\mathcal{L} = \frac{1}{\sigma} \frac{dN}{dt} \tag{4.1}$$

where  $\sigma$  is the cross-section of the physics process under consideration, N is the number of events of that type, and t is time. Luminosity is a property of the accelerator itself, independent of the type of physics process being considered. It is therefore a useful metric for characterizing the performance of the LHC.

In order to maximize luminosity (and therefore maximize the number of potentially interesting collision events), the most important factor to increase is the number of particles per bunch. That's because luminosity is proportional to the *square* of the number of particles per bunch, according to 4.2.

$$\mathcal{L} = \frac{N_b^2 n_b f_{rev}}{2\pi \Sigma_x \Sigma_y} \tag{4.2}$$

where  $N_b$  is the number of particles per bunch,  $n_b$  is the number of bunches per beam,  $f_{rev}$  is the frequency of revolutions around the LHC, and  $\Sigma_x$ ,  $\Sigma_y$  are the transverse widths of the beam in the x and y direction. This equation tells us how to maximize instantaneous luminosity: put as many protons as possible into as small a space as possible, and accelerate them as fast as possible. The peak luminosity delivered to ATLAS by the LHC is  $2.06 \times 10^{34} cm^{-2} s^{-1}$  for proton-proton collisions, and  $6.2 \times 10^{27} cm^{-2} s^{-1}$  for Pb-Pb collisions.

# 4.2 The ATLAS detector

## 4.2.1 Coordinates

The coordinate system used in this document is the standard ATLAS coordinate system, detailed here. For both Cartesian and polar coordinates, the origin is defined as the nominal interaction point. The z-axis points down the beamline. The x-y plane is perpendicular to the beamline. The positive y-direction points up, and the positive x-direction points towards the center of the LHC ring. The positive z-direction therefore points counterclockwise along the LHC, when viewed from above, as required by a right-handed coordinate system.

For cylindrical coordinates, the z-axis is defined the same way as for Cartesian coordinates. The azimuthal angle  $\phi$  is the angle from the positive x-axis, while the polar angle  $\theta$  is the angle from the beamline. A more convenient measure of angle from the beamline is the *rapidity*, because rapidity differences are invariant under Lorentz boosts in the z-direction. Rapidity is defined as:

$$y = \frac{1}{2} \ln \frac{E + p_z}{E - p_z}$$
(4.3)

Another frequently used quantity is the *pseudorapidity*, which is defined as:

$$\eta = -\ln\frac{\theta}{2} \tag{4.4}$$

Pseudorapidity differences are also invariant with respect to longitudinal Lorentz boosts. In the limit where  $p_T \ll m$ , rapidity and pseudorapidity are equal. Pseudorapidity ranges from zero to plus or minus infinity. The x-y plane, which is perpendicular to the beamline, is described by a pseudorapidity  $\eta = 0$ . The z-axis, which is parallel to the beamline, is described by pseudorapidity  $\eta = \pm \infty$ .

A distance measure in  $\eta - \phi$  space is often used, especially when describing jets. This distance,  $\Delta R$  is defined as:

$$\Delta R = \sqrt{\Delta \eta^2 + \Delta \phi^2} \tag{4.5}$$

# 4.2.2 Magnet systems

The two ATLAS magnet systems are used to curve the tracks of charged particles passing through the detector, so that their momenta can be measured. The layout of the magnet systems can be seen in figure 4.6.

The first magnetic field is produced by the central solenoid, which surrounds the entire inner detector, and is surrounded by the barrel calorimeters. It generates a uniform 2 T axial magnetic field in the inner detector. The coils are made of Al-stabilized NbTi. The length is 5.8 m, and the outer diameter is 2.56 m [23]. The second magnet system, used to bend the trajectories of muons passing through the muon spectrometer, has a more complicated geometry. It consists of a barrel toroid section, and two symmetrical end-cap toroids. The



Figure 4.6: A diagram showing the layout of the ATLAS magnet system. The central solenoid is used to apply a force that curves the trajectories of charged particles in the inner detector. In red are the toroid coils, used to bend the tracks of charged particles passing through the muon spectrometer [23].

exact shape of the resulting magnetic field is quite complex, but roughly runs in a circular direction around the calorimeters.

The axial length of the barrel toroid is 25.3 m, and the outer diameter measures 20.1 m. The average field strength is 0.5 T and the superconducting material used is similar to that used in the central solenoid[23]. The magnetic field in the region  $|\eta| < 1.4$  is dominated by the barrel toroid, while the end-cap field dominates the region  $1.6 < |\eta| < 2.7$ . The region  $1.4 < |\eta| < 1.6$  is the transition region, where both sources contribute significantly to the magnetic field. Figure 4.7 shows an end-on view of the barrel toroid, after installation, and before the calorimeters are inserted. The end-cap toroids are used to generate the magnetic field for muons passing through the end-cap region of the muon spectrometer. The properties and geometry of the end-cap toroids are similar to the barrel toroid, with peak magnetic



field reaching 4.1 T[23].

Figure 4.7: A picture of the ATLAS barrel toroid after installation. In the center of the image is the calorimeter and central solenoid, before being moved into the final position [23].

# 4.2.3 Inner detector

The inner detector consists of silicon pixel detectors, silicon strip detectors, and transition radiation trackers. It covers a region from  $R = 33 \ mm$  to  $R = 1082 \ mm$  and  $|\eta| = 0$  to  $|\eta| = 2.5$ . The layout of inner detector subsystems is shown in figure 4.8.

## **Pixel Detector**

The innermost layer of ATLAS detectors is the pixel system. Since it lies closest to the interaction point, the pixel system experiences the highest flux of any ATLAS subdetector. This means that the pixel detector must have the greatest radiation hardness, greatest resolution, and greatest occupancy of any subdetector [23]. The pixel system consists of 1744 solid-state pixel sensors, arranged into a barrel region and two endcap regions. There are a total of  $8 \times 10^7$  silicon pixels, with a total silicon surface area of 1.7  $m^2$ , and 80.4 million pixel readout channels [23].



Figure 4.8: A quarter-section plan showing the layout of inner detector subsystems and their dimensions. Not shown is the innermost layer of the pixel detector, the IBL, which was installed in May 2014 [27].

The barrel region consists of four concentric cylindrical layers, coaxial with the beamline. The two endcap regions are each made up of three disks, arranged perpendicular to the beamline. The pixel barrel envelope covers a region from z = 0 to  $|z| = 400.5 \ mm$ . The four layers are located at increasing distances from the beamline, at  $R = 33 \ mm$ , 50.5 mm, 88.5 mm, and 122.5 mm. The six endcap disks are located at  $|z| = 495 \ mm$ , 580 mm, 650 mm and cover the region 88.8  $mm < R < 149.6 \ mm$  [23]. Figure 4.8 shows a quarter-section of the entire inner detector, as well as a detailed view of the pixel subsystem, not including the Insertable B Layer (IBL), which was installed in 2014.

The ATLAS pixel sensors are solid-state silicon detectors. The basic operating principle is that charged particles passing through a material generate electron-hole pairs, which are accelerated towards opposite ends of the material via an electric field. This generates a current, which can be measured by charge-sensitive sensors at the edge of the material [29]. The active material is a *pn*-junction, operated with a reverse bias voltage, until fully depleted. This reduces the thermal noise from free charge carriers to a low enough level that electronhole pairs from signal particles can be detected [29]. The operating principle is illustrated in figure 4.9, which shows the electron-hole pairs generated in the depletion region by an incoming ionizing particle, and the directions in which the electrons and holes are accelerated under the electric field.



Figure 4.9: The basic operating principle of a solid-state particle detector. A reverse bias is applied across a *pn*-junction, creating a depletion region. When a charged particle passes through this depletion region, electron-hole pairs are created, and the electrons and holes are accelerated in opposite directions under the electric field. The flow of electrons generates a current in the readout electronics.

In ATLAS, the pixel sensors consist of an n-type bulk, with  $p^+$  implants on the back side and  $n^+$  implants on the front side. Before irradiation, the active pn junction region exists between the n-type bulk and  $p^+$ - implanted side. Irradiation leads to the reduction in the effective doping concentration, until the bulk material undergoes type inversion. After type inversion, the active pn junction region switches to the  $n^+$ -doped side [30]. This process is illustrated in figure 4.10. This  $n^+$ -in-n design allows the sensors to continue to operate both before and after large doses of radiation. Each pixel tile has 47232 pixels, laid out in a grid of 144 columns by 328 rows. Some of the pixels are grouped to common read-out channels, resulting in 46080 read-out channels. This grouping is done so that an equal number of read-out channels can be connected to each of the 16 front-end read-out chips [30]. In 128 of the columns, each pixel implant is  $382.5 \times 30 \ \mu m^2$ , with pitch (center-to-center distance) of  $400 \times 50 \ \mu m^2$ . In the remaining 16 columns, the pixel sizes are  $582.5 \times 30 \ \mu m^2$ , with a pitch of  $600 \times 50 \ \mu m^2$  [30].

A major upgrade that occurred during the long shutdown in 2014 was to install the Insertable B-Layer (IBL) to the pixel detector. The IBL became the fourth and innermost layer of the pixel detector. It provides several key improvements to the tracking system, which will allow the pixel detector to maintain good performance even in the higher-luminosity environment that will be present in the High Luminosity LHC (HL-LHC) [31]. The IBL does this by improving tracking robustness against module failures, adding measurement



Figure 4.10: Graphic illustrating how  $n^+$ -in-n pixel sensors continue to operate after the type inversion that results from irradiation. In (a), the unirradiated state, the bulk is n-type, and the depletion zone occurs between the  $p^+$ -doped back side. After type inversion, in (b), the depletion zone occurs between the now p-type bulk and the  $n^+$ -doped front side [30].

redundancy to mitigate the effects of pileup, and adding an additional measurement closer to the interaction point [31]. As part of the IBL insertion project, the original beam pipe was removed, and replaced with a smaller-radius beampipe. Precision tooling and methods for insertion were developed and practiced for two years before the procedure was finally carried out. The tolerances were extremely tight, with only a 0.2 mm gap between the IBL and inner supporting tube [32]. An image of the IBL being inserted can be seen in figure 4.11

#### Silicon Strip Tracker

After the pixel detector system, the next innermost subdetector is the SemiConductor Tracker, or SCT . The SCT consists of 4088 modules, with a total silicon surface area of 63  $m^2$  [23]. Like the pixel system, the SCT barrel region consists of four concentric cylindrical layers, coaxial with the beamline. The two endcap regions are each made up of nine disks, arranged perpendicular to the beamline. The SCT barrel envelope covers a region from z = 0 to  $|z| = 746 \ mm$ , and the four layers are located at increasing distances from the beamline, at  $R = 299 \ mm$ , 371 mm, 443 mm, and 514 mm [33]. The nine endcap disks on each side range from  $|z| = 853.8 \ mm$  to  $|z| = 2720.2 \ mm$ , and cover the region 275  $mm < R < 560 \ mm$  [23].

Like in the pixel system, the SCT sensors are solid-state silicon detectors. Instead of pixels, the base unit is a strip of silicon, ranging in length from 6 to 13 cm. The modules are arranged in back-to-back pairs at a stereo angle of 40 mrad, in order to provide twodimensional resolution. These strips are made of p-type silicon, and are embedded in n-type silicon. The average pitch is 80  $\mu m$  [34]. The resolution resulting from this design is 17  $\mu m$  in the  $r - \phi$  direction, and 580  $\mu m$  in the z-direction [34].



Figure 4.11: The IBL as it was inserted into the pixel detector [32].

#### **Transition Radiation Tracker**

Moving outwards from the pixel system and SCT, the final inner detector subsystem is the transition radiation tracker, or TRT . Unlike the pixel or SCT systems, the TRT uses proportional drift tubes, referred to as straws, as sensors. To make accurate track measurements, the TRT relies on a larger number of hits over a longer distance than the pixel system or SCT . Since the TRT measures approximately 36 hits per track, and makes measurements over a longer distance, less spatial precision is required per hit [23]. The TRT only provides tracking information in the  $R - \phi$  direction, unlike the pixel and SCT systems, which each provide three-dimensional tracking information. But transition radiation photons generated in the TRT gas-filled tubes can also aid in electron identification [23].

The TRT uses proportional drift tubes to detect charged particles. A tube is filled with a mixture of two or more gases, including an inert gas such as Xenon, and an electric field is applied across the tube. When a charged particle passes through the tube, ion pairs are generated from the inert gas. Positive and negative ions drift in opposite directions, towards the cathode and anode, respectively. If a particle's energy is completely absorbed in the tube, the number of ions produced in stopping the particle is proportional to the original energy of the particle [35]. In addition to the inert gas, it is common to add another gas, such as Carbon Dioxide, to stabilize the ionization process. This additional gas is referred to as a quencher gas.

Like the pixel system and SCT systems, the TRT consists of a barrel region and two end-cap regions. In the barrel region, straw tubes are arranged coaxial to the beamline and measure 144 cm in length. In the end-cap region, straw tubes are arranged radially, and measure 37 cm in length [23]. The TRT barrel envelope covers a region from z = 0to |z| = 780 mm, and 554 < R1082 mm, and the end-cap envelop covers a region from z = 848 to |z| = 2710 mm, and 644 < R1004 mm. This provides tracking coverage out to  $|\eta| = 2.0$  [23]. A charged track passing trough the TRT will typically produce 36 hits. Each straw provides a hit resolution of 130  $\mu$ m. The TRT has a total 351,000 total readout channels.





The TRT straw tube walls are constructed of two layers of 25  $\mu$ m-thick Kapton film. The Kapton is coated with a 0.2  $\mu$ m-thich layer of aluminum, followed by a 6  $\mu$ m-thich layer of carbon-polyimide. The Aluminum layer serves to provide electrical conductivity, and the carbon-polyimide layer is to protect the aluminum. The two Kapton layers are sealed together with a 5  $\mu$ m-thick polyurethane layer [36]. The TRT straw wall design is shown in figure 4.13

The straws are filled with a mixture of 70% Xe, 27% CO<sub>2</sub>, and 3% O<sub>2</sub>. Carbon dioxide is used as a quencher gas, which is needed to guarantee that the ionization procedure is stable. The addition of a small amount of oxygen increases the voltage difference between the working point and the breakdown voltage, further stabilizing the process [36].



Figure 4.13: Schematic of the TRT straw wall design. Two coated Kapton layers are sealed together with polyurethane [36].

In order to maximize hit efficiency, the straw diameter should be as large as possible. But there is a tradeoff: as straw diameter increases, so does the drift time. The optimal diameter to ensure acceptable time resolution for 25 ns bunch-spacing was determined to be  $4 \ cm$  [36].

#### **Tracking Resolution**

The number of layers crossed per track and the intrinsic accuracy for each inner detector subsystem is summarized in table 4.1. The TRT provides the largest number of hits per track, but has the poorest resolution, and only provides resolution in  $R - \phi$ . The Pixel and SCT systems have much better resolution in  $R - \phi$  and also provide z-resolution in the barrel region, and *R*-resolution in the disks.

	number of layers	$R - \phi$ resolution	z resolution (barrel) / $R$ resolution (disks)
Pixel	3	$10 \ \mu m$	$115 \ \mu m$
SCT	8	$17 \ \mu m$	$580~\mu m$
TRT	36	130 $\mu m$	N/A

Table 4.1: Summary of number of layers crossed per track as well as intrinsic accuracy in  $R - \phi$  and z for each of the inner detector subsystems.

## 4.2.4 Calorimeters

The ATLAS calorimeters are designed to absorb and measure the energy of both electromagnetic and hadronic showers. Different materials and geometries are used for different calorimeter subsystems, but the operating principle is the same. In each calorimeter, there is an absorber medium, and a sampling medium. When a particle strikes the absorber medium, it triggers a shower of particles which then pass into the sampling medium. The sampling medium is ionized by these showering particles, and instrumentation is used to measure the amount of ionization. In order to accurately measure shower energy, the ATLAS calorimeters are designed to fully absorb both electromagnetic and hadronic showers.

The innermost calorimeter is a high-granularity detector optimized for measuring electromagnetic showers, which covers the range  $|\eta| < 3.2$ . In the electromagnetic calorimeter, lead is used as an absorber and LAr as the sampling medium. The outer calorimeter has a coarser granularity and is optimized for measuring jets and  $E_{Tmiss}$ . It consists of scintillator tiles and uses steel as the absorber. In addition to the EM and hadronic calorimeters, there is copper-tungsten/LAr calorimeter providing coverage out to  $|\eta| < 4.9$ , known as the forward calorimeter, or FCal. The layout of the full ATLAS calorimeter system can be seen in figure 4.14

#### **Electromagnetic Calorimeters**

The electromagnetic calorimeter is located just outside the solenoid that surrounds the inner detector. It is optimized for measuring the energy of electromagnetic showers. The EM calorimeter is also designed to measure the direction of neutral particles, which cannot be tracked by the inner detector. In the barrel region, the EM calorimeter covers the range  $|\eta| < 1.475$ . In the end-cap region, it covers a range of 1.375 < |eta| < 3.2.

The calorimeters have to be thick enough to fully absorb shower energy and to minimize punch-through into the muon system. The EM calorimeter depth was designed to be 22 interaction lengths  $(X_0)$  in the barrel, and 24  $X_0$  in the end-caps. An interaction length is defined as the average distance traveled by a particle through a material when it has lost 1 - 1/e of its original energy [37].

The calorimeters consist of accordion-shaped lead absorber plates with copper-kapton electrodes in between. The accordion shape provides full symmetry in  $|\phi|$  with no gaps. The



Figure 4.14: Layout of the ATLAS calorimeter system [23].

absorber plates are grounded and the electrodes are kept at 2000 V in the barrel region and between 1000 and 2500 V in the end-cap region. The whole system is immersed in liquid argon [23]. A particle passing through the lead absorbers generates a shower of particles, which ionize the LAr. Due to the electric field between the absorber plates and electrodes, ions drift towards the electrodes. This generates a pulse in the electrodes which can then be recorded.

The calorimeter cells are arranged into three layers, with highest granularity closest to the interaction point. Cells in layer 1 have a granularity of  $\delta\phi \times \delta\eta = 0.0245 \times 0.0031$ . This very fine segmentation is useful in accurately determining the direction of incoming particles. It is also useful in discriminating between an individual photons and a neutral  $\pi$  meson decaying to two photons [37]. In the second layer, the granularity is reduced  $\delta\phi \times \delta\eta = 0.0245 \times 0.0245 \times 0.025$ . In the third and final layer, the granularity is further reduced to  $\delta\phi \times \delta\eta = 0.0245 \times 0.025$ . The geometry of the EM calorimeter cells can be seen in figure 4.15

Inside the innermost layer of the barrel EM calorimeter is an 11 mm-thin presampler, covering a range of  $|\eta| < 1.8$ . The purpose of this presampler is to correct for energy lost to material upstream of the EM calorimeter.



Figure 4.15: Schematic of EM calorimeter, showing the granularity of each layer [37].

#### Hadronic Calorimeters

The most important ATLAS subdetector for measuring jets is the hadronic calorimeter system. The hadronic calorimeter system consists of central and extended tile calorimeter barrels, two end cap calorimeters, and a forward calorimeter.

The central tile barrel covers the range  $|\eta| < 1.0$ , the extended tile barrels cover the range  $0.8 < |\eta| < 1.7$ , and the end-cap calorimeters cover the range  $1.5 < |\eta| < 3.2$ . The forward calorimeters (FCal) cover the extreme forward region,  $3.1 < |\eta| < 4.9$ . The central barrel is 5.8 m long, and the extended barrels are each 2.6 m long. Both the central and extended barrels cover the range 2.28 m < R < 4.25 m. Each of the three tile calorimeters is composed of 64 wedge-shaped modules. The geometry of the hadronic calorimeter system, not including the end-caps, can be seen in figure 4.14.

The hadronic calorimeters in the central and extended barrel regions are tile calorimeters, which consist of alternating steel absorber plates and scintillating tiles. Particles that pass through the steel plates generate hadronic showers. The hadrons from these showers then stimulate the production of photons in the scintillating tiles, which are then collected by photomultiplier tubes, producing a current that can be measured. A wavelength-shifting fiber is used to convert the ultraviolet photons produced in the scintillator into optical photons before entering the photomultipliers. A schematic of a tile calorimeter module can be seen in figure 4.16 The  $\eta$ - and depth-dependent segmentation of the tile calorimeter modules can be seen in figure 4.17

The hadronic end-cap calorimeter (HEC) consists of two cylindrical wheels on each side



Figure 4.16: Schematic of a tile calorimeter module [23].

of the barrel, each of which has a radius of  $2.03 \ m$ . Each of these wheels is made up of 32 modules. The HEC wheels closest and farthest from the interaction point are called HEC1 and HEC2. HEC1 modules have higher granularity and sample fraction than HEC2 modules. A schematic of the HEC modules can be seen in figure 4.18.

The HECs use LAr as the sampling material and copper as the absorber material. The copper plates are separated by gaps of 8.5 mm, with three electrodes in between. This creates four drift zones between each copper plate. Typical drift time across the drift zones is 430ns, at a potential difference of 1800V [23]. There are a total of 5632 readout cells. In the region  $|\eta| < 2.5$ , the readout cell size is  $\delta\phi \times \delta\eta = 0.1 \times 0.1$ , and  $\delta\phi \times \delta\eta = 0.2 \times 0.2$  elsewhere.

#### **Forward Calorimeters**

Special forward calorimeters (FCal) were designed to cover the forward region,  $3.1 < |\eta| < 4.9$ . This region experiences very high particle flux compared to the barrel or end-cap regions, so a different design was needed. Each FCal consists of three modules. The module



Figure 4.17: Schematic of the central (left) and extended (right) tile calorimeter, showing the  $\eta$ - and depth-dependent segmentation [23].



Figure 4.18: Schematic of hadronic end-cap calorimeter module [23].

closest to the interaction point (FCal1) is optimized for EM showers, while the middle and outer modules (FCal2 and FCal3) are optimized for hadronic showers. All three modules are  $45 \ cm$  deep, and use LAr as the sampling material.

FCal1 uses copper as the absorber material, while FCal2 and FCal3 use tungsten as the primary absorber. To further reduce punch-through into the muon system, a copper shield plug is placed behind FCal3. The location of the FCal with respect to the end-cap calorimeters can be seen in figure 4.19 The FCal, hadronic end-cap, and electromagnetic end-cap calorimeters are all contained in the same end-cap cryostat. The overlapping design of the end-cap and forward calorimeters allows for hermetic coverage with minimal energy loss out to  $|\eta| < 4.9$ .



Figure 4.19: Schematic showing the layout of the forward calorimeters and end-cap calorimeters [23].

In order to provide accurate energy measurements of photons, jets, and  $E_{Tmiss}$ , it is necessary for the calorimeters to absorb as much shower energy as possible. Any particles that don't get absorbed by the calorimeter are considered punch-through particles. These punch-through particles reduce the accuracy of calorimeter measurements and add noise to the muon system, so they should be minimized. For hadronic showers, the analog to a radiation length is called an interaction length,  $\lambda$ . The amount of material for each part of the calorimeter, as measured in interaction lengths, can be seen in figure 4.20.

#### **Energy Resolution**

The energy resolution of the LAr calorimeters is parameterised as [38]:

$$\frac{\sigma(E)}{E} = \frac{a}{\sqrt{E(GeV)}} \oplus b \tag{4.6}$$

Electron test beams ranging from 10 GeV to 240 GeV are used to measure the electron energy resolution of the LAr calorimeters. The resulting parameters were found not to exceed  $a = 10\%\sqrt{GeV}$  and b = 0.7% over both the EM barrel and EM end-cap calorimeters. The hadronic end-cap calorimeter has design resolution  $\frac{50\%}{\sqrt{E}} \oplus 3\%$ , while the forward calorimeter



Figure 4.20: Amount of material, in units of interaction length, versus pseudorapidity, including all material before the EM calorimeter, the EM calorimeter, and the hadronic calorimeters. The light blue indicates the amount of material in front of the muon spectrometer for  $|\eta| < 3.0$  [23].

has design resolution  $\frac{100\%}{\sqrt{E}} \oplus 10\%$ . The hadronic tile calorimeter provides a jet resolution of  $\sigma(E)/E = 50\%/\sqrt{E}$  [23].

# 4.2.5 Muon spectrometer

The outermost ATLAS system is the muon spectrometer, used to measure the momenta of charged particles that exit the calorimeter systems. Muons are not absorbed by the EM calorimeter because they undergo less bremsstrahlung than electrons due to their high mass. They are also no absorbed by the hadronic calorimeter because they do not interact strongly. As a result, muons are the most common particle to escape the calorimeter system.

The overall dimensions of the muon spectrometer, which define the demensions of ATLAS, are 44 m long by 22 m high. The muon spectrometer consists of four subsystems. The monitored drift tubes (MDTs) and cathode strip chambers (CSCs) are used for high-precision tracking, while the resistive plate chambers (RDCs) and thin-gap chambers (TGCs) are used for triggering and bunch-crossing identification. The location of each of the four muon

subsystems can be seen in figure 4.21



Figure 4.21: Layout of the muon spectrometer, indicating the location of barrel and end-cap toroids, along with the four subsystems [23].

As described in 4.2.2, the magnetic field in the muon system runs in a roughly circular direction around the beamline. As a result, charged particles are bent in the  $\eta$  direction in the muon spectrometer. In the barrel region, covering  $|\eta| < 1.0$  the muon system is laid out in three concentric cells, located at 5 m, 7.5 m, and 10 m from the interaction point. Full coverage is provided in this region, except for a small gap at  $|\eta| = 0$ , to allow for services to the solenoid, calorimeters, and inner detector systems. The size of the acceptance gap is  $|\eta| < 0.04$  in the innermost layer, and  $|\eta| < 0.08$  in the outer layers. The end-cap consists of large wheels, located at  $|z| \approx 7.4 m$ , 10.8 m, 14 m, and 21.5 m. Figure 4.22 shows a cross-section of the barrel region of the muon spectrometer in the x - y plane, and a cross-section of the barrel regions in the y - z plane. The small acceptance gap at  $\eta = 0$  can be seen.

The muon spectrometer provides high-precision momentum measurements out to  $|\eta| < 2.7$ , and triggering on high-momentum muons out to  $|\eta| < 2.4$ . The ATLAS design specification requires momentum resolution of 10% for 1 *TeV* muons. This resolution is achieved or exceeded for the range 10 *GeV* <  $p_T < 1$  *TeV*. The contribution to  $p_T$  resolution from various sources at different  $p_T$  is shown in figure 4.23. The dominant contribution at low- $p_T$  is from energy loss fluctuations, while at high- $p_T$  it's from wire resolution and autocalibration [39].

Most of the precision momentum measurements in the muon spectrometer are performed by monitored drift tubes (MDTs). The MDT subsystem covers the region  $|\eta| < 2.7$ , except for the innermost end-cap layer, where they cover out to  $|\eta| = 2.0$ . There are a total of 1174 MDT chambers, each made up of between 3 and 8 layers of drift tubes. The aluminum tubes



Figure 4.22: Cross-sections in x - y (left) and y - z (right) of the muon spectrometer [39].



Figure 4.23: The contribution to muon spectrometer  $p_T$  resolution from various sources over a range of  $p_T$  [39].

are 3 cm in diameter and range between 0.9 m and 6.2 m in length. They are filled with a mixture of 80 %Ar and 20 %CO<sub>2</sub>. The MDTs provide a resolution of 80  $\mu m$  per tube, and 35  $\mu m$  per chamber. In order to provide high-precision measurements, the locations of all

chambers and their support structures are constantly monitored. Temperature and magnetic field conditions are also monitored, in order to account for thermal expansion and changes to drift time [39]. The MDTs can be safely operated at a rate of up to 150  $Hz/cm^2$ .

In the innermost end-cap layer, covering the region  $2.0 > |\eta| > 2.7$ , the background rate is too high for safe operation of MDTs. Instead, precision momentum measurements are performed using cathode strip chambers (CSCs), which can be safely operated at rates up to 1000  $Hz/cm^2$  [23]. The CSC chambers use the same gas mixture as MDT chambers, but instead of drift tubes there are grids of anodes and cathodes, allowing for measurements of  $\eta$  and  $\phi$  based on the charge induced in the wires. There are a total of 32 CSC chambers, arranged into 4 disks of 8 chambers each. Each chamber contains 4 CSC layers, resulting in 4 independent measurements of  $\eta$  and  $\phi$  for each track. Cathodes are made of 17 mm copper strips, and anodes are gold-plated tungsten measuring 30 mm in diameter. Sense-wire pitch is 2.54 mm, but resolution is limited by readout pitch, which is 5.08 mm. This results in an  $\eta$  resolution of 60  $\mu$ m, and a  $\phi$  resolution of 5 mm [39]. Since the magnetic fields bend muons in the  $\eta$  plane, precise momentum measurements require far greater resolution in  $\eta$ .

Resistive plate chambers (RPCs) are used for the muon trigger in the barrel region  $|\eta| < 1.05$ . In the middle station, there are two layers of RPCs, used for the low- $p_T$  threshold trigger, which has a threshold of  $6 - 9 \ GeV$ . In the outer station, there is a single layer of RPCs, used for the high- $p_T$  threshold trigger, with a threshold of  $9 - 35 \ GeV$ .

Each RPC consists of two rectangular detectors, called units. Standard RPCs are paired with MDTs, and have the same dimensions. A few additional smaller RPCs are included in areas where MDTs cannot be installed due to lack of space. Each RPC provides two measurements of  $\eta$  and  $\phi$  for a charged particle passing through it. A charged particle passing through all three layers therefore generates 6  $\eta - \phi$  measurements. To reduce fake tracks, 3 out of 4 of the middle-station detectors and at least one outer layer station must register hits.

An RPC detector consists of two plastic insulating plates, separated by 2 mm, with a gas occupying the internal space, and a voltage difference of 9.8 kV across the plates. Charged particles passing through the plates generate an avalanche of charge which can then be measured by metallic strips connected to the plates. The insulator plates are made of phenolic-melaminic plastic laminate, and the gas is a mixture of C<sub>2</sub>H<sub>2</sub>F4, Iso – C<sub>4</sub>H<sub>10</sub>, and SF<sub>6</sub> in the proportions 94.7%, 5%, 0.3% [23]. The RPCs provide a time resolution of 2 ns.

In the forward region,  $1.05 < |\eta| < 2.4$ , muon triggering is provided by thin-gap chambers (TGCs). TGCs also provide a  $\phi$  measurement to complement the  $\eta$  measurement provided by the MDTs. In the MDT end-cap middle layer, there are seven layers of TGCs, and in the inner layer, there are 2 layers of TGCs. Each TGC layer consists of two concentric rings.

Like the CSCs, the TGCs are multi-wire proportional chambers (MWPCs). To reduce drift time so that an acceptable time resolution can be achieved, the drift distance must be kept small, and the electric field kept high. The wire-to-cathode distance is 1.4 mm in the TGCs, even closer than the wire-to-wire distance of 1.8 mm, and the voltage is kept at 2.9 kV. A highly quenching gas mixture of 55%CO<sub>2</sub> and 45%n – C<sub>5</sub>H<sub>12</sub> is used in the TGCs [23]. The resulting time resolution provided by the TGCs is 4 ns [39].

## 4.2.6 Trigger

Because of the very high event rate at the LHC, and the large amount of data collected per event, it is impossible to record every single event. There are 40 million bunch crossings per second, and each collision can generate 1.6 MB of raw data. The ATLAS trigger system is designed to make very fast decisions about which events to record and which to ignore, in order to reduce the event rate to approximately 1 kHz. The challenge of trigger design is to balance processing time against decision accuracy, to make sure that events that could be useful for physics analysis are not lost. The selection criteria used by each level of the trigger are known as the trigger menu. A large number of possible trigger selection criteria are designed, in order to serve the broad range of physics analyses performed at ATLAS.

The ATLAS trigger system consists of a hardware based level 1 (L1) trigger and a software-based high-level trigger (HLT). The L1 trigger reduces the event rate from the original 40 MHz to approximately 100 kHz. The HLT further reduces the event rate to approximately 1 kHz, which is the maximum rate of data transfer from the detector. The flowchart of data through the trigger system can be seen in figure 4.24

For every event, the L1 trigger system reads reduced-granularity measurements from both the hadronic and electromagnetic calorimeters. Special calorimeter cells, called trigger towers, in each calorimeter were designed for this purpose. The trigger towers have a granularity of  $\delta\eta \times \delta\phi = 0.1 \times 0.1$ . The location of the trigger towers at the back of the electromagnetic calorimeter can be seen in figure 4.15. The L1 trigger also reads data from the RPCs and TGCs of the muon spectrometer system, as discussed in 4.2.5. The calorimeter data will be used for photon, electron, tau, jet, and  $E_{Tmiss}$  triggers, and the muon data will only be used for muon triggers. Data from these two sources are fed into the central trigger processor (CTP), which makes the decision about whether or not accept each event at level 1.

If an event is accepted at L1, full detector raw data for that event is read out and temporarily stored until the HLT decision is made, and some amount of reconstruction is done at this time. Later, if an event is accepted by the HLT, the full tracking reconstruction algorithm will be applied. The L1 system then sends region-of-interest information to the HLT. Regions of interest are the subsections of the detector that the L1 trigger has identified as possibly containing interesting physics objects. The full detector information and L1 region-of-interest data are combined by the HLT, which then makes the final decision on whether or not to keep an event. Only when an event passes both the L1 and HLT decision criteria will its be permanently stored. The L1 trigger system decision time is just 2.5  $\mu s$ , while the HLT makes decisions on the order of seconds.



Figure 4.24: Flowchart outlining the level 1 and high-level trigger design and data acquisition system [40].

# Chapter 5

# Jets

A jet is a collimated spray of stable particles measurable in the detector, which arises from the production of a quark or gluon in the scattering process. Due to QCD confinement, the quarks and gluons themselves can never be observed, so jets serve as an observable proxy for these more fundamental particles. However, there is not a simple one-to-one correspondence between each jet measured in an event and a quark or gluon in the hard scattering process. Due to the complex nature of proton-proton collisions, as well as the fundamental probabilistic nature of quantum mechanics, the partonic source for a given jet can never be specified exactly. Likewise, when performing theoretical calculations, the number and property of jets that arise from a given parton produced in the hard scattering can only be specified probabilistically. Section 5.1 will explain the many different parts of a proton-proton collision that contribute to jet production.

Furthermore, the number and properties of jets measured in a single event will depend on the choice of jet reconstruction algorithm. In order for experimentalists to test the predictions of theory, it's important for standard jet definitions to be decided upon. While there is no one theoretically correct choice of jet definition, there are certain properties of jet algorithms that make them more or less desirable. The different kinds of jet algorithms and their properties will be discussed in section 5.2. Section 5.3 will explain the origin of jet mass and how it can be accurately reconstructed. And section 5.4 will explain the process of jet reconstruction and calibration in ATLAS.

# 5.1 The proton-proton collision environment

As discussed in 2.2.4, the goal of proton-proton collision experiments is to understand the interactions between fundamental particles, but due to QCD confinement it is impossible to simply collide the individual partons of interest. Instead, complicated bound states of quarks and gluons are collided, with the goal of measuring the processes that occur when constituent partons collide with each other. As a result, proton-proton collision events create a very messy environment from which the hard-scatter process has to be deduced.

#### CHAPTER 5. JETS

QCD confinement also has the consequence that the final state quarks and gluons created in LHC collisions can never be directly detected. Instead, collimated sprays of hadrons, called jets, must be measured in order to attempt to reconstruct the hard scattering process that gave rise to them.

Figure 5.1 illustrates a representative proton-proton collision event. Even though this is not a dijet or multijet production event, it is useful for understanding all the parts of a proton-proton collision that must be considered when calculating the predicted rates of multijet production.



Figure 5.1: Diagram of a proton-proton collision event. Illustrated are the initial state radiation, underlying event, hard-scatter process, final state parton shower, fragmentation, hadron decays, and final state QED radiation [41].

The incoming protons are illustrated by green blob with three incoming arrows to represent the constituent valence quarks. The initial state parton shower, governed by QCD, is shown in blue. The hard scatter process is represented by the large red circle. Quarks and gluons from the incoming proton that do not participate in the hard scatter process can nonetheless interact with each other, creating a so-called underlying event, shown in purple. All of the strongly-interacting final state quarks and gluons undergo final state parton showering, also in red. Once the final state parton shower particles reach a low enough energy, they hadronize, in a non-perturbative process represented by the light green blobs. The resulting hadrons then decay through various decay chains, shown in dark green. Photon radiation, governed by QED, can occur at any of these stages, and is shown in yellow.

Accurate modelling of multijet events is challenging for several reasons. The first challenge comes from merging the matrix-element differential cross section with the parton shower simulation An event with n jets in the final state can arise from a hard scattering event with n partons in the final state, or from an event with n-1 partons in the final state, where the additional jet arises during the parton shower. The contributions from these two paths cannot be exactly factorized, but different merging schemes have been developed to combine them in a way that avoids double counting and yields accurate estimates of jet kinematic observables [42]. Uncertainties necessarily arise from the choice of merging scheme and scale parameters. There are additional uncertainties from higher-order diagrams, which can be evaluated by observing the change in cross section as the QCD scale is varied. High jet multiplicity events require diagrams with more factors of  $\alpha_s$ , so the uncertainty from higher-order diagrams grows with jet multiplicity. Additional contributions to uncertainty come from the choice of PDF and from the underlying event. Figure 5.2 shows the sizable discrepancy in third-jet  $p_T$  and jet multiplicity distributions across several event generators. Because of the challenges in using Monte Carlo to estimate QCD multijet cross sections, the analysis presented in this thesis will use a fully data-driven background estimation method.

# 5.2 Clustering algorithms

In order to measure jets in an event, a decision has to be made on the type of jet reconstruction algorithm to use. The observable result of a collision event can consist of many clusters of stable particles or calorimeter hits. Exactly which particles to cluster together is highly non-trivial, and developing the algorithms to make these decisions is an active area of research in particle physics. The 1990 Snowmass accord defined a set of criteria which should be met by any jet algorithm. The criteria for a developing a jet reconstruction algorithm are [44, 45]:

- 1. Simple to implement in an experimental analysis
- 2. Simple to implement in the theoretical calculation
- 3. Defined at any order in perturbation theory
- 4. Yields finite cross sections at any order of perturbation theory
- 5. Yields a cross section that is relatively insensitive to hadronization



Figure 5.2: Distribution of third-leading jet  $p_T$  vs. leading jet  $p_T$  (left) and jet multiplicity (right) for events simulated with several different event generators and parton shower methods. The bottom pane of each plot shows the ratio with respect to the Pythia 8 result. [43].

Jet reconstruction algorithms that meet all of these criteria allow experimentalists to test theoretical predictions, because both the theoretical calculations and experimental analysis can use the same jet definition. Jet algorithms are defined as acting on abstract "objects". The fifth Snowmass criterion ensures that in theoretical calculations, jet algorithms can be applied at either the parton-level or hadron level. In either case, the object which the algorithm acts on will be a particle. Jet algorithms can also be applied by experimentalists to reconstruct jets from either simulated or real calorimeter hits. In this case, the object which the algorithm acts on is a cluster of calorimeter hits.

Historically, there have been two major types of jet algorithms used in particle physics experiments: cone-based algorithms and sequential recombination algorithms. Cone-based algorithms aim to cluster jets in  $\eta - \phi$  space. They are fast and easy to implement, and produce jets with regular boundaries in  $\eta - \phi$  space, but have many theoretical and practical downsides compared to sequential recombination algorithms. They typically use the highesttransverse-momentum (hardest) object in the event as a seed for a jet cone, then define a cone around that seed as the leading jet, before moving on to the next-hardest object. Cone-based algorithms have the advantage of producing regularly-shaped jet boundaries, which can simplify theoretical calculations and make experimental calibration easier [46].

#### CHAPTER 5. JETS

However, cone-based algorithms typically lack collinear safety<sup>1</sup>, which is a property that must be satisfied to meet the Snowmass criteria.

# 5.2.1 IRC safety

Jet algorithms should ideally possess infrared and collinear (IRC) safety. This means that their results should not change due to soft gluon emissions or near-collinear emissions during either the parton shower or hadronization process. IRC safety is important because soft and collinear emissions are extremely common in QCD, and create divergences in perturbative calculations if the algorithm is IRC-unsafe [44].

#### Collinear safety

A collinear-safe algorithm is insensitive to near-collinear emissions that occur during parton showering or hadron decays. Most cone-based algorithms are not collinear-safe because they seed jets with the hardest object in the event, which is very likely to change if there is a near-collinear emission.

In figure 5.3, for the collinear-unsafe cone-based algorithm, the hardest object in the event changes depending on whether the radiated gluon is reabsorbed or emitted nearly collinear to the original quark. Thus, the resulting seed will change, yielding different jet results. The loop diagram and collinear emission diagram both lead to divergences in the theoretical cross-section. In a collinear-safe algorithm, the two divergences would cancel out because both amplitudes must be added to calculate the rate of the single jet being produced. But in a collinear-unsafe algorithm, the two divergences do not cancel out, so perturbation cross-sections are not finite.

#### Infrared safety

A related concept is that of infrared safety. Jet algorithms should insensitive to soft gluon emissions. These emissions are extremely common during parton showering, and since they occur in the non-perturbative regime of QCD, very hard to accurately predict.

Figure 5.4 illustrates the result of an infrared-unsafe algorithm. A W boson decays to a quark-antiquark pair, which should result in two hard jets. Diagrams (b) and (c) each lead to IR divergences, which would cancel if the algorithm is infrared-safe. But for an infrared-unsafe algorithm, diagram (c) leads to a different number of jets from diagram (b), so the divergences do not cancel [44].

 $<sup>^{1}</sup>$ The exception to this is the SISCone algorithm [47], a cone-based algorithm which is both IR and collinear safe.


Figure 5.3: Illustration comparing collinear-safe (left) and collinear-unsafe (right) jet clustering algorithms. The emission of a near-collinear gluon changes the number of jets in the event, and results in diverges in theoretical calculations. The x-axis represents rapidity, and y-axis represents transverse momentum [44].



Figure 5.4: Illustration of infrared unsafety. In an event with a W boson decaying to two hard partons, the emission of a soft gluon can change the result of an infrared-unsafe algorithm [44].

### 5.2.2 Sequential recombination algorithms

Sequential recombination algorithms are the most commonly used jet algorithms in ATLAS today. Instead of clustering in  $\eta - \phi$  space, they cluster in transverse momentum space. The resulting jets have irregular boundaries, which adapt to soft radiation. Sequential recombination algorithms are generally slower to run, but have become more popular since the invention of the FastJet algorithm [48]. The sequential recombination algorithms discussed here are IRC safe.

All sequential recombination algorithms start by defining the distance measures:

$$d_{ij} = \min\left(k_{Ti}^{2p}, k_{Tj}^{2p}\right) \frac{\Delta_{ij}}{R^2}$$
(5.1)

$$d_{iB} = k_{Ti}^{2p} \tag{5.2}$$

where the indices i and j are used for two objects under consideration,  $d_{ij}$  is the distance between those two objects,  $k_T$  is the transverse momentum,  $\Delta_{ij}^2 \equiv (\eta_i - \eta_j)^2 + (\phi_i - \phi_j)^2$  is the distance-squared in  $\eta - \phi$  space, and p and R are two parameters that control the behavior of the algorithm. The object-beam distance,  $d_{iB}$  is also used by the algorithms, and is defined for each object individually rather than for pairs of objects. The choice of  $R^2$  roughly controls the area of the resulting jets. It is often referred to as a radius parameter, but the jets that result from these algorithms have irregular shapes, so this term is used loosely. The choice of p determines the class of sequential recombination algorithms, each of which has different properties. The steps are then all the same for all sequential recombination algorithms:

- 1. Calculate  $d_{ij}$  over all pairs of objects, and  $d_{iB}$  for each object individually
- 2. If the minimum of the set  $\{d_{ij}, d_{iB}\}$  is one of the  $d_{ij}$ 's:

Sum the four vectors of objects i and j

Add the resultant four-vector to the list of objects, and remove the four vectors for objects i and j

Go to Step 1

3. Else if the minimum of the set  $\{d_{ij}, d_{iB}\}$  is one of the  $d_{iB}$ 's:

```
Call the object i a jet
Remove the object i from the list of objects
Go to Step 1
```

An inclusive clustering algorithm stops when all objects have been clustered into a jet, while an exclusive algorithm stops when a pre-defined number of jets have been clustered [49].

The Cambridge-Aachen (C/A) algorithm has p = 0, and so does not directly use the transverse momentum of the jets. Instead the distance measures are purely in terms of pseudorapidity and azimuthal angle, so objects are clustered based on how close together they are in this radial distance measure. The boundaries of resulting jets are sensitive to random fluctuations of soft objects, such as those from pileup and the underlying event.

The  $k_T$  algorithm has p = 1, and so starts by clustering objects which are both spatially close together and have low momentum. Like with the C/A algorithm, the resulting jets are sensitive to random fluctuations in soft objects from pileup and the underlying event.

In the anti- $k_T$  algorithm, the objects which are closest together and hardest are clustered together first. The resulting jet shapes are insensitive to the details of the soft radiation.

As a result, the impact of pileup and the underlying event on jet momentum resolution is smallest for the anti- $k_T$  algorithm [50].

A comparison of the different jet clustering algorithms can be seen in figure 5.5. The same parton-level simulated event data was input into four different jet clustering algorithms. Random soft particles were also added to the parton-level event data. The resulting jets are plotted in  $y - \phi$  space, with colors labelling the different jets, and the height of each bar indicating the transverse momentum in that region. For the  $k_T$  and C/A algorithms, the jet boundaries are highly irregular, and depend strongly on the details of the random soft particles [50].

ATLAS uses the ant- $k_T$  algorithm for its jet reconstruction because of its lack of sensitivity to pileup and underlying event, as well as its regular jet boundaries. The standard choice of radius parameter used by ATLAS is R = 0.4, and a radius parameter of R = 1.0 is used for large-R jets, as discussed in the next section. This analysis uses both small-R and large-Rjets, which requires running the jet clustering algorithm twice for each event, starting from the topo-clusters each time.

### 5.3 Jet mass

As the center-of-mass energy of collisions increases, it becomes increasingly important to study the internal structure of jets, in addition to the kinematic properties of jets within events. The decay products of hadronically-decaying heavy objects, such as W or Higgs bosons, become harder to resolve into separate jets as the momentum of the original object increases. Recently, a large number of jet substructure observables have been developed, and an entire subfield has emerged dedicated to the study of the internal properties of jets.

For a two-pronged decay, for example  $W \to q\bar{q}$ , the angular separation between the decay products is well approximated by:

$$\Delta R \approx \frac{2m}{p_T} \tag{5.3}$$

where  $p_T$  is the transverse momentum of the W, and m is the W mass. For W bosons with  $p_T \gg m$ , jets with radius parameter R = 0.4 will no longer resolve the decay products into separate jets, but instead will contain the all decay products of the W into a single jet. Figure 5.6 illustrates the boosting effect for the simulated decay  $Z' \rightarrow t\bar{t}$ , where the Z'represents a new heavy resonance. Since  $m_{Z'} \gg m_t$ , the top quarks are highly boosted, so their decay products cannot be resolved by jets with R = 0.4. On the other hand, jets with R = 1.0 will capture all decay products with high efficiency, so the substructure of R = 1.0jets can potentially be used to reconstruct these boosted tops.

When dealing with boosted objects, larger radius parameters are often used, in order to increase the probability of capturing the full decay products of a boosted object. Jets with large radius parameters, typically R = 1.0 or R = 1.2 are referred to as fat jets.



Figure 5.5: Results of running four different jet clustering algorithm on the same set of parton-level simulated event data. SISCone (lower-left) is a cone-based algorithm, while the other three are sequential recombination algorithms defined by different values of p [50].

The earliest-developed, and most straightforward jet substructure observable is mass. The mass of a jet is calculated from the four-vectors of its constituent objects:

$$m^2 = \left(\sum E_i\right)^2 - \left(\sum \vec{p_i}\right)^2 \tag{5.4}$$

For a jet that contains the collimated decay products of an individual boosted resonance, the mass of the jet corresponds to the mass of the resonance,  $m_{jet} \approx m_{resonance}$ .

For a QCD jet, i.e. one that is a result of a high- $p_T$  gluon or light quark, one might expect the mass to also be very low or close to zero. But in fact perturbative QCD processes lead to a nonzero expected mass for light quark and gluon jets. Calculations of perturbative QCD jet mass are beyond the scope of this thesis, but the resulting mass is proportional to the jet radius parameter, R, and the transverse momentum  $p_T$  of the jet. Jets arising from light quarks will have different masses than jets arising from gluons. After taking relative production cross-sections for quark and gluon jets into account at different energy



Figure 5.6: Radial separation between top decay products vs. top-quark  $p_T$  for top quarks produced from a theoretical heavy resonance, Z' [51].

scales, the resulting relationship between jet  $p_T$  and mass at next-to-leading order (NLO) is well-approximated by [52]:

$$m \approx 0.2 p_T R \tag{5.5}$$

Additional contributions to jet mass come from initial state radiation (ISR), pileup (PU), the underlying event (UE), and multi-parton interaction (MPI). A variety of so-called grooming techniques are used to remove the dependence on this unassociated radiation, so that the groomed jet mass depends only on the mass of the boosted object. Trimming is a grooming method applied to fat jets after they have been clustered, typically with the anti- $k_T$ algorithm. Trimming is applied separately to each fat jet, and consists of the following steps:

- 1. Recluster the constituents with a smaller radius parameter,  $R_{sub}$ , resulting in one or more subjets
- 2. Reject any subjet with  $p_T < f_{cut} p_T^{jet}$
- 3. Sum the four-vectors of the subjets that are not rejected to form the final trimmed jet

In the first step, a radius parameter and clustering algorithm must be chosen. The subjet radius parameter must be smaller than the original radius parameter. Often, the  $k_T$ 

algorithm is used for subjet clustering, in order to preserve as much of the FSR as possible, because rejecting FSR results in reduced jet mass resolution [53]. The values of  $R_{sub}$  and  $f_{cut}$ can be chosen to reduce pileup sensitivity and maximize jet mass resolution for the signal of interest. The trimming procedure is illustrated in figure 5.7.



Figure 5.7: Illustration of the trimming procedure, one of several grooming methods used to remove the contribution of unassociated radiation to fat jet mass [54].

The effect of different grooming techniques, including trimming, on jet mass resolution can be seen in figure 5.8. For dijet events, high-mass jets are highly suppressed by grooming, which leads to a better signal-to-background ratio for signals of hadronically decaying boosted objects with QCD dijet backgrounds. For  $t\bar{t}$  events, the top-mass peak is somewhat visible before grooming, although much wider than after grooming. The W-mass peak, which is originally completely hidden due to unassociated radiation effects, can be recovered with any of the grooming methods.

# 5.4 Reconstruction in ATLAS

The goal of jet measurements in ATLAS is to capture and reconstruct both the energy and momentum of jets leaving the collision point. Calorimeter jets use clusters of calorimeter cell hits as inputs to the clustering algorithms described in 5.2.

#### 5.4.1 Topological cell clusters

Topological cell clusters, or topo-clusters, are three-dimensional clusters of calorimeter cell energy measurements. By clustering groups of calorimeter cells into topo-clusters, the number of inputs to the jet reconstruction algorithm is reduced. Topo-clustering also reduces calorimeter noise by rejecting cell signals not associated with other nearby significant cell signals [56]. Topo-clusters serve as the main input to the jet clustering algorithms when



Figure 5.8: Jet mass distributions from simulated LHC proton-proton collisions events before and after grooming, for various grooming techniques. HERWIG is used to simulate  $\sqrt{s} =$ 7 TeV QCD dijet events (left) and Standard Model  $t\bar{t}$  events (right) with parton-level  $p_T$ between 500 and 600 GeV [55].

reconstructing calorimeter jets in ATLAS . The two sources of calorimeter cell noise are electronic noise and pileup.

The algorithm is based on the signal-to-noise ratio in each cell,  $\zeta_{cell} = E_{cell}/\sigma_{cell}$ . There are three tunable parameters used in the algorithm, labeled S, N, and P, where  $S > N \ge P$ . For purposes of the algorithm, a cell is considered to be a *neighbor* of another cell if the two cells are adjacent in the same layer, or if they are in adjacent layers and have any overlap [56].

The algorithm is defined as follows:

- 1. Label any cell with  $\zeta_{cell} > S$  a seed cell.
- 2. Collect all neighboring cells to a seed cell into a proto-cluster.
- 3. If any cell in the proto-cluster has  $\zeta_{cell} > N$ , collect its neighbors into the proto-cluster.
- 4. Repeat the previous step until the set of neighbors collected has  $\zeta_{cell} < N$ .
- 5. Reject any cells with  $\zeta_{cell} < P$ .

At any time in the algorithm, if any cell with  $\zeta_{cell} > N$  belongs two proto-clusters, then the two proto-clusters are merged [56]. Once the algorithm terminates, the final set of topo-clusters are used as the inputs to the desired jet clustering algorithm.

A result of this algorithm is that isolated cells with  $\zeta_{cell} < N$  are rejected, which reduces that amount of noise entering the jet clustering algorithm. These cells are less likely to contain signal than cells with  $\zeta_{cell} < N$  that neighbor cells with higher signal-to-noise ratio. This clustering algorithm does not guarantee that all the energy of a given particle in the jet will be captured by a single topo-cluster, nor does it guarantee that a single topo-cluster only contains energy from a single particle. So topo-clusters are not measurements of individual particles in the jet.

The three parameters S, N, and P are tuned on test-beam data with known energy to maximize the measured energy while minimizing the energy resolution [57]. Results of the test beam energy measurements with 180 GeV and 20 GeV pions can be seen in 5.9. Based

(a) Average of the energy deposited in all clusters(b) Resolution of the distribution of the energy deposited in all clusters by 180 GeV pions
 (b) Resolution of the distribution of the energy deposited in all clusters by 180 GeV pions



(c) Average energy deposited in all clusters by 20 GeV pions

(d) Resolution of the distribution of the energy deposited in all clusters by 20 GeV pions



Figure 5.9: Average energy and energy resolution measured for 180 GeV and 20 GeV pions with different values of the topo-clustering parameters S, N, and P. The x-axis of each plot represents choices for all three parameters, and the y-axis indicates either the average energy deposited in each cluster  $\langle E_{clusters} \rangle$  or the energy resolution  $RMS/ \langle E_{clusters} \rangle$  [57].

on these measurements, the default parameter values chosen for ATLAS topo-clustering are S = 4, N = 2, P = 0. The topo-cluster four momenta are used as the input objects to the jet clustering algorithms. Other properties of the topo-clusters, known as cluster moments, are used in calibration of the topo-clusters.

#### 5.4.2 Topo-cluster calibration

Calibration of topo-clusters is performed before inputting them into the jet clustering algorithms. Calibration is needed in order to correct for the calorimeters' differing response to electromagnetic and hadronic showers, to correct for signal loss arising from the clustering algorithm, and to account for signal loss due to inactive material. The calibration strategy known as Local Cell Weighting (LCW) is used to correct for all three of these. Calibration is done using Monte Carlo simulations of neutral and charged pions in a simulated detector. Pions with energies up to 2 TeV and over a range of  $\eta$  values are used.

The ATLAS calorimeters are non-compensating calorimeters, which means they have a lower response for hadronic showers than for electromagnetic showers, for the same energy deposited. As a result, separate calibrations are needed for hadronic and electromagnetic showers. A classification algorithm is used to determine the probability that each topo-cluster arose from an electromagnetic shower. This probability is called  $P^{EM}$ . The calibration weight for cell *i* in cluster *j* is then:

$$w_{i} = P_{i}^{EM} w_{i}^{EM} + \left(1 - P_{i}^{EM}\right) w_{i}^{had}$$
(5.6)

where  $P_j^{EM}$  is the predicted probability that cluster j arose from an electromagnetic shower,  $w_i^{EM}$  is the calibration weight for cell i assuming the electromagnetic response, and  $w_i^{had}$  is the calibration weight for cell i assuming the hadronic response [56].

Cluster classification probabilities are determined using the topo-cluster depth and signal density, as hadronic showers tend to deposit energy deeper in the calorimeter and have lower energy density [56]. Figure 5.10 shows the EM classification probability as a function of these two topo-cluster moments. The cell signal density is measured as energy density of the cell normalized to the total energy of the cluster to which it belongs.

The hadronic calibration weight is then determined as:

$$w_i^{had} = \frac{E_i^{dep}}{E_i^{EM}} \tag{5.7}$$

where  $E_i^{dep}$  is the true energy deposited in the cell, and  $E_i^{EM}$  is the energy measured in the cell.

During the topo-clustering process, cells with signal fractions belows the necessary thresholds can be rejected, resulting in a certain amount of signal energy being lost. Lost cells with true signal energy deposited are attributed to nearby clusters. The relevant search area for a cluster depends on the cluster  $\eta$ , and ranges from 14 deg to 60 deg [56]. Lost cells can be attributed to more than one cluster, with a weight proportional to the energy deposited in each cluster. An out-of-cluster correction weight is then calculated as

$$w_j^{ooc} = \frac{E_j^{ooc} + E_j^{dep}}{E_j^{dep}}$$
(5.8)



Figure 5.10: Cluster EM probability as a function of normalized cell signal density and depth. Cell that are farther away from their cluster's center, and with lower energy density are more likely to arise from hadronic showers. The red line indicates the 50% boundary: cells falling on that line have a 50% probability of arising from electromagnetic showers, while clusters below that line have a higher probability [56].

where  $E_j^{ooc}$  is the total energy deposited in lost cells associated to the cluster, including a fraction of the energy from lost cells shared by multiple clusters. This correction is determined separately for electromagnetic and hadronic showers, using neutral and charged pions respectively[56]

Dead material is also accounted for using simulated pions in a simulated detector. Figure 5.11 shows an illustration of the lost cell and dead material calibration procedure.

#### 5.4.3 Jet calibration

For large-R jets, the LCW calibration method is applied to topological clusters as described in 5.4.2 before inputting the topo-clusters into the jet clustering algorithm. After jet clustering, further calibration is required on the resulting jets in order to ensure that the measured energy matches the true energy deposited in the detector. This is known as the jet energy scale, or JES, calibration. Several different energy scales are defined in order to describe topo-clusters or jets at different stages in the calibration process. Those scales are defined



Figure 5.11: Illustration of the procedure used to account for signal lost to lost cells and dead material [56].

as follows:

- EM scale: energy as measured directly by the calorimeters
- LCW scale: energy after applying the LCW method
- EM+JES scale: energy after applying jet-level calibration, if LCW was not used
- LCW+JES scale: energy after applying LCW and jet-level calibrations

#### **JES** Calibration

For the JES calibration of large-R jets, the energy correction procedure described in [58] is applied. Unlike in [58], however, the Monte Carlo used for calibration does include pileup. The reason for this is that large-R jets do not receive a pileup-specific calibration before the JES calibration, unlike standard jets [51]. The calibration is done by measuring the JES response in bins of true jet energy ( $E_{truth}$ ) and  $\eta$ . The response is:

$$\mathcal{R} = E_{meas} / E_{truth} \tag{5.9}$$

For each  $(E_{truth}, \eta)$  bin, the average response  $\langle \mathcal{R} \rangle$  and average measured energy  $\langle E_{meas} \rangle$  are measured. A one-dimensional calibration function  $\mathcal{F}_k(E_{meas})$  is then obtained by fitting to the distribution of  $\langle E_{meas} \rangle$  vs.  $\langle \mathcal{R} \rangle$  in each  $\eta$  bin k [58]. Once the calibration functions  $\mathcal{F}_k(E_{meas})$  are determined from Monte Carlo, they are applied to jets measured in data. The calibrated energy for a jet measured with  $\eta$  in bin k is:

$$E_{calib} = \frac{E_{meas}}{\mathcal{F}_k \left( E_{meas} \right)} \tag{5.10}$$

[58]

#### **JMS** Calibration

For large-R jets, the jet mass scale (JMS) must be calibrated in addition to the jet energy scale. A similar procedure to that described in 5.4.3 is applied to the JMS calibration, using the jet mass response in place of the jet energy response. The jet mass response is defined analogously to the jet energy response:

$$\mathcal{R}^{mass} = m_{meas}/m_{true} \tag{5.11}$$

The jet mass response before and after calibration can be seen in figure 5.12.



Figure 5.12: Jet mass response as a function of  $\eta$  before and after JMS calibration, for anti- $k_T$  jets with R = 1.0 at the LCW scale, without grooming [51].

#### 5.4.4 *b*-tagging

The identification of jets containing b hadrons, referred to as b-jets, is very important for many different ATLAS physics analyses. B-tagging is the process of identifying these jets and distinguishing them from jets that do not contain b hadrons. Jets containing c hadrons can also be distinguished using similar methods. Jets that don't contain either b hadrons or c hadrons are known as light-flavor jets. B-tagging and c-tagging are collectively known as flavor tagging. B-tagging is used in this analysis to suppress the dominant QCD multijet contribution to the background, in order to increase the signal to background ratio.

The main property of b hadrons that allows b-jets to be distinguished from light flavor jets is their relatively long lifetime, on the order of picoseconds. Because of their long lifetime, b hadrons can travel a measurable distance away from the primary interaction point before decaying. For example, a 50 GeV b hadron will travel an average of 3 mm before decaying [59]. Additionally, b-jets are more likely to contain muons, a fact which can be exploited by b-tagging algorithms to discriminate between b-jets and light flavor jets [59].

In this analysis, three different b-tagging algorithms are combined into a multivariate algorithm referred to as MV2. Each of the three base algorithms uses the relatively large separation distance of b hadron decay vertices away from the primary vertex.

#### Impact parameter algorithm

In the impact parameter algorithms, IP2D and IP3D, individual tracks belonging to a jet are traced back towards the interaction point, and their distance of closest approach to the interaction point, known as the impact parameter, is calculated. Both the transverse impact parameter,  $d_0$ , and the longitudinal impact parameter,  $z_0 \sin \theta$  are calculated. Their significances,  $d_0/\sigma_{d_0}$  and  $z_0 \sin \theta/\sigma_{z_0 \sin \theta}$  are also calculated. The IP significances are considered positive if the point of closest approach is on the same side of the primary vertex as the direction of the jet, and negative otherwise. IP2D uses only transverse IP significance, and IP3D uses both transverse and longitudinal IP significance. Histograms of the IP significances are calculated using a simulated  $t\bar{t}$  sample for b-jets and light flavor jets separately. These histograms serve as the probability density functions (PDFs) from which a likelihood ratio discriminant is constructed by summing over the likelihood ratios for all tracks in a jet [60].

Figure 5.13 shows the distribution of transverse and longitudinal impact parameter significances for both heavy quark (b and c) and light quark jets from a simulated  $t\bar{t}$  sample. Large positive IP significances are much more likely to occur for b-jets than for light jets. Large negative IP significances are more likely to be from noise, and so do not provide good signal discrimination.

#### Secondary vertex algorithm

The secondary vertex algorithm (SVX) also makes use of the long b hadron lifetime and mean flight path. Rather than calculating impact parameters for individual tracks, multiple tracks



Figure 5.13: Transverse and longitudinal impact parameter significance distributions for a sample of simulated  $t\bar{t}$  events [60].

are traced back to find where they intersect other tracks, in order to reconstruct secondary vertices [61]. Some filtering is done to reject long-lived particles and other background processes. Candidate tracks are required to be a certain significant distance from the primary vertex, and the tracks have to pass a goodness-of-fit requirement. Figure 5.14 shows the secondary vertex reconstruction rate versus jet  $p_T$  and jet  $\eta$  for b-jets, c-jets, and light flavor jets in a simulated sample of  $t\bar{t}$  events. B-jets are much more likely than light flavor jets to contain a reconstructed secondary vertex that passes all of the selection criteria outlined in [60].

#### Multi-vertex algorithm

The third algorithm used to identify b-jets from their mean flight distance from the primary vertex is the multi-vertex reconstruction algorithm, also known as JetFitter. Rather than reconstruct a single secondary vertex, JetFitter attempts to reconstruct the entire b hadron decay chain [62]. Unlike the SVX algorithm, JetFitter finds multiple vertices that are required to lie in a straight line, along the proposed flight path of the b hadron. The difference between the SVX and JetFitter algorithm is illustrated in figure 5.15 The efficiency to reconstruct a b hadron decay vertex as a function of jet  $p_T$  and  $|\eta|$  for b-jets, c-jets, and light flavor jets can be seen in 5.16



Figure 5.14: Secondary vertex reconstruction rate versus  $p_T$  and  $\eta$  for b-jets, c-jets, and light flavor jets from a simulated sample of  $t\bar{t}events$  [60].



secondary vertex finder fits all displaced tracks to an inclusive vertex.

(b) JetFitter performs a multi-vertex fit using the b-hadron flight direction constraint.

Figure 5.15: Illustration of the difference between the secondary vertex (a) and multi-vertex (b) algorithms [62].

#### Multivariate algorithm

To get the maximum b-tagging performance, all three of the above mentioned algorithms are combined into a multi-variate algorithm known as MV2. The algorithm is a boosted decision tree (BDT) which combines the inputs from all three base algorithms mentioned



Figure 5.16: JetFitter vertex reconstruction rate versus  $p_T$  and  $\eta$  for b-jets, c-jets, and light flavor jets from a simulated sample of  $t\bar{t}$  events [63].

above. The BDT is trained on a training sample for which b-jets are labeled as signal, and a mixture of light jets and c-jets are labeled as background [63]. Different versions of the BDT can be obtained by varying the fraction of c-jets and light quark jets in the background training sample. In this analysis, a mixture of 80% light quark jets and 20% c-jets are used in training the BDT, resulting in a discriminant known as MV2c20 [63]. In addition to the inputs from the three base algorithms, MV2 also includes the  $p_T$  and  $\eta$  of the jet. There are a total of 24 features input to the BDT, summarized in table 5.1. The output of the MV2c20 BDT for b-jets, c-jets, and light flavor jets from a simulated sample of  $t\bar{t}$  events can be seen in figure 5.17.

The BDT reduces the selection criteria from 24 variables to a single one-dimensional discriminant. The performance of the algorithm in terms of b-tagging efficiency and light flavor jet rejection depends on the choice of threshold value for this discriminant. Figure 5.18 shows the b-tagging efficiency versus c-jet and light-jet rejection for the range of MV2 output cut values. As already mentioned, MV2c20 is trained with 20% c-jets in the background sample. Comparing this to MV2c00, in which only light jets are included in the background training sample, there is clearly improved light-jet rejection by including the c-jets in the background training sample.

In this analysis, the threshold value is chosen such that the b-jet efficiency is 70%, resulting in a c-jet rejection of 8.1,  $\tau$  rejection of 26, and light flavor rejection of 440 [63]. Small-Rjets are used in this analysis because the *b*-tagging algorithm is only calibrated for jets with R = 0.4.

Variable	Description
$p_T(jet)$	Jet transverse momentum
Kinematics $\eta(jet)$	Jet pseudo-rapidity
$\log(P_b/P_{\text{light}})$	Likelihood ratio between the <i>b</i> - and light jet hypotheses
$\log(P_b/P_c)$	Likelihood ratio between the b- and c-jet hypotheses
$\log(P_c/P_{\text{light}})$	Likelihood ratio between the c- and light jet hypotheses
m(SV)	Invariant mass of tracks at the secondary vertex assuming
	pion masses
$f_{\rm E}({ m SV})$	Fraction of the charged jet energy in the secondary vertex
$N_{\text{TrkAtVtx}}(\text{SV})$	Number of tracks used in the secondary vertex
$N_{2\mathrm{TrkVtx}}(\mathrm{SV})$	Number of two track vertex candidates
$L_{xy}(SV)$	Transverse distance between the primary and secondary
,	vertices
$L_{xyz}(SV)$	Distance between the primary and secondary vertices
$S_{xyz}(SV)$	Distance between the primary and secondary vertices di-
-	vided by its uncertainty
$\Delta R(\text{jet}, \text{SV})$	$\Delta R$ between the jet axis and the direction of the secondary
	vertex relative to the primary vertex
$N_{2\mathrm{TrkVtx}}(\mathrm{JF})$	Number of 2-track vertex candidates (prior to decay chain
	fit)
m(JF)	Invariant mass of tracks from displaced vertices assuming
	pion masses
$S_{xyz}(JF)$	Significance of the average distance between the primary
-	and displaced vertices
$f_{\rm E}({ m JF})$	Fraction of the charged jet energy in the secondary vertices
$N_{1-trk vertices}(JF)$	Number of displaced vertices with one track
$N_{\geq 2-\text{trk vertices}}(\text{JF})$	Number of displaced vertices with more than one track
N <sub>TrkAtVtx</sub> (JF)	Number of tracks from displaced vertices with at least two
	tracks
$\Delta R(\vec{p}_{\rm jet},\vec{p}_{\rm vtx})$	$\Delta R$ between the jet axis and the vectorial sum of the mo-
	menta of all tracks attached to displaced vertices
	$Variable$ $p_{T}(jet)$ $\eta(jet)$ $log(P_b/P_{light})$ $log(P_c/P_{light})$ $m(SV)$ $f_{E}(SV)$ $N_{TrkAtVtx}(SV)$ $L_{xyz}(SV)$ $L_{xyz}(SV)$ $\Delta R(jet, SV)$ $N_{2TrkVtx}(JF)$ $m(JF)$ $S_{xyz}(JF)$ $f_{E}(JF)$ $N_{1-trk vertices}(JF)$ $N_{TrkAtVtx}(JF)$ $\Delta R(\vec{p}_{jet}, \vec{p}_{vtx})$

Table 5.1: Description of each of the 24 inputs for the MV2 BDT [63].



Figure 5.17: MV2c20 BDT output for b-jets, c-jets, and light flavor jets from a simulated sample of  $t\bar{t}$  events [63].



Figure 5.18: B-tagging efficiency versus light-jet (a) and c-jet (b) rejection for a range of MV2 output cut values. The red curves are for MV2c20, in which 20% c-jets are included in the background training sample. The blue curves are for Mv2c00, which only includes light jets in the background training sample. The improved c-jet rejection and worsened light-jet rejection can seen by comparing the two [63].

# Chapter 6

# Analysis Overview

# 6.1 The RPV UDD signature

This analysis is a search for pair-produced gluinos decaying to a large number of standard model quarks via R-parity-violating decays, as described in 3.3.4. In the direct decay model, each gluino decays directly to three standard-model quarks via an effective vertex with an off-shell squark propagator. In the cascade decay model, each gluino first decays to a neutralino and two quarks, and the neutralino then decays similarly to three quarks vai the RPV effective vertex. Both types of signal events would result in a large number of high- $p_T$  jets in the detector, because they have a large number of quarks in the final state, and each quark must be boosted due to the large mass of the gluinos. Figure 6.1 shows the diagrams for the two signal models under consideration.

# 6.2 Previous searches and limits

A similar search was performed using 20.3  $fb^{-1}$  of 8 TeV ATLAS data from Run 1, and limits were set on the gluino and neutralino masses [64]. Two different strategies were employed in this analysis. The first strategy looked for an excess of events with high small-Rjet multiplicities. Signal regions required  $\geq 6$  or  $\geq 7$  jets, and different *b*-jet multiplicity requirements were applied to create signal regions sensitive to different heavy-flavor branching fractions. In the jet-counting analysis, backgrounds were estimated by extrapolating event yields from lower jet multiplicity regions, using a scaling factor derived from multijet Monte Carlo. This strategy was mainly aimed at the direct-decay model.

The second strategy used the sum of large-R jet masses as discriminating variable, and was mainly aimed at the cascade-decay model. This strategy derived jet mass templates from signal-depleted control regions, and used those templates to generate the estimated background mass distribution in the signal regions.

For the cascade decay model, the observed limit on the gluino mass ranged from 800 GeV to 1 TeV, depending on the neutralino mass. Figure 6.2 shows the 95% CL lower bounds in



Figure 6.1: Diagrams for the two decay processes that are the subject of this search. The direct decay (left) and cascade decay (right) both involve effective RPV vertices containing off-shell squark propagators.

the  $(m_{\tilde{g}}, m_{\tilde{\chi}})$  plane from the search for the cascade-decay signal. For the direct decay model, different limits were placed on the gluino mass for different assumptions about the flavor composition of the final states. In the case of light-quark only decays, gluino masses above 917 GeV were excluded [64]. Figure 6.3 shows the 95% CL upper bounds on the cross-section for the direct decay model over a range of gluino masses, for various assumptions about the flavor composition of the final states.

In Run 1, CMS also performed a search for RPV decays of gluinos with high jetmultiplicity final states [65]. The analysis assumed that  $\lambda_{tbs}''$  is the largest of the UDD coupling constants, motivated by minimal flavor-violating (MFV) SUSY [66]. Based on this assumption, the analysis specifically looked at the direct-decay model, in which the gluinos each decay to exactly one top, one bottom, and one strange quark, via an off-shell stop. The analysis looked at events with zero or one-lepton final states high jet multiplicity, using *b*-jet multiplicity as a discriminating variable[65]. For the fully-hadronic channel, the dominant background was QCD multijets, and for the single-lepton channel, the dominant background was from  $t\bar{t}$ . In both cases, backgrounds were estimated from Monte Carlo with corrections derived from control-region data. The single-lepton channel ended up setting the strongest limit on the gluino mass, excluding  $pp \to \tilde{g}\tilde{g} \to tbs$  for gluinos with mass less than 1.03 TeV [65].

The increased luminosity and center-of-mass energy of collisions in Run 2 creates an opportunity to extend the reach of the search for RPV SUSY in multijet final states. Figure 11.2 shows the expected limit for gluino masses under the cascade decay scenario increasing to



Figure 6.2: Observed and expected limits in the  $(m_{\tilde{g}}, m_{\tilde{\chi}})$  plane for the cascade-decay model in ATLAS Run 1 with  $\sqrt{s} = 8 TeV$  [64].

up to  $1.9 \ TeV$ , and up to  $1.2 \ TeV$  under the direct-decay scenario.

### 6.3 Search strategy and discriminating variables

The search strategy is similar to the total-jet-mass analysis from ATLAS Run 1 [64]. Signal regions include events with a large number of high-mass large-R jets, and backgrounds are estimated using jet mass templates derived from control region data. The primary discriminating variable is  $M_J^{\Sigma}$ , defined as:

$$M_J^{\Sigma} = \sum_{j=1}^4 m_{jet}^j \tag{6.1}$$

where  $m_{jet}$  is large-R jet mass, and the sum is over the first four highest- $p_T$  jets in the event. Large-R jets are reconstructed with R = 1.0 and are required to have  $p_T > 200 \text{ GeV}$  and  $|\eta| < 2.0$ . In the case where an event has fewer than 4 large-R jets passing the kinematic thresholds, the sum over all large-R jets in the event is used. This observable is sensitive to the signal because there are large numbers of quarks in the final state, and each quark is likely to have high  $p_T$  due to the large gluino mass. Background events, which mainly come



Figure 6.3: Observed and expected limits on  $m_{\tilde{g}}$ ), along with theoretical cross-section for the direct-decay model in ATLAS Run 1 with  $\sqrt{s} = 8 \ TeV$  [64]. Different limits are set for four different branching fraction scenarios. In sub-figure (a), only light-quark decays are allowed. In sub-figure (b), each gluino decay results in a top quark in the final state. In sub-figure (c), each gluino decay results in a bottom quark in the final state. In sub-figure (d), each gluino decay results in bottom quark in the final state [64].

from QCD multijet production, tend to have much lower jet mass on average, as explained in 5.3. Unlike with the boosted scenarios described in 5.3, the jet mass for signal events does not come from capturing the decay products of a single boosted heavy object in an individual jet. Instead, the high jet mass in signal events results from the high probability that two or more decay products, possibly from different parent particles, can accidentally overlap inside the same large-R jet.  $M_J^{\Sigma}$  provides good separation between signal and background because it takes into account both the energy and angular structure of an event, unlike a purely energy-dependent observable like  $H_T$  [67, 68]. The second discriminating variable is  $|\Delta \eta_{12}|$ , the pseudorapidity difference between the first two leading jets in an event.

For events with high jet multiplicity, the signal has smaller  $|\Delta \eta_{12}|$  than the background. Distributions of the two discriminating variables are shown in figure 6.4 for data as well as background and signal Monte Carlo. The different event generators all show similar  $M_J^{\Sigma}$ distributions for QCD multijets, and a clear difference can be seen between the signal and background distributions. For signal events, the location of the  $M_J^{\Sigma}$  peak depends on the gluino mass, moving to higher values for larger masses. The shape of the  $|\Delta \eta_{12}|$  distribution does not depend as strongly on the gluino mass, but there is still a clear difference between the signal and background distributions. In the analysis, a requirement of small  $|\Delta \eta_{12}| < 1.4$ can therefore be used to further suppress the background.



Figure 6.4: Distributions of the two main discriminating observables, (a) the scalar sum of the four leading large-R jets,  $M_J^{\Sigma}$  and (b) the difference in pseudorapidity between the two leading jets,  $|\Delta\eta_{12}|$ . Selected events have  $\geq 4$  large-R jets. Distributions are shown for both data and simulated signal and background samples. The red and green signal distributions are for the cascade decay mode, and the violet distribution is the direct decay mode, for the superpartner masses indicated [69].

A data-driven method is used to predict the background yield in the signal regions, as well as the uncertainties on those predictions. The method assumes that for background events, the probably distribution for the mass of an individual jet depends mainly on the  $p_T$ ,  $\eta$ , and flavor of that jet, and does not depend strongly on other details of the event kinematics. Using this assumption, templates for jet mass can be built from jets in a background-dominated region and used to predict the mass distribution for jets in signal regions. Templates are binned in  $p_T$  and  $\eta$ , and are derived separately for b-matched and non-b-matched jets. Each template is the histogram of jet mass within a bin. The histogram is used as the jet mass probability distribution, conditioned on  $p_T$ ,  $\eta$ , and flavor. Randomized jet masses, known as dressed masses, are generated from these templates for each jet in the signal region sample. Summing the dressed masses for each of the up to four leading jets in an event gives the dressed  $M_J^{\Sigma}$  for that event. The dressed  $M_J^{\Sigma}$  distribution for each signal region is used to estimate the expected background contribution to that region. This method depends on the assumption that the jet mass in background events depends only on  $p_T$ ,  $\eta$ , and flavor. Independent regions are defined for measuring the extent to which this assumption is violated in data, and this will serve as a systematic uncertainty on the background estimation. These regions will be called uncertainty determination regions (UDRs). Background-dominated validation regions will also be defined to test that the dressed  $M_J^{\Sigma}$  distributions are within the uncertainties derived from the UDRs.

The method is similar to that used in the Run-1 version of the analysis [64], with a few important differences. In the Run-1 analysis, the templates were smoothed with a kernel density estimate before sampling the dressed masses. In this version of the analysis, the templates remain binned. This allows for an estimate of the statistical uncertainty from the control sample size. By Poisson fluctuating each template bin before sampling, the statistical uncertainty is propagated to the dressed  $M_J^{\Sigma}$  distributions. Secondly, in the Run-1 version of the analysis, two separate sets of templates were generated: one set for the leading two jets in each event, and a separate set for the third leading jet. In this analysis, the templates are instead divided into b-matched and non-b-matched jets. The discrepancy in template shape between b-matched and non-b-matched jets was seen to be larger than that between the third leading jet and first two leading jets. Finally, the Run-1 version of the analysis used Monte-Carlo non-closure as one contribution to the background systematic uncertainty. In this analysis, a data-driven method is used instead. This is due to the fact that the sample size of available simulated data was not large enough to make an accurate estimate of the non-closure. More details of the template-based background estimation method and how uncertainties are derived from data will be given in 9.1.

Once the data-driven background estimation and uncertainties have been obtained, a comparison can be done between the observed and expected yields in the signal regions. If no significant excess is observed, limits can be set on gluino and neutralino masses under the cascade and direct decay model assumptions. Limits are set by first calculating a 95% CL upper bound on the signal production cross-section, and then identifying the mass points for which the theoretical cross-section lies outside this bound. The details of the limit-setting procedure are described in 11.2.

# Chapter 7

# Monte Carlo Models of Signal and Background

# 7.1 Signal modeling

Simulated gluino pair-production events are generated for use in the analysis. Separate samples are generated for the direct decay and cascade decay models described in 3.3.4. For the direct decay model, events are generated for gluino mass ranging from 900 GeV to 1.8 TeV. For the cascade model, the gluino mass is varied from 750 GeV up to 2.1 TeV, with the neutralino mass ranging from 450 GeV to 1.9 TeV. For all signal points, the neutralino mass is lower than the gluino mass, as is required by the decay modes of interest. Signal events are simulated for a discrete set of mass points. Figure 7.1 indicates the  $(m_{\tilde{g}}, m_{\tilde{\chi}_1^0})$  points for which signal evens were generated. This grid of signal points is chosen to cover the range of masses that the analysis is expected to be sensitive to. The grid points are spaced closer together in the high- $m_{\tilde{g}}$  region, near where the expected limits (11.2) lie.

### 7.1.1 Event generation

Matrix elements for gluino pair production are generated with MADGRAPH\_AMC@NLO v2.3.3, which is a convenient framework for calculating arbitrary matrix elements at up to next-to-leading order (NLO) [70]. MADGRAPH allows the user to specify the theory being used and values for the free parameters, then it constructs and calculates the necessary Feynman diagrams. In this analysis, matrix elements are calculated at leading order (LO). In the pair-production matrix elements, up two two additional partons are allowed to be produced along with the gluino pair.

To model the parts of the collision environment described in section 5.1, such as the underlying event, parton shower, and fragmentation, PYTHIA 8.186 is used [71]. Interfacing MADGRAPH and PYTHIA is made possible by a standardized file format known as Les Houches Event Format (LHEF) [72].



Figure 7.1: The grid of gluino and neutralino masses for which simulated cascade decay events were generated. The triangles indicate the mass points for which simulated events were generated.

As discussed in section 5.1, there are non-perturbative processes that occur in protonproton collisions, which cannot be directly calculated from first principles. This processes can be parameterized and approximated in a program like PYTHIA, requiring many free parameters with values that must be constrained using data. The process of inferring these parameters so that the calculations match the data is known as tuning. For this analysis the A14 set of tuned parameter values [73, 74] and the NNPDF2.3LO set [75] of PDFs are used to model the underlying event in PYTHIA. EvtGen v1.2.0 [76] is used for modelling b- and c-hadron decays.

#### 7.1.2 Cross-section

In order to estimate the expected signal yield and uncertainty, the nominal production cross section and its uncertainty must be known. Gluino pair-production cross-sections have been calculated at a range of center-of-mass energies, and the results are summarized in [77]. The gluino pair-production cross-section and its uncertainty over a range of gluino masses can be seen in 7.2. The uncertainty band comes from calculating the cross section with different



PDF sets and with different factorization and renormalization scales [77].

Figure 7.2: Gluino pair-production cross-section and uncertainty for  $\sqrt{13} TeV$  proton-proton collisions at the LHC, calculated at NLO+NLL [77].

#### 7.1.3 Fast simulation with Atlfast-II

Simulation of the detector response to physics events requires large amounts of computational resources. In order to reduce the cost of generating large samples of simulated events, a faster version of the detector response simulation, called Atlfast-II has been developed [78]. Fast simulation with Atlfast-II (AFII) can greatly reduce the time required to generate these simulated events, but makes several simplifications that can lead to less precise results.

A comparison between the full detector simulation (FullSim) and AFII was performed to find out if using AFII would reduce the sensitivity of the search or otherwise affect the analysis. Very good agreement between FullSim and AFII-simulated events is observed across a range of kinematic observables relevant to this analysis. As a result, AFII simulation is used for generating signal events for all of the grid points used in the limit-setting. Full simulation events are also generated for a few representative signal points.

For the FullSim/AFII comparison, full simulation events for the decay modes described in chapter 6 are not available, but a very similar signal is used. In this case, pair-produced gluinos each decay to a neutralino  $+ t\bar{t}$ , and the neutralinos each decay through a  $\lambda''$  vertex to three light quarks. This process is illustrated in 7.3. The benchmark model used in this study has  $m_{\tilde{g}} = 1250 \ GeV$  and  $m_{\tilde{\chi}} = 450 \ GeV$ . Events used in this study are required to have  $H_T > 1 \ TeV$  and lead jet  $p_T > 200 \ GeV$ . Large-*R* jets are fully calibrated using FullSim calibration and required to have  $p_T$  of at least 100 GeV and  $|\eta|$  less than 2.8. Jet mass response is defined as  $m_{reco}/m_{truth}$ , where  $m_{reco}$  is the reconstructed jet mass and  $m_{truth}$ is the mass of jets clustered from truth particles.

Figure 7.4 shows the large-R jet mass response vs. truth mass and vs. truth  $p_T$  for large-R jets with  $p_T > 100 \ GeV$  and  $|\eta| < 2.8$ . The points and error bars show the best-fit mean and width for a Gaussian fit to the core of the distribution in each bin. Figure 7.5 shows the distribution of large-R jet kinematic observables  $m, \eta, \phi$ , and  $p_T$  for both FullSim and AFII, along with the ratio of each bin in the bottom panel of each plot. Figure 7.6 shows the large-R jet multiplicity histogram for reconstructed jets using both AFII and FullSim, and the distributions of  $M_J^{\Sigma}$  for events with at least 5 large-R jets. In all cases, very good agreement is observed between FullSim and AFII distributions.



Figure 7.3: The process used for the AFII/FullSim comparison study. The signal is very similar to the cascade decay model used in the search, but the gluinos decay to a neutralino+ $t\bar{t}$  rather than a neutralino and two light quarks.



Figure 7.4: Jet mass response vs true jet mass (left) and truth jet  $p_T$  for simulated signal events using both the full detector simulation (full sim) and Atlfast-II (fast sim). Points and error bars show the best-fit mean and width for a Gaussian fit to the core of the distribution in each bin. The model used for this study is similar to the cascade decay model described in 6, except each gluino decays to a neutralino +  $t\bar{t}$  rather than a neutralino and two light quarks. The gluino mass is 1250 GeV and neutralino mass is 450 GeV. Jets are required to have  $p_T$  greater than 100 GeV and  $|\eta| < 2.8$ .

# 7.2 Background modeling

Background estimation for this analysis is done with a fully data-driven method that does not rely on Monte Carlo to make predictions. However, simulated background events are generated and studied in order to test and validate the data-driven background estimation method. This process is detailed in 9.4.

The Pythia8 multijet background sample is generated using Pythia8 8.186 [71]. Like with the signal samples described in 7.1.1, the A14 underlying-event tune [71, 73] and the NNPDF2.3LO PDF sets are used [75]. Multijet samples are also generated with Sherpa and Herwig++. The  $t\bar{t}$  events are generated with Powheg-Box v2 [79] with the CT10 PDF set [80].

The differential multijet cross section falls drastically with increasing  $p_T$ , so multijet events are not generated with a probability distribution proportional to the differential cross section. If they were, the result would be enormous numbers of events with only low- $p_T$ jets, and very few events with high- $p_T$ , resulting in very large statistical uncertainties for the high- $p_T$  jet events. Instead, multijet events are generated in slices, where each slice has a minimum requirement on the leading-jet  $p_T$ . Approximately equal numbers of events can then be generated for each slice, and re-weighted according to the relative cross-section of each slice. Using this method, the statistical uncertainty is more constant across a wide range of jet  $p_T$ . The lowest- $p_T$  slice that passes the event selection for this analysis has a lead-jet range of 160 GeV - 400 GeV. For this slice, just under 16 million simulated events



Figure 7.5: Kinematic distributions of large-R jets for simulated events using both the full detector simulation (full sim) and Atlfast-II (fast sim). The model used for this study is similar to the cascade decay model described in 6, except each gluino decays to a neutralino  $+ t\bar{t}$  rather than a neutralino and two light quarks. The gluino mass is 1250 GeV and neutralino mass is 450 GeV. Jets are required to have  $p_T$  greater than 100 GeV and  $|\eta| < 2.8$ .

are generated, which corresponds to an effective luminosity of 1.9  $fb^{-1}$ .



Figure 7.6: Jet multiplicity and distribution (left) and  $M_J^{\Sigma}$  for events with at least 5 large-R jets (right) for simulated events using both the full detector simulation (full sim) and Atlfast-II (fast sim). The model used for this study is similar to the cascade decay model described in 6, except each gluino decays to a neutralino +  $t\bar{t}$  rather than a neutralino and two light quarks. The gluino mass is 1250 GeV and neutralino mass is 450 GeV. Jets are required to have  $p_T$  greater than 100 GeV and  $|\eta| < 2.8$ .

# Chapter 8

# Data Sample, Event Selection, and Jet Definitions

### 8.1 Data sample

This analysis uses the entirety of the 2015 and 2016 ATLAS datasets, comprising 36.1  $fb^{-1}$  (±2.1%) of integrated luminosity, with  $\sqrt{s} = 13 \ TeV$ . A good runs list (GRL) is used to select only those data-taking periods in which all detectors are fully functional [81].

### 8.2 Trigger

Events are required to pass an  $H_T$ -based trigger, where  $H_T$  is the scalar sum of the  $p_T$  of all jets in an event. The jets used for the calculation of  $H_T$  in the trigger are level-one jets with  $p_T > 100 \ GeV$ . To pass the trigger, an event must have  $H_T > 1 \ TeV$ . Figure 8.1 shows the trigger efficiency versus large-R jet  $p_T$  threshold for events with  $\geq 4$  and  $\geq 5$  large-R jets. For events with five or more large-R jets, the trigger efficiency is 100% when the jet  $p_T$  threshold is 200 GeV or above. For events with four or more jets, an additional requirement on the leading jet  $p_T$  is needed to ensure full trigger efficiency at this jet- $p_T$  threshold. For the  $\geq 4$ -jet regions, the leading jet will be required to have a  $p_T$  of at least 440 GeV.

## 8.3 Event selection

Events considered by this analysis must meet the following pre-selection criteria:

- In the list of good luminosity blocks in the GRL as described in 8.1.
- No errors in the LAr or tile calorimeters or the inner detector when the events were measured



Figure 8.1: Trigger efficiency versus large-R jet  $p_T$  threshold for events with  $\geq 4$  and  $\geq 5$  large-R jets. For events with  $\geq 4$  large-R jets, and additional requirement of leading-jet  $p_T > 400 \text{ GeV}$  ensures 100% trigger efficiency.

- Pass the  $H_T$  trigger as described in 8.2.
- At least one primary vertex from at least two tracks with  $p_T > 400 MeV$  each
- At least one large-R jet (see 8.4) with  $p_T > 440 \ GeV$

# 8.4 Jet definitions, b-tagging, and b-matching

In this analysis, two different types of jets are defined. Large-R jets are reconstructed using the anti- $k_T$  algorithm with R = 1.0 and are trimmed by re-clustering the constituents of each jet with  $R_{sub-jet} = 0.2$  and rejecting any sub-jet with  $p_T^{sub-jet}/p_T^{jet} < 0.05$ . The trimmed large-R jets are required to have  $p_T > 200 \text{ GeV}$ .

Small-*R* jets are reconstructed using the anti- $k_T$  algorithm with R = 0.4. To be considered for b-tagging, a small-R jet must have  $p_T$  of at least 50 GeV and  $|\eta|$  less than 2.5. The fixed-efficiency 70% working point is used for b-tagging.

Only large-R jets are used for the analysis, but the b-tagging algorithm is only calibrated for small-R jets. So a matching procedure is used to identify large-R jets that can be associated with a b-tagged small-R jet. Large-R jets found to be within  $\Delta R = 1.0$  of a b-tagged small-R jet in the same event are referred to as b-matched jets. Jet mass templates are derived separately for b-matched and non-b-matched jets.

For details of how jets are reconstructed, trimmed, and calibrated, as well as the definition of different jet parameters, see chapter 5. The algorithm used to identify *b*-jets is described in chapter 5.4.4 as well as [82, 83].

The control, validation, and signal regions defined in section 8.5 can be segmented into b-tag and b-veto regions. Events with at least one b-tagged small-R jet are considered b-tag events, while those without are labelled as b-veto. When b-tagging is not part of the selection requirements for a region, it is called a b-inclusive region.

# 8.5 Control, signal, validation, and uncertainty determination regions

Events are divided into control, signal, validation, and uncertainty determination regions. Figure 8.2 illustrates how the different regions are used in the analysis. Jet mass templates are constructed from control region (CR) events and used predict the mass distribution in the uncertainty determination region (UDR), validation region (VR), and signal region (SR), as discussed in section 9.1. The discrepancy between predicted and observed jet masses in the UDR is taken as the systematic uncertainty of the background estimation method. Observed and predicted masses in the validation regions can be compared to check that any discrepancy falls within this uncertainty, as a way of validating the method. Finally the observed yields in the signal region can be compared to the predicted background yield in the signal region.

Table 8.1 gives the cuts used to define each of these regions. Control region events are required to have exactly three large-R jets each with  $p_T > 200 \ GeV$ . If a control region event has at least one *b*-matched large-R jet, the additional requirement of  $|\Delta \eta_{12}| > 1.4$  is applied, in order to suppress potential signal contamination.

There are a total of five partially-overlapping signal regions, all of which require  $|\Delta \eta_{12}| < 1.4$ . Regions are defined by the minimum number of large-R jets required  $(N_{jet})$ , by whether or not a b-tagged jet is required to be present in the event, and by a minimum requirement on  $M_J^{\Sigma}$ . There are two signal regions which require four or more large-R jets with  $p_T > 200 \text{ GeV}$ , and three signal regions which require five or more. Four-jet signal regions require the  $p_T$  of the leading large-R jet to be greater than 400 GeV. The b-tag signal regions each require at least one b-tagged small-R jet per event, and are the most sensitive to the RPV gluino direct and cascade decay models. Signal regions without the b-tag requirement are also defined, referred to as inclusive signal regions. These inclusive regions are less sensitive to the RPV gluino decay signal, but can be sensitive to other potential BSM signals.



Figure 8.2: Workflow illustrating how the different regions are used in the analysis.

The minimum value of  $M_J^{\Sigma}$  for each signal region is optimized based on signal sensitivity separately for each signal region. For the 5-jet *b*-tag signal region, the optimal  $M_J^{\Sigma}$  cut is 0.8 TeV for the cascade decay model, and 0.6 TeV for the direct decay model. The signal regions corresponding to these two  $M_J^{\Sigma}$  cuts will be referred to as 5jSRb1 and 5jSRb2, respectively.

The difference between the signal and validation regions is a reversal of the  $|\Delta\eta_{12}|$  requirement, and removal of the  $M_J^{\Sigma}$  requirement. Requiring large  $|\Delta\eta|_{12}$  reduces the signal contribution to these high-multiplicity regions. There is no  $M_J^{\Sigma}$  requirement for the validation region, so that the performance of the template method over the full range of  $M_J^{\Sigma}$  can be evaluated.

There are two uncertainty determination regions (UDRs), which are used to derive the data-driven background systematic uncertainty. The high- $p_T$  UDR, referred to as as UDR1, consists of events with exactly two large-R jets, with at least one having  $p_T > 400 \text{ GeV}$ . The low- $p_T$  UDR, referred to as UDR2, consists of events with exactly four large-R jets, all of which have  $p_T < 400 \text{ GeV}$ . The UDRs are independent of the control, validation, and signal regions.
		$N_{jet} \ (p_T > 200 \ GeV)$	b-tag	<i>b</i> -match	$p_{T,1}$	$ \Delta \eta_{12} $	$M_J^{\Sigma}$
CD	3jCRb	= 3	-	Yes	-	-	-
Un	3jCR	=3	-	No	-	-	-
UDR	UDR1	= 2	-	-	$> 400 \ GeV$	-	-
	UDR2	=4	-	-	< 400~GeV	-	-
	4jVRb	$\geq 4$	Yes	-	> 400 ~GeV	> 1.4	-
$\mathbf{V}\mathbf{D}$	5jVRb	$\geq 5$	Yes	-	-	> 1.4	-
٧K	4jVR	$\geq 4$	-	-	> 400 ~GeV	> 1.4	-
	$5 \mathrm{jVR}$	$\geq 5$	-	-	-	> 1.4	-
	4jSRb	$\geq 4$	Yes	-	$> 400 \ GeV$	< 1.4	$> 1.0 \ TeV$
	5jSRb	$\geq 5$	Yes	-	-	< 1.4	$> 0.8 \ TeV$
SR		$\geq 5$	Yes	-	-	< 1.4	$> 0.6 \ TeV$
	4jSR	$\geq 4$	-	-	> 400 ~GeV	< 1.4	$> 1.0 \ TeV$
	5jSR	$\geq 5$	-	-	-	< 1.4	$> 0.8 \ TeV$

Table 8.1: Summary of the requirements defining the control, uncertainty determination, validation, and signal regions. Requirements are placed on the large-R jet multiplicity  $(N_{jet})$ , the presence or absence of a b-tagged small-R jet (*b*-tag), the  $p_T$  of the leading jet  $(p_{T,1})$ , the pseudorapidity difference between the two leading jets  $(|\Delta \eta_{12}|)$ , and the scalar sum of the first four leading jets in the event  $(M_J^{\Sigma})$  [69].

## Chapter 9

## **Background estimation method**

### 9.1 Jet mass templates

#### 9.1.1 Template creation

Jet mass templates are derived from a signal-depleted control region consisting of events with exactly three large-R jets. For each  $p_T$ ,  $|\eta|$ , and b-match bin, the distribution of individual jet masses in that bin is taken as the template. The templates combine to form the binned conditional probability distribution:  $p(m|p_T, |\eta|, b - match)$ .

Separate templates are created for b-matched and non-b-matched jets. For the b-matched templates, only events with  $|\Delta\eta_{1,2}| > 1.4$  are included in the templates. Templates derived from b-matched jets are used to dress b-matched jets in the kinematic sample, and templates derived from non-b-matched jets are used to dress non-b-matched jets in the kinematic sample. Each template is a one-dimensional histogram of log  $(m/p_T)$ , with 50 bins. Jets with log  $(m/p_T) < -7$  are excluded from the templates. Example templates are shown in figure 9.1 for two representative  $p_T$ - $|\eta|$  bins. A clear difference can be seen between the templates derived from b-matched jets, for the same  $p_T$ - $|\eta|$  bin. This feature can be seen in both data and simulation. Additionally, agreement between data and simulation is observed in the general template shapes.

Templates are binned in  $p_T$  and  $|\eta|$ . The  $p_T$  bins are approximately logarithmic, while the  $|\eta|$  bin boundaries are at 0.0, 0.5, 1.0, and 1.5. The template binning and number of jets contributing to each bin are shown in figure 9.2.

#### 9.1.2 Template validation

Dressed mass response plots are created by plotting the average dressed and average kinematic jet mass in each  $p_T$  bin. The dressed mass response for the control region is shown in figure 9.3. Good agreement between average dressed and kinematic masses is observed



Figure 9.1: Two representative template distributions used in the analysis, showing a comparison between data in black and simulation in red. The solid markers show the templates derived from b-matched jets, while the empty markers show the templates derived from nonb-matched jets. In (a), template jets are required to have 600  $GeV < p_T < 644 \ GeV$  and  $0.5 < |\eta| < 1.0$ . In (b), templates jets are required to have 733  $GeV < p_T < 811 \ GeV$  and  $1.5 < |\eta| < 2.0$  [69].



Figure 9.2: Number of jets contributing to each template bin for the non-b-matched (left) and b-matched (right) templates.

in this region, because the dressing procedure is applied to the same jets from which the templates are derived.

Dressed mass response plots in other regions are used to evaluate how well the mass templates generalize to events with different jet multiplicities. In the absence of signal events, a disagreement between the average dressed and kinematic masses would indicate that an individual jet mass is dependent on the number of jets in that jet's event, violating the assumptions of the template method. A data-driven method is used to estimate the extent to which this assumption is violated, and the size of the effect this can have on the background estimation uncertainty. This method is described in 9.1.4.



Figure 9.3: Average dressed and kinematic jet masses for each  $p_T$  bin in the control region

#### 9.1.3 Background estimation using jet mass templates

For each jet in the kinematic region, a dressed mass is generated by sampling from the template corresponding to its  $p_T$ ,  $|\eta|$  and b-match bin. To generate a dressed mass, the empirical cumulative distribution function (ECDF) is calculated for the template. A uniform random number, y, in the range [0, 1) is then generated. The inverse of the ECDF,  $\Phi^{-1}(y)$ , gives a randomized log  $(m/p_T)$  bin. A second uniform random number, x, is sampled from the range  $[x_1, x_2)$ , where  $x_1, x_2$  are the edges of the selected bin. The dressed mass is then computed as  $m_{dressed} = p_T e^x$ . To obtain a dressed  $M_J^{\Sigma}$  for an event, one dressed mass is generated for each jet, and the dressed masses are summed. For events with more than four jets, only the first four leading jets are included in the sum.

To obtain the nominal dressed  $M_J^{\Sigma}$  distribution,  $n_{toys}$  histograms of  $M_J^{\Sigma}$  are created, where each histogram is generated by dressing all events in the sample once. For each  $M_J^{\Sigma}$ bin, the average bin content over all histograms is taken as the nominal value, and the standard deviation of bin contents is taken as one contribution to the statistical uncertainty. The  $M_J^{\Sigma}$  histograms are binned as follows. There are ten equal-width bins covering the range  $0 \ TeV \leq M_J^{\Sigma} < 0.5 \ TeV$ . The next three bins cover the ranges  $0.5 \ TeV \leq M_J^{\Sigma} < 0.6 \ TeV$ ,  $0.6 \ TeV \leq M_J^{\Sigma} < 0.8 \ TeV$ , and  $0.8 \ TeV \leq M_J^{\Sigma} < 1.0 \ TeV$ . The final bin is  $M_J^{\Sigma} \geq 1.0 \ TeV$ . The dressed  $M_J^{\Sigma}$  distributions are scaled such that the dressed yield in the range  $0.2 \ TeV < M_J^{\Sigma} < 0.4 \ TeV$  is equal to the kinematic yield in the same range. Separate scale factors are derived for each of the validation and signal regions.

To determine the nominal predicted background yield, one thousand toys are generated, where a toy consists of a dressed  $M_J^{\Sigma}$  value for each event in the kinematic sample. For each toy, the number of events with dressed  $M_J^{\Sigma}$  greater than the signal region  $M_J^{\Sigma}$  cut are counted, giving a distribution of one thousand dressed background yields. The central value of this distribution is multiplied by the scale factor to obtain the nominal background prediction. The standard deviation of this distribution is multiplied by the scale factor to obtain the statistical uncertainty on the background prediction.

Systematically-shifted background yield predictions are determined by repeating the above procedure for the systematically-shifted dressed  $M_J^{\Sigma}$  values. The systematic uncertainties are taken as the difference between the nominal and systematically-shifted background yield predictions. Scale factors are only derived from the nominal  $M_J^{\Sigma}$  distributions and applied to both the nominal and systematically-shifted predictions. The two systematic uncertainties are symmetrized by taking the maximum of the downward-shifted and upward-shifted uncertainties.

#### 9.1.4 Systematic uncertainty

The template method relies on the assumption that the probability density function (PDF) for a jet in a background event to have mass m depends only on the jets  $p_T$ ,  $\eta$ , and b-match status of the individual jet. It also assumes that the mass PDF of a jet is independent of the masses of other jets in the event. Because these assumptions are known to be only approximately true, the method will have some inherent bias, which should be accounted for as a systematic uncertainty. To understand this uncertainty, signal-depleted regions called UDRs are defined, which are independent from signal and validation regions. The predicted jet mass in these regions can be compared to the measured jet mass to determine the level of systematic uncertainty inherent to the method.

As can be seen in figure 9.4, the dressing procedure tends to under-predict jet masses in UDR1 and over-predict jet masses in UDR2. The degree of discrepancy depends strongly on  $p_T$  and the choice of UDR. Separate uncorrelated systematic uncertainties are derived for jets with  $p_T < 400 \ GeV$  and those with  $p_T > 400 \ GeV$ . Since the discrepancy is larger in UDR1 than UDR2 for jets with  $p_T < 400 \ GeV$ , uDR1 is used to derive the uncertainty for those jets. For jets with  $p_T < 400 \ GeV$ , uncertainties are correlated across  $p_T$  and  $|\eta|$  bins, and likewise for jets with  $p_T > 400 \ GeV$ .

Systematic uncertainties are binned in  $p_T$  and  $|\eta|$ . The lowest  $p_T$  bin is for jets with  $p_T < 400 \ GeV$ . The second bin is for jets with  $400 \ GeV \le p_T < 544 \ GeV$ , and the highest bin is for jets with  $p_T \ge 544 \ GeV$ . For jets with  $p_T > 400 \ GeV$ , uncertainties are derived



Figure 9.4: Jet mass response plots showing the discrepancy between average dressed and kinematic jet masses in the two uncertainty determination regions, binned by  $p_T$  and  $|\eta|$ . Since the discrepancy is always larger in UDR1, only UDR1 is used to derive uncertainties [69].

only from UDR1. For jets with  $p_T < 400 \ GeV$ , uncertainties are derived from both UDR1 and UDR2, and the maximum uncertainty is used.

For each  $p_T$  bin in the UDR dressed mass response, a fractional error is calculated as  $e_i = (\langle m_{kin} \rangle - \langle m_{dressed} \rangle) / \langle m_{dressed} \rangle$ . For the lowest and highest  $p_T$  systematic

bins, the root-mean-square of fractional errors is taken as the systematic error. For the intermediate systematic bin, the maximum fractional error is taken.

Two separate, uncorrelated systematic uncertainties are derived. The first uncertainty accounts for the discrepancy between dressed and kinematic masses for jets with  $p_T \geq 400 \ GeV$ . The second accounts for the discrepancy for jets with  $p_T < 400 \ GeV$ . To propagate the low- $p_T$  systematic, two shifted  $M_J^{\Sigma}$  values are calculated for each dressed  $M_J^{\Sigma}$ . The first shifted value is obtained by increasing the dressed mass of every low- $p_T$  jet by its corresponding fractional uncertainty. This yields  $n_{toys}$  histograms of shifted  $M_J^{\Sigma}$ . The average value of each bin content over all toys is taken to obtain the systematically-shifted  $M_J^{\Sigma}$  distribution. The second shifted distribution is obtained by *decreasing* the dressed mass of every low- $p_T$  jet by its corresponding fractional uncertainty, and averaging over all the toys to obtain a downwards-shifted distribution of  $M_J^{\Sigma}$ . The same procedure is used to propagate the high- $p_T$  systematic, but the high- $p_T$  jets are shifted instead of the low- $p_T$  jets.

# 9.2 Effect of signal contamination on background estimation

The presence of signal events in the kinematic sample can affect both the nominal background prediction as well as the systematic uncertainty. This effect can be quantified using a signal injection test. The background prediction is first calculated from a data sample only, and then calculated from a data sample injected with simulated signal events.

One example of this effect can be seen in figure 9.5. For this example, only the first 14.8  $fb^{-1}$  of data are used. The injected signal simulates the cascade decay mode of gluinos with mass  $m_{\tilde{g}} = 1.8 \ TeV$  and neutralinos with mass  $m_{\tilde{\chi}} = 50 \ GeV$ . For this particular choice of masses and decay mode, signal contamination increases the background prediction by approximately 10%, from 18.3 events to 20.2 events. The systematic uncertainty is increased from 8.9 events to 13.3 events due to the signal contamination.

The effect of signal contamination on the background prediction has to be measured separately for each pair of  $m_{\tilde{g}}$ ,  $m_{\tilde{\chi}}$  values. For each gluino and neutralino mass point, the background prediction is generated using only simulated signal events in the kinematic sample. That is, the templates are still derived from data, but the dressing is only performed on simulated signal. The signal events don't need to be included in the templates because the template control region choice reduces any signal contribution in that region to a negligible amount. The prediction from the signal-only kinematic sample is then compared to the data-only prediction. Figure 9.6 shows the ratio of background prediction from signal-only events over the background prediction from data alone. During hypothesis testing, these ratios will be used to subtract off the signal contamination contribution to the background prediction at each signal point separately. The signal region used for this test is the 5-jet, b-tag region with  $M_J^{\Sigma} > 0.8 \ TeV$  (5jSRb1).



Figure 9.5: Signal injection test. The background prediction is first run only on a data sample (a), and then on a data sample injected with simulated signal events (b). The increase in background prediction and systematic uncertainty can be seen.

### 9.3 Effect of $t\bar{t}$ on background estimation uncertainty

Monte Carlo was used to determine if the presence of  $t\bar{t}$  events in the uncertainty determination regions would affect the result of the data-driven background estimation. Figure 9.7 shows the jet mass response in UDR1 for Pythia8 multijet-only Monte Carlo events and for multi-jet plus  $t\bar{t}$ . Figure 9.7 shows the same comparison for UDR2. In both cases, the presence of  $t\bar{t}$  events has a negligible impact on the resulting uncertainty values.

### 9.4 Background estimation method applied to multijet Monte Carlo

The background estimation method was applied to simulated background events from Pythia8 multijets and  $t\bar{t}$ . Other multijet Monte Carlo generators were found to disagree significantly with data in the CR and UDRs, and so are not presented here. The predicted and observed  $M_J^{\Sigma}$  distributions were compared for the various signal regions, and agreement within uncertainties is observed. The number of simulated events available resulted in larger statistical uncertainty than is present in data. The equivalent luminosity available for this study was 1.9  $fb^{-1}$ .

Figure 9.9 shows the predicted and observed  $M_J^{\Sigma}$  distributions for Pythia8 multijets and  $t\bar{t}$  in the  $\geq 4$ -jet b-inclusive region with  $|\Delta \eta_{1,2}| < 1.4$ . Figure 9.10 shows the same comparison for events with at least one b-tagged jet. In both cases, there is agreement between the two



Figure 9.6: The effect of signal contamination on the nominal background prediction. At each neutralino and gluino mass point, simulated signal events are used in the kinematic sample to generate a background prediction. These predictions are compared to the prediction from data only. In both cases, the templates are drawn only from data.

distributions within systematic uncertainties. Figures 9.11 and 9.12 show similar comparisons for the  $\geq$  5-jet regions. In this case the number of simulated events is very small, and so the statistical uncertainty begins to dominate over the systematic uncertainty. Regardless, the observed  $M_J^{\Sigma}$  distributions agree with the dressed distributions, within the uncertainties.

This should not be seen as a true test of the background estimation method. If significant differences between the predicted and observed  $M_J^{\Sigma}$  distributions *had* been found, it would not necessarily indicate a flaw in the method. As discussed in 6, Monte Carlo generators are not expected to perform well in these kinematic regions, hence the need for a data-driven background estimation method in the first place.



(c) b-matched jets, UDR1, multijet plus  $t\bar{t}$  (d): b-matched jets, UDR1, multijet-only

Figure 9.7: Jet mass response in UDR1 for Pythia8 multijet and  $t\bar{t}$  (left column) and Pythia8 multijet only (right column). The top two plots show the response for non-b-matched jets only, and the bottom two plots show the response for b-matched jets only.



Figure 9.8: Jet mass response in UDR2 for Pythia8 multijet and  $t\bar{t}$  (left column) and Pythia8 multijet only (right column). The top two plots show the response for non-b-matched jets only, and the bottom two plots show the response for b-matched jets only.



Figure 9.9: Predicted and observed  $M_J^{\Sigma}$  distributions for Pythia8 multijet and  $t\bar{t}$  Monte Carlo events in the 4jSR region. The red histogram shows the predicted distribution including statistical uncertainty. The green and blue histograms show the systematic uncertainty. The left plot includes only the high- $p_T$  systematic uncertainty, and the right plot shows only the low- $p_T$  systematic uncertainty.



Figure 9.10: Predicted and observed  $M_J^{\Sigma}$  distributions for Pythia8 multijet and  $t\bar{t}$  Monte Carlo events in the 4jSRb region. The red histogram shows the predicted distribution including statistical uncertainty. The green and blue histograms show the systematic uncertainty. The left plot includes only the high- $p_T$  systematic uncertainty, and the right plot shows only the low- $p_T$  systematic uncertainty.



Figure 9.11: Predicted and observed  $M_J^{\Sigma}$  distributions for Pythia8 multijet and  $t\bar{t}$  Monte Carlo events in the 5jSR region. The red histogram shows the predicted distribution including statistical uncertainty. The green and blue histograms show the systematic uncertainty. The left plot includes only the high- $p_T$  systematic uncertainty, and the right plot shows only the low- $p_T$  systematic uncertainty.





(b) Prediction showing low  $p_{\rm T}$  variation

Figure 9.12: Predicted and observed  $M_J^{\Sigma}$  distributions for Pythia8 multijet and  $t\bar{t}$  Monte Carlo events in the 5jSRb region. The red histogram shows the predicted distribution including statistical uncertainty. The green and blue histograms show the systematic uncertainty. The left plot includes only the high- $p_T$  systematic uncertainty, and the right plot shows only the low- $p_T$  systematic uncertainty.

# Chapter 10 Signal systematics

There are several factors that contribute to the systematic uncertainty on the signal yields, which must be accounted for when checking for significant excesses in the signal region or when setting limits on signal production. The dominant sources of systematic error are the large-R jet mass scale (JMS), the small-R b-tagging uncertainty, the Monte Carlo statistical uncertainty, and the modelling uncertainty, which includes uncertainty on parton distribution functions (PDFs), QCD scale uncertainty, and initial state radiation (ISR) modelling uncertainty. The size of the contribution from each source of uncertainty depends on the signal point being evaluated. To evaluate the size of a systematic uncertainty from a given source, a nuisance parameter is varied up and down, and the variation in the signal yield as compared to the nominal value is taken as the size of the uncertainy. When a systematic contains multiple components, those components are treated as uncorrelated, and their contributions are combined in quadrature.

For example, to evaluate the contribution to the signal efficiency uncertainty from PDF, QCD scale,  $\alpha_s$  and ISR modelling uncertainties, truth-level signal simulation samples are created where different parameters are varied in the generation. For PDF uncertainties, the internal event weights in the PDF set are varied up and down during the generation. For the QCD scale and  $\alpha_s$  uncertainties, the value of these parameters are varied up and down during the generation. For ISR uncertainties, the value of the matching scale,  $q_{cut}$  is varied up and down during the generation. For ISR uncertainties, the value of the matching scale,  $q_{cut}$  is varied up and down during the generation. Figures ?? show the nominal and systematically-shifted distributions of four observables that contribute to the signal region definition for four representative signal points. The nuisance parameters varied in these plots are the event internal weights of the PDF set, the QCD scale, and  $\alpha_s$ . The PDF and QCD scale contributions to the uncertainty are highest at low gluino mass, reaching a maximum of 25% at  $m_{\tilde{g}} = 1.0 \ TeV$ , the lowest gluino mass studied. For higher masses, these uncertainties drop to only a few percent. For reconstruction-level uncertainties such as b-tagging efficiency, the nominal signal samples are used, and the detector simulation is run with different nuisance parameters separately shifted up or down.

For the large-R JMS, there are four components, called the baseline, modeling, statistical, and tracking components. These components are derived from the  $R_{trk}$  method [51]. The



Figure 10.1: Nominal and systematically-shifted  $N_{jet}$  distributions for four signal points, showing the shifted distributions for internal generator event weights, QCD scale, and  $\alpha_s$ .

JMS uncertainty is largest for  $m_{\tilde{g}} = 1.0 \ TeV$ , at  $\approx 24\%$ , and drops to  $\approx 8\%$  for signal points with  $m_{\tilde{g}} = 1.8 \ TeV$ . It is generally dominated by the tracking uncertainty, followed by the baseline uncertainty. The b-tagging uncertainty is evaluated by varying a set of 25 nuisance parameters. The result is an uncertainty on the signal efficiency of between 15% and 25%. This uncertainty is only applied to the b-tag signal regions. The luminosity uncertainty is 3.2%, and other uncertainties such as small-R jet energy scale (JES), small-R jet energy resolution (JER), and pileup are found to be negligible. Table10.1 gives a summary of the sources of systematic uncertainty size for each source. Because the size of the uncertainty depends strongly on the particular signal point, these are only approximate ranges, except in the case of luminosity, where the uncertainty is the same regardless of signal point. The JMS uncertainty is highest for the lowest-mass signal points, and decreases with gluino mass. The *b*-tagging efficiency uncertainty is only relevant for signal regions that include a *b*-tag



Figure 10.2: Nominal and systematically-shifted  $N_{b-jet}$  distributions for four signal points, showing the shifted distributions for internal generator event weights, QCD scale, and  $\alpha_s$ .

requirement.



Figure 10.3: Nominal and systematically-shifted  $|\Delta \eta_{1,2}|$  distributions for four signal points, showing the shifted distributions for internal generator event weights, QCD scale, and  $\alpha_s$ .



Figure 10.4: Nominal and systematically-shifted  $M_J^{\Sigma}$  distributions for four signal points, showing the shifted distributions for internal generator event weights, QCD scale, and  $\alpha_s$ .

source	number of nuisance parameters	size of uncertainty
b-tagging efficiency	25	15% - 25%
Large-R JMS	4	baseline: $4\% - 10\%$ modeling: $3\% - 6\%$ statistical: $5\% - 8\%$ tracking: $9\% - 17\%$
PDF, QCD Scale, and $\alpha_s$	3	5% - 25%
luminosity	1	3.2%

Table 10.1: Summary of contributions to systematic uncertainties on the signal yield from various sources, including the number of nuisance parameters and approximate range of uncertainty size for each.

## Chapter 11

### Results

### 11.1 Predicted and observed yields

Figure 11.1 shows the predicted and observed  $M_J^{\Sigma}$  distributions in the four-jet and five-jet regions, along with the total estimated uncertainty. The red solid line indicates the estimated yield in each  $M_J^{\Sigma}$  bin, and the red shaded area shows the total uncertainty, including statistical and systematic uncertainties. The gray dashed lines are the  $\pm 1\sigma$  bands from the statistical uncertainty only. There is good agreement between predicted and observed event yields across a range of  $\pm 1\sigma$  values, within the total uncertainty. The exception is in the five-jet signal region, where a slight excess of events is observed. After statistical analysis, discussed in 11.2, the excess is found to not be significant. The  $M_J^{\Sigma}$  distributions for two different cascade decay signal mass points are also shown on the plots, to visualize the effect that the existence of signal events would hve on the distributions.

In the four-jet signal regions, no excess of events was observed. In the five-jet signal regions, an excess of events were observed, but these excesses were not found to be statistically significant. Predicted and observed yields in all signal regions are shown in table 11.1, including the number of events in the corresponding normalization region,  $N_{NR}$ , and the statistical and systematic uncertainties on the background yield predictions in those regions. Table 11.1 also shows the number of events in the normalization region,  $N_{NR}$ , corresponding to each signal region. The central value of the prediction for each signal region is shown, as well as the statistical uncertainty, and the systematic uncertainties from the low- $p_T$  and high- $p_T$  uncertainty determination regions.

### **11.2** Statistical Interpretation

The data are interpreted in a frequentist framework by defining a likelihood-ratio test statistic, and calculating a p-value under the null hypothesis that no signal is present. If the hypothesis cannot be rejected, then there is no evidence for the signal process, and limits can be set on the signal production cross section. The limit-setting procedure uses the  $CL_s$ 



Figure 11.1: Distribution of  $M_J^{\Sigma}$  in the four-jet inclusive (4jSR), four-jet b-tag (4jSRb), five-jet inclusive (5jSR), and five-jet b-tag (5jSRb) regions. The solid red lines show the predicted  $M_J^{\Sigma}$  distributions, along with the total uncertainty shaded in red. The black dots show the observed  $M_J^{\Sigma}$  distributions. The corresponding  $M_J^{\Sigma}$  cuts are indicated with the solid black vertical line and arrow for each region. Two separate signal regions are defined for the 5jSRb region, one with an  $M_J^{\Sigma}$  cut of 0.6 TeV, and one with a cut of 0.8 TeV. The green and blue dashed lines show the predicted  $M_J^{\Sigma}$  contribution from cascade-decay signal events from two different mass points.

method [84, 85]. The likelihood function for an observed yield k and expected yield  $\lambda$  is:

$$L(\mu) = P(k|\lambda) \prod_{l} G(0|\theta_l, 1)$$
(11.1)

Region	$N_{NR}$	$\geq M_J^{\Sigma} \ [TeV]$	Expected (	±	(stat.)	±	$(high-p_T)$	±	$(\text{low-}p_T))$	Observed
4jSRb	64081	1.0	23.6	$\pm$	4.6	$\pm$	6.1	$\pm$	1.7	15
4jSR	224862	1.0	82.0	$\pm$	7.6	$\pm$	15.8	$\pm$	4.4	82
$5jSRb_1$	2177	0.8	7.0	$\pm$	2.4	$\pm$	1.9	$\pm$	0.7	10
$5jSRb_2$	2177	0.6	44.0	$\pm$	7.5	$\pm$	11.2	$\pm$	7.2	61
5jSR	6592	0.8	18.0	$\pm$	3.7	$\pm$	4.6	$\pm$	1.5	31

Table 11.1: Predicted and observed yields in all signal regions used in the analysis. The number of events in the corresponding normalization regions,  $N_{NR}$  is shown. For each signal region, the minimum value of  $M_J^{\Sigma}$  used to define that region is also shown. Additionally, the statistical uncertainty on the background yield, as well as the two systematic uncertainties, derived from the high- $p_T$  and low- $p_T$  UDRs are shown [69].

where  $\mu$  is a signal-strength parameter,  $P(k|\lambda)$  is the Poisson distribution with mean  $\lambda$ , and  $G(0|\theta_l, 1)$  is a log-normal constraint term for systematic l. Uncertainties are incorporated into the likelihood function as nuisance parameters, by parameterizing the expected yield as:

$$\lambda = \mu \prod_{i} (1 + \theta_i \sigma_i) S_0 + B_0 \prod_{j} (1 + \sigma_{b,j} \sigma_{b,j}) - \mu \Delta b (1 + \theta_{b,stat} \sigma_{b,stat})$$
(11.2)

where  $\theta_i$  is the nuisance parameter for signal systematic *i* with size  $\sigma_i$ , and  $\theta_{b,j}$  is the nuisance parameter for background systematic *j* with size  $\sigma_{b,j}$ . Recall that each nuisance parameter is constrained by a log-normal constraint in the likelihood function.  $S_0$  is the nominal signal yield, and  $B_0$  is the predicted background yield. As described in 9.2, the presence of signal contamination in the control region can increase the expected background prediction. To correct for this, the signal contamination contribution to the background prediction must be subtracted off from the total expected yield  $\lambda$ . This contribution is proportional to the signal strength, and is subject to statistical, but not systematic uncertainties. So the term used to correct for signal contamination is  $-\mu\Delta b(1 + \theta_{b,stat}\sigma_{b,stat})$ .

The test statistic  $q_{\mu}$  is a ratio of two different likelihood function fits:

$$q_{\mu} = -2\ln\frac{L(\mu,\hat{\theta})}{L(\hat{\mu},\hat{\hat{\theta}})} \tag{11.3}$$

where  $\hat{\theta}$  are the best-fit values of the parameters when the likelihood function is fit to data with the signal-strength parameter  $\mu$  kept fixed to the value of the null hypothesis, and  $\hat{\hat{\theta}}$ are the best-fit values of the parameters when  $\mu$  is allowed to vary in the fit, along with the other parameters.

To test for evidence of the signal process, a hypothesis test is conducted with the null hypothesis that  $\mu = 0$ , meaning that the signal process does not exist. The frequentist interpretation of a *p*-value is the probability of observing an excess at least as large as what

was seen in the data, assuming that the signal process does not exist. A *p*-value of  $3 \times 10^{-7}$  or less is required to reject the null hypothesis. The *p*-value threshold is equivalent to the false-positive rate, so the probability of incorrectly rejecting the null hypothesis due to a statistical fluctuation will be  $3 \times 10^{-7}$  for this test. A separate *p*-value is calculated for each of the signal regions. The *p*-values for each of the signal regions can be seen in table 11.2. None of the *p*-values are below the threshold, so no evidence of the signal process was found.

Signal Region	$M_J^{\Sigma}$ requirement	Expected limit $[fb]$	Observed limit $[fb]$	$p_0$ -value
4jSRb	$> 1.0 \ TeV$	$0.53_{-0.12}^{+0.20}$	0.37	0.5
4jSR	$> 1.0 \ TeV$	$1.12_{-0.32}^{+0.50}$	1.50	0.24
$5jSRb_1$	$> 0.8 \ TeV$	$0.24_{-0.06}^{+0.10}$	0.34	0.26
$5jSRb_2$	$> 0.6 \ TeV$	$0.86^{+0.40}_{-0.20}$	1.32	0.20
5jSR	$> 0.8 \ TeV$	$0.44_{-0.10}^{+0.18}$	0.84	0.062

Table 11.2: Observed and expected limits on gluino pair-production cross section for each of the signal regions, along with the p-value for any excess observed in each region [69].

Using the observed yields, upper bounds can be set on the cross-section of the two signal models, as well as model-independent upper bounds. The same test statistic can be used in the limit-setting procedure as was used in the discovery test. In this case, the signal strength parameter  $\mu$  will be fixed to 1 in the numerator of the likelihood-ratio. This corresponds to a null hypothesis of a signal process existing with cross-section equal to the theoretical prediction. For a given signal point, if the *p*-value falls below 0.05, the signal plus background hypothesis will be rejected, and that point will be considered excluded. Since signal yields are only determined at discrete mass points, the limits are smoothly interpolated between points.

The resulting limits for the cascade decay scenario are shown in figure 11.2. The observed limit and its theoretical uncertainty are shown in red, and the expected limit is shown in black with 1 sigma band shown in yellow. Mass points to the left of the solid red curve are excluded. For comparison, the limits set by the Run-1 analysis are shown in gray. The result is a significant increase in the excluded mass space for this model. However, due to the slight excess in the signal region, the observed limit is less than expected [69]. The lower bound on gluino mass varies from 1 TeV to 1.875 TeV depending on the value of  $m_{\tilde{\chi}_1^0}$ .

For the direct decay scenario, a similar limit-setting method was used, with *p*-values calculated over a range of gluino masses and also over a range of  $\mu$  values. For a given gluino mass and value of  $\mu$ , if the *p*-value is less than 0.05, the gluinos with that mass and production cross-section are considered excluded. This allows for lower bounds to be placed on the gluino cross-section over a range of gluino mass values.

The resulting limits for the direct-decay model are shown in figure 11.3. The theoretical  $\tilde{g}\tilde{g}$  production cross-section as a function of  $m_{\tilde{g}}$  is shown as a red dashed line, with gray band indicating the uncertainty. The expected limit is shown as a black dashed line with green



Figure 11.2: Predicted and observed limits for the cascade decay model over a range of  $m_{\tilde{g}}$  and  $m_{\tilde{\chi}_1^0}$  values. Both predicted and observed limits are shown with  $1\sigma$  uncertainty bands. Mass points to the left of the red solid curve are excluded. The gray dashed curve shows the limits from the Run-1 analysis. Due to slight excess in the signal region, the observed limit is less than expected [69].

and yellow bands indicating the one- and two-sigma uncertainty extent, and the observed limit is a solid black line. The limits are upper bounds on the production cross-section for each mass point. Because of the small excess of events observed in the signal region, the upper bound on the production cross section is higher than the theoretical production cross section in the entire gluino mass range that was tested, from 900 GeV to 1.8 TeV. So no new limits can be set on the mass of the gluino decaying through this channel.



Figure 11.3: Predicted and observed cross-section limits for the direct decay model over a range of  $m_{\tilde{g}}$  values. The black solid curve indicates the upper bound on the signal production cross section for each gluino mass point. Also shown are the estimated gluino pair production cross-sections at each mass point [69]. Since the theoretical cross-section is smaller than the upper bound for every mass point studied, this signal cannot be excluded for any mass point.

# Chapter 12

# Conclusion

A search was performed for supersymmetric particles called gluinos decaying through Rparity-violating vertices to high sum-of-jet-mass final states. The search was based on  $36.1 \ fb^{-1}$  of  $\sqrt{s} = 13 \ TeVe$  proton-proton collision data taken by the ATLAS detector at the LHC in 2015 and 2016. There were two signal decay processes under consideration, one in which the gluinos each decay to three Standard Model quarks, and one in which the gluino decays to a neutralino and two quarks, and the neutralino decays to three quarks. Either of those processes would result to an excess in the number of events with a large sum of leading jet masses over the Standard Model background.

A data-driven jet mass template method was used to estimate the contribution from the Standard Model backgrounds. New features of this analysis are using a data-driven estimate of background statistical uncertainty derived from the jet mass templates, and using b-tagging to boost the expected signal-to-noise ratio. No significant excess above the background prediction was found in any of the signal regions. Based on the observed yields, limits were placed on the production of gluinos decaying through either the direct decay or cascade decay scenarios.

For the direct decay mode, 95% CL upper bounds on the gluino production cross section range from 0.8 fb for a gluino with mass of 900 GeV, to 0.011 fb for a gluino with mass 1.8 TeV. These upper bounds are above the theoretical cross-section for gluinos in this mass range, so the search failed to exclude gluinos decaying via this process. For the cascade decay mode, a range of gluino and neutralino mass points could be excluded based on the cross-section upper bounds. Depending on the neutralino mass, gluinos with masses between 1 TeV and 1.8 TeV can be excluded, with the maximum gluino mass excluded for neutralino mass around 1 TeV. This search has provided the strongest limits yet for the R-parityviolating UDD decays of gluinos, and significantly extends the reach from the Run-1 analysis.

# Bibliography

- D. Hanneke, S. Fogwell, and G. Gabrielse. "New Measurement of the Electron Magnetic Moment and the Fine Structure Constant". In: *Phys. Rev. Lett.* 100 (12 Mar. 2008), p. 120801. DOI: 10.1103/PhysRevLett.100.120801. URL: https://link.aps.org/ doi/10.1103/PhysRevLett.100.120801.
- Standard Model Summary Plots Spring 2019. Tech. rep. ATL-PHYS-PUB-2019-010. Geneva: CERN, Mar. 2019. URL: http://cds.cern.ch/record/2668559.
- [3] G. Aad et al. "Observation of a new particle in the search for the Standard Model Higgs boson with the ATLAS detector at the LHC". In: *Physics Letters B* 716.1 (2012), pp. 1–29. ISSN: 0370-2693. DOI: https://doi.org/10.1016/j.physletb.2012.08.020. URL: http://www.sciencedirect.com/science/article/pii/S037026931200857X.
- [4] S. Chatrchyan et al. "Observation of a new boson at a mass of 125 GeV with the CMS experiment at the LHC". In: *Physics Letters B* 716.1 (2012), pp. 30-61. ISSN: 0370-2693. DOI: https://doi.org/10.1016/j.physletb.2012.08.021. URL: http://www.sciencedirect.com/science/article/pii/S0370269312008581.
- K.A. Olive and Particle Data Group. "Review of Particle Physics". In: Chinese Physics C 38.9 (2014), p. 090001. URL: http://stacks.iop.org/1674-1137/38/i=9/a=090001.
- [6] Guido Altarelli and G. Parisi. "Asymptotic Freedom in Parton Language". In: Nucl. Phys. B126 (1977), pp. 298–318. DOI: 10.1016/0550-3213(77)90384-4.
- [7] Richard D. Ball et al. "Parton distributions for the LHC Run II". In: JHEP 04 (2015),
   p. 040. DOI: 10.1007/JHEP04(2015)040. arXiv: 1410.8849 [hep-ph].
- [8] A. D. Martin, W. J. Stirling, and G. Thorne R. S.and Watt. "Parton distributions for the LHC". In: *The European Physical Journal C* 63.2 (Sept. 2009), pp. 189–285.
- R. Barbieri and G.F. Giudice. "Upper bounds on supersymmetric particle masses". In: Nuclear Physics B 306.1 (1988), pp. 63-76. ISSN: 0550-3213. DOI: https://doi.org/ 10.1016/0550-3213(88)90171-X. URL: http://www.sciencedirect.com/science/ article/pii/055032138890171X.
- [10] A. J. Buras et al. "Aspects of the Grand Unification of Strong, Weak and Electromagnetic Interactions". In: Nucl. Phys. B135 (1978), pp. 66–92. DOI: 10.1016/0550– 3213(78)90214-6.

- [11] M. Tanabashi et al. "Review of Particle Physics". In: *Phys. Rev.* D98.3 (2018), p. 030001.
   DOI: 10.1103/PhysRevD.98.030001.
- [12] Edward W. Kolb and Michael S. Turner. "The Early Universe". In: Front. Phys. 69 (1990), pp. 1–547.
- T. D. Brandt. "Constraints on MACHO Dark Matter from Compact Stellar Systems in Ultra-faint Dwarf Galaxies". In: *The Astrophysical Journal* 824, L31 (June 2016), p. L31. DOI: 10.3847/2041-8205/824/2/L31. arXiv: 1605.03665.
- [14] Leonard Susskind. "Dynamics of spontaneous symmetry breaking in the Weinberg-Salam theory". In: *Phys. Rev. D* 20 (10 Nov. 1979), pp. 2619–2625. DOI: 10.1103/ PhysRevD.20.2619. URL: https://link.aps.org/doi/10.1103/PhysRevD.20.2619.
- [15] S. P. Martin. "A Supersymmetry Primer". In: Perspectives on Supersymmetry. Ed. by G. L. Kane. World Scientific Publishing Co, 1998, pp. 1–98. DOI: 10.1142/ 9789812839657\_0001.
- [16] H. Murayama. "Supersymmetry Phenomenology". In: Particle Physics. Ed. by G. Senjanovi ć and A. Y. Smirnov. 2000, p. 296. eprint: hep-ph/0002232.
- [17] ATLAS Collaboration. "Supersymmetry Public Results". 2019. URL: https://twiki. cern.ch/twiki/bin/view/AtlasPublic/SupersymmetryPublicResults.
- [18] R. N. Mohapatra. "Supersymmetric Grand Unification". In: ArXiv High Energy Physics
   Phenomenology e-prints (Jan. 1998). eprint: hep-ph/9801235.
- G. Jungman, M. Kamionkowski, and K. Griest. "Supersymmetric dark matter". In: *Phys.Rept.* 267 (Mar. 1996), pp. 195–373. DOI: 10.1016/0370-1573(95)00058-5. eprint: hep-ph/9506380.
- [20] Savas Dimopoulos and Howard Georgi. "Softly broken supersymmetry and SU(5)". In: Nuclear Physics B 193.1 (1981), pp. 150-162. ISSN: 0550-3213. DOI: https://doi. org/10.1016/0550-3213(81)90522-8. URL: http://www.sciencedirect.com/ science/article/pii/0550321381905228.
- G. 't Hooft. "Symmetry Breaking through Bell-Jackiw Anomalies". In: *Phys. Rev. Lett.* 37 (1 July 1976), pp. 8–11. DOI: 10.1103/PhysRevLett.37.8. URL: https://link.aps.org/doi/10.1103/PhysRevLett.37.8.
- [22] Ian Hinchliffe and Thomas Kaeding. "B- and L-violating couplings in the minimal supersymmetric standard model". In: Phys. Rev. D 47 (1 Jan. 1993), pp. 279-284. DOI: 10.1103/PhysRevD.47.279. URL: https://link.aps.org/doi/10.1103/PhysRevD.47.279.
- [23] ATLAS Collaboration. "The ATLAS Experiment at the CERN Large Hadron Collider". In: JINST 3 (2008), S08003.
- [24] "LHC Guide". Mar. 2017. URL: http://cds.cern.ch/record/2255762.
- [25] ATLAS Collaboration. "Luminosity Public Results". 2018. URL: https://twiki. cern.ch/twiki/bin/view/AtlasPublic/LuminosityPublicResultsRun2.

#### **BIBLIOGRAPHY**

- [26] Apollinari G. et al. High-Luminosity Large Hadron Collider (HL-LHC): Technical Design Report V. 0.1. CERN Yellow Reports: Monographs. Geneva: CERN, 2017. DOI: 10.23731/CYRM-2017-004. URL: http://cds.cern.ch/record/2284929.
- [27] Lyndon Evans and Philip Bryant. "LHC Machine". In: Journal of Instrumentation 3.08 (2008), S08001. URL: http://stacks.iop.org/1748-0221/3/i=08/a=S08001.
- [28] AC Team. "Diagram of an LHC dipole magnet. Schéma d'un aimant dipôle du LHC". June 1999. URL: https://cds.cern.ch/record/40524.
- [29] Helmuth Spieler. Semiconductor Detector Systems. Oxford Science Publications, 2005.
- [30] ATLAS Collaboration. "ATLAS Pixel Detector Electronics and Sensors". In: JINST 3 (2008), P07007.
- [31] ATLAS Collaboration. "ATLAS Insertable B-Layer Technical Design Report". In: *CERN/LHCC 2010-013*. 2010.
- [32] ATLAS Collaboration. A New Sub Detector for ATLAS. 2014. URL: http://atlasexperiment. org/news/2014/a-new-sub-detector-for-ATLAS.html (visited on 06/02/2014).
- [33] ATLAS Collaboration. "The barrel modules of the ATLAS semiconductor tracker". In: Nuclear Instruments and Methods in Physics Research 568 (2 2006).
- [34] P Haefner. "The ATLAS silicon microstrip tracker. Operation and performance". In: JINST 5 (2010), p. C12050.
- [35] Glenn F. Knoll. Radiation Detection and Measurement. John Wiley and Sons, 2000.
- [36] ATLAS Collaboration. "The ATLAS Transition Radiation Tracker (TRT) proportional drift tube: design and performance". In: *JINST* 2 (2013), P02013.
- [37] N. Nikiforou. "Performance of the ATLAS Liquid Argon Calorimeter after three years of LHC operation and plans for a future upgrade". In: *ArXiv e-prints* (June 2013). arXiv: 1306.6756 [physics.ins-det].
- [38] M. Aharrouche et al. "Energy linearity and resolution of the ATLAS electromagnetic barrel calorimeter in an electron test-beam". In: Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment 568.2 (2006), pp. 601–623. ISSN: 0168-9002. DOI: https://doi.org/10. 1016/j.nima.2006.07.053.URL: http://www.sciencedirect.com/science/ article/pii/S0168900206013222.
- [39] S. Palestini. "The muon spectrometer of the ATLAS experiment". In: Nucl. Phys. Proc. Suppl. 125 (2003). [,337(2003)], pp. 337–345. DOI: 10.1016/S0920-5632(03)91013-9.
- [40] Morad Aaboud et al. "Performance of the ATLAS Trigger System in 2015". In: Eur. Phys. J. C 77.CERN-EP-2016-241. 5 (Nov. 2016). 77 pages in total, author list starting page 61, 50 figures, 1 table. Published in Eur. Phys. J. C. All figures including auxiliary figures are available at http://atlas.web.cern.ch/Atlas/GROUPS/PHYSICS/PAPERS/TRIG-2016-01/, 317. 76 p. DOI: 10.1140/epjc/s10052-017-4852-3. URL: https://cds. cern.ch/record/2235584.

- [41] Tanju Gleisberg et al. "SHERPA 1. alpha: A Proof of concept version". In: JHEP 02 (2004), p. 056. DOI: 10.1088/1126-6708/2004/02/056. arXiv: hep-ph/0311263 [hep-ph].
- [42] J Alwall et al. "Comparative study of various algorithms for the merging of parton showers and matrix elements in hadronic collisions". In: *Eur. Phys. J. C* 53.arXiv:0706.2569. CERN-PH-TH-2007-066. LU-TP-07-13. KA-TP-06-2007. DCPT-07-62. IPPP-07-31. 3 (June 2007), 473-500. mult. p. DOI: 10.1140/epjc/s10052-007-0490-5. URL: http://cds.cern.ch/record/1042163.
- [43] Multijet simulation for 13 TeV ATLAS Analyses. Tech. rep. ATL-PHYS-PUB-2019-017. Geneva: CERN, Apr. 2019. URL: http://cds.cern.ch/record/2672252.
- [44] Gavin P. Salam. "Towards jetography". In: The European Physical Journal C 67.3 (June 2010), pp. 637–686. ISSN: 1434-6052. DOI: 10.1140/epjc/s10052-010-1314-6. URL: https://doi.org/10.1140/epjc/s10052-010-1314-6.
- [45] John E. Huth et al. "Toward a standardization of jet definitions". In: 1990 DPF Summer Study on High-energy Physics: Research Directions for the Decade (Snowmass 90) Snowmass, Colorado, June 25-July 13, 1990. 1990, pp. 0134-136. URL: http://lss.fnal.gov/cgi-bin/find\_paper.pl?conf-90-249.
- [46] G. C. Blazey et al. "Run II Jet Physics: Proceedings of the Run II QCD and Weak Boson Physics Workshop". In: ArXiv High Energy Physics - Experiment e-prints (May 2000). eprint: hep-ex/0005012.
- [47] Victor Coco et al. "Jets and jet algorithms". In: Proceedings, HERA and the LHC Workshop Series on the implications of HERA for LHC physics: 2006-2008. 2009, pp. 182–204. DOI: 10.3204/DESY-PR0C-2009-02/54.
- [48] Matteo Cacciari, Gavin P. Salam, and Gregory Soyez. "FastJet user manual". In: *The European Physical Journal C* 72.3 (Mar. 2012), p. 1896. ISSN: 1434-6052. DOI: 10. 1140/epjc/s10052-012-1896-2. URL: https://doi.org/10.1140/epjc/s10052-012-1896-2.
- [49] Ryan Atkin. "Review of jet reconstruction algorithms". In: Journal of Physics: Conference Series 645.1 (2015), p. 012008. URL: http://stacks.iop.org/1742-6596/ 645/i=1/a=012008.
- [50] M. Cacciari, G. P. Salam, and G. Soyez. "The anti- $k_t$  jet clustering algorithm". In: Journal of High Energy Physics 4, 063 (Apr. 2008), p. 063. DOI: 10.1088/1126-6708/2008/04/063. arXiv: 0802.1189 [hep-ph].
- [51] Atlas Collaboration. "Performance of jet substructure techniques for large-R jets in proton-proton collisions at  $\sqrt{s} = 7$  TeV using the ATLAS detector". In: *JHEP* 1309 (2013).
- [52] S. D. Ellis et al. "Jets in hadron-hadron collisions". In: Prog. Part. Nucl. Phys. 60 (2008), pp. 484–551. DOI: 10.1016/j.ppnp.2007.12.002. arXiv: 0712.2447 [hep-ph].

- [53] Jessie Shelton. "TASI Lectures on Jet Substructure". In: (Feb. 2013).
- [54] E N Thompson. "Jet Grooming at ATLAS". In: (Oct. 2012). URL: http://cds.cern. ch/record/1491166.
- [55] A. Abdesselam et al. "Boosted Objects: A Probe of Beyond the Standard Model Physics". In: Eur. Phys. J. C71 (2011), p. 1661. DOI: 10.1140/epjc/s10052-011-1661-y. arXiv: 1012.5412 [hep-ph].
- [56] Georges Aad et al. "Topological cell clustering in the ATLAS calorimeters and its performance in LHC Run 1". In: *Eur. Phys. J.* C77 (2017), p. 490. DOI: 10.1140/ epjc/s10052-017-5004-5. arXiv: 1603.02934 [hep-ex].
- [57] Peter Speckmayer, Tancredi Carli, and Christian Wolfgang Fabjan. "Energy Measurement of Hadrons with the CERN ATLAS Calorimeter". Presented on 18 Jun 2008. 2008. URL: http://cds.cern.ch/record/1112036.
- [58] The ATLAS Collaboration. "Jet energy measurement with the ATLAS detector in proton-proton collisions at  $\sqrt{s} = 7$  TeV". In: *The European Physical Journal C* 73.3 (Mar. 2013), p. 2304. ISSN: 1434-6052. DOI: 10.1140/epjc/s10052-013-2304-2. URL: https://doi.org/10.1140/epjc/s10052-013-2304-2.
- [59] ATLAS Collaboration. "Performance of b -jet identification in the ATLAS experiment". In: Journal of Instrumentation 11.04 (2016), P04008. URL: http://stacks. iop.org/1748-0221/11/i=04/a=P04008.
- [60] Optimisation of the ATLAS b-tagging performance for the 2016 LHC Run. Tech. rep. ATL-PHYS-PUB-2016-012. Geneva: CERN, June 2016. URL: http://cds.cern.ch/ record/2160731.
- [61] Commissioning of the ATLAS high-performance b-tagging algorithms in the 7 TeV collision data. Tech. rep. ATLAS-CONF-2011-102. Geneva: CERN, July 2011. URL: http://cds.cern.ch/record/1369219.
- [62] G Piacquadio and C Weiser. "A new inclusive secondary vertex algorithm for b-jet tagging in ATLAS". In: Journal of Physics: Conference Series 119.3 (2008), p. 032032. URL: http://stacks.iop.org/1742-6596/119/i=3/a=032032.
- [63] Expected performance of the ATLAS b-tagging algorithms in Run-2. Tech. rep. ATL-PHYS-PUB-2015-022. Geneva: CERN, July 2015. URL: https://cds.cern.ch/ record/2037697.
- [64] ATLAS Collaboration. "Search for massive supersymmetric particles decaying to many jets using the ATLAS detector in pp collisions at  $\sqrt{s} = 8 \ TeV$ ". In: *Phys. Rev. D* (2015).
- [65] CMS Collaboration. "Searches for *R*-parity-violating supersymmetry in *pp* collisions at √s = 8 TeV in final states with 0-4 leptons". In: *Phys. Rev. D* 94 (11 Dec. 2016), p. 112009. DOI: 10.1103/PhysRevD.94.112009. URL: https://link.aps.org/doi/10.1103/PhysRevD.94.112009.

#### BIBLIOGRAPHY

- [66] Csaba Csáki, Yuval Grossman, and Ben Heidenreich. "Minimal flavor violation supersymmetry: A natural theory for *R*-parity violation". In: *Phys. Rev. D* 85 (9 May 2012), p. 095009. DOI: 10.1103/PhysRevD.85.095009. URL: https://link.aps.org/doi/ 10.1103/PhysRevD.85.095009.
- [67] A. Hook et al. "High Multiplicity Searches at the LHC Using Jet Masses". In: JHEP 3 (2012), p. 9.
- [68] S. El Hedri et al. "Learning How to Count: A High Multiplicity Search for the LHC". In: JHEP (2013), p. 136.
- [69] "Search for R-parity-violating supersymmetric particles in multi-jet final states produced in p-p collisions at s=13 TeV using the ATLAS detector at the LHC". In: *Physics Letters B* 785 (2018), pp. 136-158. ISSN: 0370-2693. DOI: https://doi.org/ 10.1016/j.physletb.2018.08.021. URL: http://www.sciencedirect.com/ science/article/pii/S0370269318306270.
- [70] J. Alwall et al. "The automated computation of tree-level and next-to-leading order differential cross sections, and their matching to parton shower simulations". In: JHEP 07 (2014), p. 079. DOI: 10.1007/JHEP07(2014)079. arXiv: 1405.0301 [hep-ph].
- [71] Torbjörn Sjöstrand et al. "An Introduction to PYTHIA 8.2". In: Comput. Phys. Commun. 191 (2015), pp. 159–177. DOI: 10.1016/j.cpc.2015.01.024. arXiv: 1410.3012
   [hep-ph].
- Johan Alwall et al. "A Standard format for Les Houches event files". In: Comput. Phys. Commun. 176 (2007), pp. 300-304. DOI: 10.1016/j.cpc.2006.11.010. arXiv: hep-ph/0609017 [hep-ph].
- [73] Andrew Buckley. "ATLAS Pythia 8 tunes to 7 TeV data". In: Proceedings of the Sixth International Workshop on Multiple Partonic Interactions at the Large Hadron Collider. CERN. Geneva: CERN, 2014, p. 29.
- [74] Summary of ATLAS Pythia 8 tunes. Tech. rep. ATL-PHYS-PUB-2012-003. Geneva: CERN, Aug. 2012. URL: https://cds.cern.ch/record/1474107.
- [75] Richard D. Ball et al. "Parton distributions with LHC data". In: Nucl. Phys. B867 (2013), pp. 244–289. DOI: 10.1016/j.nuclphysb.2012.10.003. arXiv: 1207.1303
   [hep-ph].
- [76] David J. Lange. "The EvtGen particle decay simulation package". In: Nuclear Instruments and Methods in Physics Research Section A: Accelerators, Spectrometers, Detectors and Associated Equipment 462.1 (2001). BEAUTY2000, Proceedings of the 7th Int. Conf. on B-Physics at Hadron Machines, pp. 152–155. ISSN: 0168-9002. DOI: https: //doi.org/10.1016/S0168-9002(01)00089-4. URL: http://www.sciencedirect. com/science/article/pii/S0168900201000894.
- [77] Christoph Borschensky et al. "Squark and gluino production cross sections in pp collisions at  $\sqrt{s} = 13$ , 14, 33 and 100 TeV". In: *Eur. Phys. J.* C74.12 (2014), p. 3174. DOI: 10.1140/epjc/s10052-014-3174-y. arXiv: 1407.5066 [hep-ph].

#### **BIBLIOGRAPHY**

- [78] Wolfgang Lukas. "Fast Simulation for ATLAS: Atlfast-II and ISF". In: Journal of Physics: Conference Series 396.2 (Dec. 2012), p. 022031. DOI: 10.1088/1742-6596/396/2/022031. URL: https://doi.org/10.1088%2F1742-6596%2F396%2F2%2F022031.
- Simone Alioli et al. "A general framework for implementing NLO calculations in shower Monte Carlo programs: the POWHEG BOX". In: Journal of High Energy Physics 2010.6 (June 2010), p. 43. ISSN: 1029-8479. DOI: 10.1007/JHEP06(2010)043. URL: https://doi.org/10.1007/JHEP06(2010)043.
- [80] Hung-Liang Lai et al. "New parton distributions for collider physics". In: Phys. Rev. D 82 (7 Oct. 2010), p. 074024. DOI: 10.1103/PhysRevD.82.074024. URL: https: //link.aps.org/doi/10.1103/PhysRevD.82.074024.
- [81] Peter Onyisi. "Event-Driven Messaging for Offline Data Quality Monitoring at AT-LAS". In: Journal of Physics: Conference Series 664.6 (Dec. 2015), p. 062045. DOI: 10.1088/1742-6596/664/6/062045. URL: https://doi.org/10.1088%2F1742-6596%2F664%2F664%2F062045.
- [82] ATLAS Collaboration. "Performance of b-Jet Identification in the ATLAS Experiment". In: *JINST* 11 (2016), P04008.
- [83] ATLAS Collaboration. "Optimisation of the ATLAS b-tagging performance for the 2016 LHC Run". In: (2016). URL: http://cds.cern.ch/record/2160731.
- [84] Glen Cowan et al. "Asymptotic formulae for likelihood-based tests of new physics". In: Eur. Phys. J. C71 (2011). [Erratum: Eur. Phys. J.C73,2501(2013)], p. 1554. DOI: 10.1140/epjc/s10052-011-1554-0, 10.1140/epjc/s10052-013-2501-z. arXiv: 1007.1727 [physics.data-an].
- [85] A L Read. "Presentation of search results: the CL s technique". In: Journal of Physics G: Nuclear and Particle Physics 28.10 (2002), p. 2693. URL: http://stacks.iop. org/0954-3899/28/i=10/a=313.